

ELECTRONIC EXECUTION AND DELIVERY

TO:	Galleria Developments (Block 5 A) Inc. (the "Vendor")
RE:	Purchaser(s):
	(the " Purchaser ")
	Purchase of Unit, Level, being Suite No in the condominium developmen referred to as "Galleria 01" (the "Condominium") by the Purchaser from the Vendor pursuant to ar agreement of purchase and sale (the "Purchase Agreement") executed at or about the same time as the document hereof.

For good and valuable consideration, the receipt and sufficiency of which is acknowledged by the Purchaser, the Purchaser hereby agrees as follows:

- 1. The Purchaser elects to execute and deliver the Purchase Agreement and related documentation, and to receive the Purchase Agreement signed by all parties thereto (the "Fully Executed Purchase Agreement"), disclosure statement for the Condominium (the "Disclosure Statement") and any other documents, notices or other instruments that the Vendor is required to deliver pursuant to the Purchase Agreement or applicable law, electronically using:
 - (a) any electronic signing and delivery platform that the Vendor may use from time to time in the Vendor's sole and unfettered discretion;
 - (b) electronic mail (to the electronic mail address that the Purchaser has entered on the Tarion Addendum attached to the Purchase Agreement (its "**Email Address**")); and/or
 - (c) any other electronic service or device.
- 2. By making the election in Section 1 above, the Purchaser acknowledges and agrees that the Vendor is not obligated to deliver the Purchase Agreement, the Disclosure Statement or any other documents in connection with the Purchase Agreement or Disclosure Statement (each, a "Document" and all of them, the "Documents"); however, notwithstanding the foregoing, the Vendor may elect in its sole and unfettered discretion from time to time to deliver or have its agents or solicitors deliver, or to receive from the Purchaser, one or more Documents in paper form. If the Vendor does elect to deliver or receive one or more Document in paper form, such delivery shall not bind the Vendor to deliver or receive any other Document in paper form.
- 3. Any Document executed and/or delivered electronically by the Purchaser or the Vendor or the Vendor's solicitors or agents shall have the same force and effect as, and for all other intents and purposes shall be deemed to be, an original paper form of that Document.
- 4. The Purchaser agrees that any Document attached to an email message sent to the Purchaser's Email Address by the Vendor, any of its agents or solicitors or any electronic signature platform used by the Vendor (including without limitation DocuSign (as hereinafter defined)), shall be deemed to have been immediately received by the Purchaser unless the sender receives back an email message indicating non-delivery. The foregoing shall apply notwithstanding that the Purchaser may not have opened such email message or the Purchaser's email service provider or application may have classified the email message as SPAM or otherwise diverted the email message.
- 5. Without limiting any of the foregoing, the Purchaser hereby designates the platform operated by DocuSign Inc. (such platform "DocuSign") as an information system that it will use for the purpose of receiving the Disclosure Statement and the Fully Executed Purchase Agreement and the Purchaser will have received the Disclosure Statement and the Fully Executed Purchase Agreement upon the earlier of the following:
 - (a) when the Purchaser enters DocuSign's online platform and becomes capable of retrieving and processing the Disclosure Statement and the Fully Executed Purchase Agreement (as described in Section 22(3)(a) of the *Electronic Commerce Act* (Ontario)); and
 - (b) when the Purchaser receives the Disclosure Statement and Fully Executed Purchase Agreement as attachments to an email sent to the Purchaser's Email Address in accordance with Section 4 above
- 6. The Purchaser acknowledges that it is not under any obligation to sign this electronic execution and delivery document, and in the event that it does not do so, the Purchase Agreement may be executed

in paper form and all Documents will be delivered in paper form pursuant to the Vendor's standard paper form procedures.

7. The Purchaser is advised that it is in its best interests to review all Documents provided electronically thoroughly and diligently and to seek legal representation immediately to assist in such review, and that in accordance with the provisions of Subsection 73(1) of the *Condominium Act, 1998* (Ontario), the Purchaser may rescind (cancel) the Purchase Agreement within 10 calendar days of receipt of the Fully Executed Purchase Agreement and a Disclosure Statement, without penalty, and have its deposit returned.

DATED this day	of, 20
	Signature:
Witness	Purchaser Name:
	Signature:
Witness	Purchaser Name:

	N1 °			
UNIT NO.:LEVEL NO.: SUITE NO.: AGREEMENT OF PURC	HASE AND SALE			
	U =			
The undersigned, (collectively, the "Purchaser") hereby agrees with GALLERIA DEVELOPMENTS (BLOCK 5 A) INC. (the "Vendor") on the terms, conditions and provisions set out below and in the Schedules attached hereto, to purchase the above-referenced proposed Unit, being the above-referenced suite to be located on lands on the south side of Dupont Street and the west side of Dufferin Street, in the City of Toronto in the development known as "Galleria 01" in the master development known as "Galleria on the Park" (a floor plan of which is set out in Schedule "C"), finished in accordance with the specifications listed in Schedule "A", together with an undivided interest in the common elements appurtenant to the Property and the exclusive use of those parts of the common elements attached to such Property, if any, as set out in the Declaration and together with Parking Unit(s) and Storage Locker Unit(s) (as defined in the Creating Documents), which Parking Unit(s) and Storage Locker Unit(s), if any, shall be allocated by the Vendor prior to the Closing Date in such location as determined by the Vendor in its sole and unfettered discretion (hereinafter collectively called the "Purchased Home"), all in accordance with the Creating Documents to be registered against the Land, as hereinafter defined.				
1. PURCHASE PRICE				
The Purchase Price of the Purchased Home shall be the sum of (\$)			
DOLLARS (inclusive of HST, net of the HST new housing rebate (if any), as more particularly set forth in and subject to the provisions of Section 5.3 of Schedule "B") of lawful money of Canada payable as follows:				
(a) The sum of (\$5,000.00) FIVE THOUSAND DOLLARS, as an initial deposit by cheque payable to McMillan LLP, in trust, accompanying this Purchase Agreement as an initial deposit upon the execution of this Purchase Agreement;				
(b) the sum of (\$) DOLLARS by cheque post-dated to the 30 th day following execution of this Purchase Agreement (and representing together with the deposit paid in 1(a), 5% of the Purchase Price) payable to McMillan LLP, in trust, accompanying this Purchase Agreement as a further deposit pending completion or other termination of this Purchase Agreement and to be credited on account of the Purchase Price on the Closing Date;				
(c) the sum of (\$) DOLLARS by cheque payable to McMillan LLP, in trust, accompanying this Purchase Agreement and post-dated to the 150 th day following execution of this Purchase Agreement (and representing 5% of the Purchase Price), as a further deposit pending completion or other termination of this Purchase Agreement and to be credited on account of the Purchase Price on the Closing Date;				
Purchase Agreement (and representing 5% of the Purchase Price), as a furth and to be credited on account of the Purchase Price on the Closing Date;	Purchase Agreement and post-dated to the 400 th day following execution of this ler deposit pending completion or other termination of this Purchase Agreement			
(e) the sum of (\$) DOLLARS by cheque payable to McMillan LLP, in trust, on the Firm Occup pending completion or other termination of this Purchase Agreement and to be	pancy Date (and representing 5% of the Purchase Price), as a further deposit the credited on account of the Purchase Price on the Closing Date;			
(f) the amount required to be paid pursuant to Section 3.4(10) of Schedule "B" by cheque payable to McMillan LLP, in trust, on the Firm Occupancy Date, as a further deposit pending completion or other termination of this Purchase Agreement and to be credited on account of the Purchase Price on the Closing Date;				
(g) the balance of the Purchase Price by certified cheque drawn on the trust account of the Purchaser's Solicitor payable to the Vendor or as the Vendor may in writing direct on the Closing Date, unless the Purchaser provides to the Vendor, within the time period required by the Condo Act, the notice contemplated by Section 3.4(6) of Schedule "B", in which case the balance of the Purchase Price shall be paid on the Firm Occupancy Date, as provided for in Section 3.4(6) of Schedule "B"; and				
(h) all proper and usual adjustments including the adjustments specificall Closing Date.	y provided under this Purchase Agreement, calculated as of and paid on the			
The Purchaser agrees that on taking possession of the Purchased Home and on the Closing Date the Purchaser will not for any reason whatsoever request, nor will the Vendor be obligated to permit, any holdback of any part of the Purchase Price or any other monies payable by the Purchaser under this				
Purchase Agreement for any reason whatsoever. 2. CLOSING				
(a) The Vendor presently estimates that the tentative date on which the Purchaser will be able and obliged to occupy the Purchased Home shall be the First Tentative Occupancy Date as set out on and in accordance with the Statement of Critical Dates forming part of this Purchase Agreement and the Purchaser agrees to occupy the Purchased Home on such date or such earlier or extended date as permitted by this Purchase Agreement.				
(b) The purchase and sale of the Purchased Home shall be completed after the Creating Documents have been registered, on the later of the Firm Occupancy Date and a date fixed by the Vendor in accordance with Section 9.1 of Schedule "B" of this Purchase Agreement, or such later date as permitted				
	DOD ALL PROVISIONS OF THIS PURCHASE AGREEMENT, INCLUDING			
THOSE IN SCHEDULES "A", "B", "C", (if any) TO THIS PURCHASE AGREEMENT WHICH ARE AN INTEGRAL PART HEREOF. DATED this day of, 20				
DATED this day of, 20 SIGNED, SEALED & DELIVERED in the presence of				
	ature:			
Witness Purchaser N Date o	Name:			
Di Licence/Pass	river's			
	ature:			
Date of	Name: Birth:			
D	river's port #			
Purchaser 1 Address for Service:	Purchaser 2 Address for Service:			
Name:Street Address:	Name:Street Address:			
City:Postal:	City Province: Postal:			
Province: Postal: Res Phone No.:	Province: Postal: Res Phone No.:			
Bus. Phone No.:	Bus. Phone No.:			
Fax No.: Email:	Fax No.: Email:			
The Vendor accepts the above Offer and agrees to complete this transaction in accordance with the terms hereof.				
ACCEPTED this day of, 20				
	PMENTS (BLOCK 5 A) INC.			
McMillan LLP by its authorized repres				
Suite 4400, 181 Bay Street, Brookfield Place Toronto, Ontario, M5J 2T3 Attention: New Homes and				
Condominiums Law Clerk Tel: (416) 865-7151 Fax: (647) 722-6716				

Vendor's Address: c/o 2235 Sheppard Avenue East, Suite 1700, Toronto, ON M2J 5B5



SCHEDULE "A" - FEATURES AND FINISHES

BUILDING FEATURES AND AMENITIES

- Ground floor lobby with mail room, parcel storage and stroller storage room
- Concierge services
- Garbage chute system with bi-sorter for refuse and organics, plus dedicated recycling chute
- Residential parking, electric vehicle charging stations and lockers (all subject to availability)
- Bike spaces
- Resident security access

SUITE FEATURES

- Suites have approximately 9' ceiling heights**
- Smooth finish painted ceilings throughout
- Off-white paint throughout suite interiors
- · Solid core entry door
- Composite hollow core interior doors, painted, with hardware
- Sliding closet doors where shown on plan
- White décor style switches and receptacles
- Selection of laminate flooring in Living/Dining Room, Kitchen, Den, Flex, Bedrooms, Media area and Foyer, including Foyer closet where shown on plan
- Individual sub metering for hydro, water, heating and cooling
- Four pipe fan coil heating/cooling system

ELECTRICAL AND TECHNICAL FEATURES

- Light fixture in Foyer, Kitchen, Den, Flex room, Dining Room and Bedroom(s) as per plan
- Outlet for cable and internet connections as per plan

IN-SUITE LAUNDRY

- Stacked washer and dryer
- Pre-selected white porcelain or ceramic tile flooring

KITCHENS

- Selection of flat paneled cabinets and finishes
- Selection of quartz countertops
- Stainless steel undermount sink and single lever faucet and matching sidespray
- Selection of porcelain or ceramic tile backsplash

APPLIANCES

Studio, 1 bed, 1 + den, 1 + flex and 2 bedrooms suites

- Paneled 24" refrigerator
- 24" built-in oven and 24" ceramic cooktop
- Paneled 24" dishwasher
- Full-size countertop microwave, in selected suites

3 bed and up

- Paneled 30" refrigerator
- 30" built-in oven and 30" ceramic cooktop
- Paneled 24" dishwasher
- Full-size countertop microwave, in selected suites

BATHROOMS

- Vanity with above-counter sink
- Selection of porcelain or ceramic tile flooring
- Vanity mirror
- Decorative light fixture
- Shower stall with framed glass door panel and tile base
- Soaker tub where shown on plan
- Selection of porcelain or ceramic tile for tub or shower surrounds extending to the underside of ceiling
- Single lever faucet
- Shower/tub fixtures
- Accessories including toilet paper holder and towel bar

Please note: Prices and specifications are subject to change without notice. Vendor shall have the right to substitute materials for those provided in the plans and specifications provided that such materials are of quality equal to or better than the material provided for in the plan and specifications. Colour, texture, appearance, etc. of feature and finishes installed in the unit may vary from Vendor's samples as a result of normal manufacturing and installation processes. Floors and specific finishes will depend on Vendor's décor packages as selected. Amenities and their components are subject to change based on governmental Vendor, or construction requirements.

** Ceiling heights are measured from the finished concrete floor slab to the underside of the concrete slab above. In areas where bulkheads or dropped ceilings are required, the ceiling height will be less than the specified 9'.

Decorative and upgrade items are displayed in the model suite and are not part of the standard unit. Vignette and sales office are for display purposes only.

E. & O.E.



AGREEMENT OF PURCHASE AND SALE

SCHEDULE "B"

PART I - DEFINITIONS

1.1 **DEFINITIONS**

The meaning of words and phrases used in this Purchase Agreement shall have the meanings given to them in the Condo Act, and the Condominium Documents unless otherwise provided for herein.

- "Addendum" means the Tarion Addendum attached to this Purchase Agreement, including the accompanying Statement of Critical Dates.
- "Agreement Cover" means the cover page executed by the parties hereto to which this Schedule is attached.
- "Alternate Entity" has the meaning given to it in Section 10.1(4) of this Schedule.
- "Approving Authority" has the meaning given to it in Section 2 of Schedule A of the Addendum.
- "<u>Budget Statement</u>" means the operating budget for the first year following registration of the Creating Documents provided to the Purchaser with the Disclosure Statement.
- "<u>Building</u>", when used in this Schedule only, means the Block 5 Building, as that term is defined in the Disclosure Statement.
- "Business Day" has the meaning given to it in Section 12 of the Addendum.
- "CCP" has the meaning given to it in Section 6.1 of this Schedule.
- "Closing" means the moment at which the transaction of purchase and sale contemplated by this Purchase Agreement is completed.
- "Closing Date" has the meaning given to it in Paragraph 2(b) of the Agreement Cover.
- "Closing Date Notice" has the meaning given to it in Section 9.1(1) of this Schedule.
- "Commitment" has the meaning given to it in Section 2.2(2) of this Schedule.
- "Common Expenses" means the expenses of the performance of the objects and duties of the Corporation and any expenses specified as common expenses in the Condo Act or in the Creating Documents.
- "Condo Act" means the Condominium Act, 1998, S.O. 1998, c. 19 and any amendments and regulations, or any replacement legislation and the regulations to the Condo Act.
- "Condominium" means the condominium which will be registered against the Lands or a part thereof pursuant to the provisions of the Condo Act, including part of the Building described in the Creating Documents and all improvements to be constructed on the Lands.
- "<u>Condominium Documents</u>" means, collectively, the Creating Documents, the Condominium by-laws and rules from time to time, and the Disclosure Statement (including the Budget Statement) and shall also include any amendments thereto from time to time.
- "Confirmation of Receipt" means the Confirmation of Receipt form prescribed from time to time by Tarion, under which the Purchaser or the Purchaser's Designate acknowledges receipt of the HIP.
- "Construction Activities" has the meaning given to it in the Disclosure Statement.
- "Construction Warranties Agreement" has the meaning given to it in Section 6.1(4) of this Schedule.
- "Corporation" means the condominium corporation created under the Condo Act on registration of the Creating Documents.
- "Creating Documents" means the Declaration and the Description required to be registered under the Condo Act in order to create the Condominium.

- "Critical Dates" has the meaning given to it in Section 12 of the Addendum.
- "Declaration" means the proposed Declaration of the Condominium.
- "Delayed Occupancy Date" has the meaning given to it in Section 12 of the Addendum.
- "<u>Deposit</u>" means the initial deposit and further deposits referred to in Paragraph 1 of the Agreement Cover, when and if paid, including Interest.
- "<u>Designate</u>" means a representative of the Purchaser, if any, appointed by the Purchaser and approved by the Vendor in writing on a form provided by the Vendor, such representative being authorized to, among other things, make selections on behalf of the Purchaser and to complete the CCP, the PDI Form and the Confirmation of Receipt on behalf of the Purchaser, all in accordance with this Purchase Agreement.
- "<u>Developers</u>" has the meaning given to it in the Disclosure Statement.
- "<u>Disclosure Statement</u>" means the disclosure statement furnished under the Condo Act and delivered to the Purchaser as part of the Condominium Documents.
- "Donee" has the meaning given to it in Section 11.1(4) of this Schedule.
- "Donor" has the meaning given to it in Section 11.1(4) of this Schedule.
- "ECA" has the meaning given to it in Section 9.1(3) of this Schedule.
- "Escrow Agreement" has the meaning given to it in Section 9.5(1) of this Purchase Agreement.
- "Extras" means any upgrades, or changes to the Property or extras in regard to the Property ordered or requested by the Purchaser, if any, which are accepted in writing by the Vendor in the Vendor's sole and unfettered discretion or which are contracted for separately between the Vendor and Purchaser and not as part of this Purchase Agreement.
- "Firm Occupancy Date" has the meaning given to it in Section 12 of the Addendum.
- "First Tentative Occupancy Date" has the meaning given to it in Section 12 of the Addendum.
- "Galleria Master Development" has the meaning given to it in the Disclosure Statement.
- "Galleria Master Development Lands" has the meaning given to it in the Disclosure Statement.
- "HIP" has the meaning given to it in Section 6.1(2) of this Schedule.
- "HST" has the meaning given to it in Section 5.2 of this Schedule.
- "HST Act" means the Excise Tax Act (Canada) and any amendments and regulations thereto or any replacement legislation thereof and the regulations thereto.
- "include" and "including" mean, respectively, "include without limitation" and "including without limitation"
- "Interest" means only interest, if any, which the Vendor is required to pay to the Purchaser on any Deposit pursuant to the Condo Act, calculated at the prescribed rate under the Condo Act.
- "Land Registry Office" means the Land Registry Office for the Land Titles Division in which the Lands are recorded and registered.
- "Lands" means the Condominium Lands, as that term is defined in the Disclosure Statement.
- "Occupancy" has the meaning given to it in Section 12 of the Addendum.
- "<u>Occupancy Arrangement</u>" means the arrangement for the occupancy of the Purchased Home on the Firm Occupancy Date set out in Section 3.4 of this Schedule.
- "Occupancy Fee" means the monthly fee referred to in Section 3.4 of this Schedule.
- "Parking Unit" has the meaning given to it in the Creating Documents.
- "PDI" has the meaning given to it in Section 6.1(1) of this Schedule.
- "PDI Form" has the meaning given to it in Section 6.1(1) of this Schedule.
- "Permanent Lender" has the meaning given to it in Section 2.2(2) of this Schedule.

- "Permitted Encumbrances" has the meaning given to it in Section 4.2(1) of this Schedule.
- "<u>Prime Rate</u>" means the fluctuating annual rate of interest established from time to time by The Toronto-Dominion Bank as the base rate it will use to determine interest rates in Canadian dollar loans to customers in Canada and quoted or designated by it as its commercial prime rate.
- "Property" has the meaning given to it in Section 12 of the Addendum and Properties means more than one Property.
- "<u>Purchase Agreement</u>" means this Agreement of Purchase and Sale and includes the Agreement Cover, the Addendum and all Schedules attached to the Agreement Cover which are incorporated by reference.
- "Purchase Price" means the purchase price referred to in Paragraph 1 of the Agreement Cover.
- "Purchased Home" means the Property and any Parking Unit(s) and/or Storage Locker Unit(s) purchased together with the Property as indicated on the Agreement Cover.
- "<u>Purchaser's Solicitors</u>" means a solicitor in good standing with the Law Society of Ontario to represent the Purchaser, selected by the Purchaser to act for and on behalf of the Purchaser in connection with this transaction.
- "Rebate" has the meaning given to it in Section 5.3(1) of this Schedule.
- "Reciprocal Agreements" has the meaning given to it in the Disclosure Statement.
- "Residential Unit" has the meaning given to it in the Creating Documents.
- "<u>Schedule</u>" means this Schedule "B" to the Purchase Agreement when used in this Schedule, unless the context otherwise requires.
- "Shared Facilities" has the meaning ascribed thereto in the Disclosure Statement.
- "Smart Home Devices" has the meaning given to it in the Disclosure Statement.
- "Smart Home Services" means any and all services supplied in respect of Smart Home Devices.
- "Smart Home Services Charges" means any and all charges in respect of Smart Home Services.
- "Smart Home Services Provider" means any provider of Smart Home Services.
- "<u>Smart Meter</u>" means an individual consumption meter installed for the purposes of monitoring the consumption of hydro services provided to the Property.
- "<u>Smart Metering</u>" any service in connection with the Smart Meters for the Property including the installation, maintenance and reading thereof.
- "Smart Metering Charges" means any and all charges in respect of Smart Metering.
- "<u>Smart Metering Provider</u>" means any public or private entity which provides any Smart Metering to the Property.
- "Statement of Critical Dates" has the meaning given to it in Section 12 of the Addendum.
- "Storage Locker Unit" has the meaning given to it in the Creating Documents.
- "Tarion" means Tarion Warranty Corporation and its successors and assigns.
- "TERS" has the meaning given to it in Section 9.5 of this Schedule.
- "The ONHWP Act" has the meaning given to it in Section 12 of the Addendum.
- "<u>Unauthorized Work</u>" has the meaning given to it in Section 3.4(14) of this Schedule.
- "Unavoidable Delay" has the meaning given to it in Section 12 of the Addendum.
- "Unavoidable Delay Period" has the meaning given to it in Section 12 of the Addendum.
- "<u>Uncompleted Extras</u>" has the meaning given to it in Section 6.2(9) of this Schedule.
- "<u>Unit</u>" has the meaning given to it in the Creating Documents.
- "<u>Utility Services</u>" means, collectively, the following services: electricity, hot water, cold water, heating and cooling;

- 4 -

"Utility Service Charges" means any and all charges in respect of Utility Services.

"<u>Utility Service Provider</u>" means any public or private entity which provides Utility Services to the Property.

"Vendor's Lender" has the meaning given to it in Section 2.2(3) of this Schedule.

"<u>Vendor's Solicitors</u>" means the firm of McMillan LLP, Brookfield Place, Suite 4400, 181 Bay Street, Toronto, Ontario M5J 2T3.

PART II - FINANCIAL MATTERS

2.1 INTEREST

The Vendor shall pay or credit Interest to the Purchaser on the Deposit on the Firm Occupancy Date, or the Closing Date, as determined by the Vendor in its sole and unfettered discretion and in accordance with the Condo Act. The Purchaser agrees that the Vendor shall have the right, in its sole and unfettered discretion, to deduct or set off the amount of any Interest obligation to the Purchaser, against any other amounts owing by the Purchaser to the Vendor under this Purchase Agreement, as a result of any default under this Purchase Agreement by the Purchaser or otherwise. The Purchaser acknowledges that for the purposes of the payment of Interest, the Deposit shall be deemed not to have been received until the day on which any particular Deposit cheque is deposited into the Vendor's Solicitors' bank account.

2.2 PURCHASER'S FINANCIAL INFORMATION AND FINTRAC REQUIREMENTS

- The Purchaser hereby warrants and represents that the Purchaser has (or will have, on or (1) before the Closing Date) the financial resources to complete the purchase and sale transaction contemplated herein on an all-cash basis to the Vendor, either wholly from the Purchaser's own resources or from third party financing that will ultimately be secured by one or more mortgages given by the Purchaser and registered against the Purchased Home on or shortly after the Closing Date. All Deposits due or payable by the Purchaser to the Vendor in accordance with the terms and provisions of this Purchase Agreement shall be made or tendered by way of a bank draft or cheque drawn on (or issued by) one of Bank of Montreal, Bank of Nova Scotia, Canadian Imperial Bank of Commerce, Royal Bank of Canada, Toronto-Dominion Bank or HSBC Bank Canada on the express understanding that if a cheque or bank draft is delivered to the Vendor by a third party on behalf of the Purchaser (i.e. drawn on the bank account of such third party, rather than on the bank account of the Purchaser), then such bank draft or cheque shall be deemed to be a payment made by such third party as agent for and on behalf of the Purchaser, in which case it is agreed that the certificate confirming that such Deposits are being held in a designated trust account by the Vendor's Solicitors pending completion or termination of this transaction or the provision of prescribed security in respect of same shall be issued and delivered directly to the Purchaser only, and not to such third party.
- (2) The Purchaser is hereby notified that a consumer report containing credit and personal information may be referred to at any time in connection with this transaction and the Purchaser hereby consents to such report being obtained by the Vendor. The Purchaser agrees to submit to the Vendor within five (5) days of demand made by any of the Vendor or the Vendor's Solicitors (which demand may be made at any time or times) any application made by or on behalf of the Purchaser to a third party institutional lender for the financing of any part of the Purchase Price (the "Permanent Lender") in respect of a commitment for permanent mortgage financing in respect of the Purchased Home from a Permanent Lender (the "Commitment") and the Purchaser, by executing this Purchase Agreement, hereby irrevocably authorizes and directs the Permanent Lender to release to the Vendor, at such time or times as the Vendor may request, all information and documentation in the Permanent Lender's possession and control respecting the Commitment, including any application made by or on behalf of the Purchaser and any Commitment.
- (3) In addition to any other remedies permitted herein, if the Purchaser fails to provide a Commitment from a Permanent Lender or if the Commitment is not satisfactory to the Vendor or its construction lender, in their sole and unfettered discretion, then the Vendor may require that the Purchaser forthwith apply to a lending institution designated by the Vendor (the "Vendor's Lender") to obtain a mortgage for the Purchaser's financing at a principal amount not less than the Purchaser's financing as set out in the Commitment, for a term of three (3) to five (5) years as designated by the Vendor and on terms then offered by the Vendor's Lender for similar financing. It is understood and agreed that: (i) the Vendor's Lender may vary the Purchaser's financing provisions, but not the principal amount; and (ii) the Vendor may vary the term of the Purchaser's financing by decreasing it to not less than three (3) years or increasing it to not more than five (5) years.
- (4) At any time prior to the Closing Date, the Vendor may require the Purchaser to provide documentation that the Vendor deems necessary for the Vendor or any agent thereof to comply with, or verify compliance with, any rules, regulations or guidelines published or otherwise issued by the Financial Transactions and Reports Analysis Centre of Canada or

any applicable law, including any provision of the *Proceeds of Crime (Money Laundering)* and *Terrorist Financing Act* (PCMLTFA) or any similar or successor legislation, or any regulations or rules published or otherwise issued pursuant thereto. Such documentation may include without limitation personal or corporate identification documentation for the Purchaser. Any such documentation shall be delivered to the Vendor as set out in this Section 2.2(4) of this Schedule within five (5) days of a written request therefor by the Vendor addressed to the Purchaser. The Purchaser shall submit such documentation in any format required by the Vendor including notarized or commissioned copies and the Vendor may require that the Purchaser present such documentation in person at the Vendor's corporate office or the Vendor's sales office in Toronto, Ontario. The requirements of this Section 2.2(4) of this Schedule shall apply notwithstanding any prior verification of the Purchaser's identification documentation, or any other documentation, by the Vendor or any agent thereof.

- (5) If the Purchaser fails to submit information, evidence and/or documentation required by this Purchase Agreement to be submitted by the Purchaser within the time periods set forth in this Purchase Agreement, or as otherwise required by the Vendor or the Vendor's Solicitors, or if the information, evidence and/or documentation submitted is, in whole or in part, false or misleading, or if the Purchaser fails to disclose any relevant facts pertaining to the Purchaser's financial circumstances or abilities, then the Purchaser shall be deemed to be in default under this Purchase Agreement, and the provisions of Section 8.1(3) of this Schedule shall apply.
- (6) If this Purchase Agreement is not terminated pursuant to Section 10 of the Addendum and subsequently there is a material adverse change in the financial circumstances of the Purchaser to the extent that the Purchaser does not have the ability to pay any Deposit or the cash balance of the Purchase Price on the Firm Occupancy Date or the Closing Date, due to circumstances within the control of the Purchaser, or as a result of materially inaccurate or incomplete information being provided to the Vendor by the Purchaser, which shall be determined in the sole and unfettered discretion of the Vendor, the Purchaser shall be deemed to be in default under this Purchase Agreement and the provisions of Section 8.1(3) of this Schedule shall apply.
- (7) The Purchaser covenants and agrees that the Purchaser, or any other person approved in writing by the Vendor, which approval shall be in the sole and unfettered discretion of the Vendor, shall personally occupy the Purchased Home within fourteen (14) days from the Firm Occupancy Date and shall continue to personally occupy the Purchased Home until the Closing Date.
- (8) The Purchaser further covenants and agrees that until Closing:
 - (a) the Purchaser will not either directly or indirectly list for sale, offer to sell, advertise to sell, sell, mortgage, pledge, lien or in any way encumber the Purchased Home:
 - (b) if a Writ of Execution is filed against the Purchaser and/or the Purchased Home the Purchaser will forthwith have the execution removed; and
 - (c) if a Writ of Execution is registered against person(s) with a similar name(s), the Purchaser shall execute and provide all documents required by the Vendor in its sole and unfettered discretion, to evidence that the Purchaser is not the same person(s) named in such a Writ of Execution(s), sufficient to enable the Vendor to obtain a clear Certificate as to Writs of Execution from the Land Registry Office.

2.3 RECEIPT OF DEPOSIT

The Vendor's Solicitors shall provide the Purchaser with evidence of compliance with Section 81 of the Condo Act, in the prescribed form. Such evidence shall be provided by written notice to the Purchaser delivered in accordance with Section 11.2 of this Schedule.

2.4 RELEASE OF DEPOSIT

The Purchaser agrees, and hereby so irrevocably directs the Vendor's Solicitors, that the Vendor's Solicitors may release any part of the Deposit to the Vendor upon: (i) the Vendor ensuring that security of a prescribed class for the Deposit has been issued in accordance with the requirements of the Condo Act; (ii) the Vendor's Solicitors receiving a written request for release; and (iii) the Vendor's Solicitors receiving confirmation from the Vendor that it has arranged such security and that such security is in full force and effect as of the date of such request and confirmation. The Vendor's Solicitors shall have absolutely no obligation to make any independent enquiry in regard to the information set forth in such request and confirmation, including as to its veracity, and shall have absolutely no liability or obligation to the Purchaser in regard to any part of the Deposit released in accordance herewith.

2.5 WRITS OF EXECUTION

On each of the Firm Occupancy Date and the Closing Date, the Purchaser, at the Purchaser's sole expense, shall deliver to the Vendor a clear and up-to-date Certificate as to Writs of Execution in respect of the Purchaser's name from the Land Registry Office and shall also provide the Vendor with such other information, documentation and other evidence as may be required in the sole and unfettered discretion of the Vendor's Solicitors to enable the Vendor to obtain a clear Certificate as to Writs of Execution with respect to the Purchaser from such office.

2.6 RECEIPT OF PURCHASE AGREEMENT

(a) If this Purchase Agreement is signed by the Purchaser in paper format, the Purchaser shall attend at the Vendor's sales office to pick up a fully executed copy of this Purchase Agreement and to execute an acknowledgement confirming the Purchaser's receipt of a fully executed copy of the Purchase Agreement and the Condominium Documents within 10 calendar days of the date that the Vendor sends written notice to the Purchaser at the address or email address set out on the Agreement Cover, advising the Purchaser that the Purchase Agreement has been accepted by the Vendor and is ready for pick up. A failure by the Purchaser to so attend and to execute the said acknowledgement shall constitute a default of the Purchaser under this Purchase Agreement and the provisions of Section 8.1(3) of this Schedule shall apply without requiring the delivery of further notice.

(b) If this Purchase Agreement is signed by the Purchaser electronically and delivered to the Purchaser electronically (either by email or by any other platform) the Purchaser shall, at the Vendor's Election attend at the Vendor's sales office to execute an acknowledgement, or sign such an acknowledgement electronically, confirming the Purchaser's receipt of a fully executed copy of the Purchase Agreement and the Condominium Documents within 10 calendar days of the date that the Vendor sends written notice to the Purchaser at the address or email address set out on the Agreement Cover, advising the Purchaser that the Purchase Agreement has been accepted by the Vendor. A failure by the Purchaser to so execute the said acknowledgement shall constitute a default of the Purchaser under this Purchase Agreement and the provisions of Section 8.1(3) of this Schedule shall apply without requiring the delivery of further notice.

2.7 VENDOR'S LIEN

The Purchaser agrees that: (i) the Vendor shall have a vendor's lien for unpaid purchase monies (including any monies owing for Extras, interest, Rebate, adjustments and HST) outstanding on the Closing Date; and (ii) the Vendor shall be entitled to register a notice of lien against the Purchased Home any time after the Closing Date in respect of any such unpaid purchase monies.

2.8 CANADIAN FUNDS

Any reference made to money or funds herein shall be interpreted to be in Canadian funds, unless otherwise provided.

PART III – OCCUPANCY

3.1 OCCUPANCY

On the Firm Occupancy Date, the Purchaser shall take Occupancy of the Purchased Home as a monthly licensee, at a monthly fee paid in advance, determined in accordance with the provisions of this Part III.

3.2 SUBSTANTIAL COMPLETION

- (1) If the Condominium is completed so as to permit access thereto and the Purchased Home has been finished so as to permit occupancy thereof in accordance with the minimum standards required by the Approving Authority, the Purchased Home shall be deemed to be substantially completed for all purposes of this Purchase Agreement, notwithstanding that there remains any other work to be completed in the Purchased Home, or in any other Unit or common areas of the Condominium or the Building.
- (2) Except where the Purchaser and the Vendor have agreed that the Purchaser shall be responsible for certain conditions of occupancy and subject to paragraph 9 of the Addendum, the Purchased Home shall be deemed to be substantially completed when the interior work of the Property has been finished to the minimum standards allowed by the Approving Authority so that the Property may be lawfully occupied notwithstanding that there remains other work within the Purchased Home and/or the common elements to be completed which has not been completed or any such areas are not available for occupancy. Except where the Purchaser is responsible (pursuant to an express written agreement) for certain conditions of occupancy, the Purchaser shall not occupy the Purchased Home until the Approving Authority has permitted same or consented thereto, if such consent is required.
- (3) In respect of the Occupancy of the Purchased Home, the Purchaser shall only require the documentation set out in Section 9 of the Addendum and shall not call for the production of any other certificate issued by any Approving Authority in respect of such Occupancy. The Purchaser acknowledges that failure by the Vendor to complete the common elements of the Condominium or other Units in the Condominium before either the Firm Occupancy Date or the Closing Date, shall not be deemed to be a failure to substantially complete the Purchased Home and that the Purchaser shall not be entitled to maintain any holdback of any part of the Purchase Price for completion of the common elements or the Purchased Home, or otherwise, nor be entitled to any payment or other compensation with respect thereto.

3.3 EXTENSIONS/ACCELERATIONS

- The Purchaser and the Vendor hereby acknowledge and agree that any extension or (1) acceleration of any of the Critical Dates shall be governed by the terms of the Addendum. If the Creating Documents have been registered on title to the Lands and the Vendor's Solicitors have notified the Purchaser or the Purchaser's Solicitors of same, and the Vendor's Solicitors have also confirmed in writing to the Purchaser or the Purchaser's Solicitors that they are ready to deliver a registerable deed/transfer of title to the Purchased Home to the Purchaser in accordance with the terms and provisions hereof, then the Firm Occupancy Date or Delayed Occupancy Date, as the case may be, shall also constitute and be construed as the "Closing Date" under this Purchase Agreement, and the provisions of this Purchase Agreement shall then be amended accordingly mutatis mutandis to give effect to same. Notwithstanding anything contained in this Section 3.3, the Vendor may, in its sole and unfettered discretion, on one occasion, unilaterally extend the Firm Occupancy Date, the Delayed Occupancy Date or the Closing Date, as the case may be, for one (1) Business Day to avoid the necessity of tender where the Purchaser is not ready to close on the Firm Occupancy Date, Delayed Occupancy Date or Closing Date, as the case may be. The Purchaser acknowledges that delayed occupancy compensation will not be payable to the Purchaser for such period and that the Vendor may not impose any penalty or interest charge upon the Purchaser with respect to such extension.
- (2) The provisions of the Addendum reflect Tarion's policies, regulations and/or guidelines on extensions of the First Tentative Occupancy Date, but it is expressly understood and agreed by the parties hereto that any failure to provide notice(s) of the extension(s) of the First Tentative Occupancy Date, Subsequent Tentative Occupancy Dates (as defined in the Addendum) or Firm Occupancy Date, in accordance with the provisions of the Addendum shall only give rise to a damage claim by the Purchaser against the Vendor up to a maximum of \$7,500.00, as more particularly set forth in the Regulations to The ONHWP Act, and under no circumstances shall the Purchaser be entitled to terminate this transaction or otherwise rescind this Purchase Agreement as a result thereof, other than in accordance with the Addendum.

3.4 OCCUPANCY ARRANGEMENT

The Purchaser agrees to take possession of the Purchased Home on the Firm Occupancy Date and occupy it as a licensee on the following terms and conditions:

- (1) The Purchaser shall use the Property solely for residential purposes as a private, single family residence, to be occupied by the Purchaser or any other person approved in writing by the Vendor (in its sole and unfettered discretion) prior to such Occupancy, and in accordance with the provisions of the Condominium Documents and this Purchase Agreement.
- (2) The right to Occupancy of the Purchased Home is a personal right which cannot be assigned or sublet without the prior written consent of the Vendor, which consent may be withheld in its sole and unfettered discretion. Where:
 - (a) a request is made of the Vendor to consent to an assignment, subletting or sublicensing of the right to Occupancy of the Purchased Home; and
 - (b) the Vendor consents to the assignment, subletting or sublicensing,

the consent shall be on terms and conditions satisfactory to the Vendor in its sole and unfettered discretion and the Vendor may charge a fee for consenting to such assignment, subletting or sublicensing.

- (3) The Purchaser shall pay the Occupancy Fee to the Vendor as an occupancy charge on the first day of each month in advance during such Occupancy. No part of the Occupancy Fee shall be credited as payment on account of the Purchase Price of the Purchased Home, but such payment shall be a charge for Occupancy only. If the Firm Occupancy Date is not the first day of the month, the Purchaser shall pay on the Firm Occupancy Date, a pro rata amount for the balance of the month in which the Firm Occupancy Date occurs.
- (4) The monthly Occupancy Fee to be charged by the Vendor for the Purchaser's Occupancy of the Purchased Home from the Firm Occupancy Date to the Closing Date, shall be the maximum amount allowed under the Condo Act. That portion of the Occupancy Fee equivalent to the interest component contemplated under the Condo Act shall be ultimately determined at Closing and shall be calculated on a monthly basis on the unpaid balance of the Purchase Price at the prescribed rate; however the Vendor reserves the right to charge the same interest rate on the Firm Occupancy Date of all Units in the Condominium and to readjust to the prescribed rate for each such Unit on the Closing Date. The Purchaser acknowledges and agrees that those portions of the Occupancy Fee equivalent to the realty tax (which may be estimated as if the Purchased Home has been assessed as fully completed by the relevant taxing authority for the calendar year in which the transaction closes, and for the following calendar year, notwithstanding that same may not have been

paid or levied on the Closing Date) and common expense components contemplated under the Condo Act, may be subject to one or more recalculations and/or adjustments by the Vendor, at the Vendor's sole and unfettered discretion, and accordingly the Vendor may increase or decrease the Occupancy Fee, at any time prior to the Closing Date, with any readjustment to be made 30 days after the Vendor's demand therefor, and with a final readjustment (if any) of the monthly Occupancy Fee to be made on the Closing Date, or within a reasonable period of time thereafter, in the Vendor's sole and unfettered discretion (but without any obligation whatsoever on the part of the Vendor to make or effect any such readjustment with respect to that portion of the Occupancy Fee equivalent to the common expense component contemplated under the Condo Act provided that it was reasonably estimated in accordance with the Condo Act).

- (5) Upon payment of the Occupancy Fee and the amount required to be paid pursuant to Section 3.4(10) of this Schedule and delivery of the items referred to in Sections 3.4(15) and 3.4(21) of this Schedule, the Vendor grants to the Purchaser a licence to occupy the Purchased Home from the Firm Occupancy Date until the Closing Date, subject to rights of termination and other conditions as set forth in this Purchase Agreement.
- (6) Notwithstanding any other provision to the contrary in this Purchase Agreement, the Purchaser may, by written notice given to the Vendor before the expiry of the time period for rescinding the Purchase Agreement as provided for in Section 73(2) of the Condo Act, elect to pay the balance of the Purchase Price on the Firm Occupancy Date, in which case:
 - (a) such amount shall be payable by the Purchaser to the Vendor's Solicitors, in trust, by certified cheque or bank draft on the Firm Occupancy Date as a further Deposit pending completion or other termination of this Purchase Agreement and to be credited on account of the Purchase Price on the Closing Date;
 - (b) there shall be no interest component to the Occupancy Fee; and
 - (c) such amount may be released to the Vendor in accordance with Section 2.4 of this Schedule.

In the event that the Purchaser elects to pay the balance of the Purchase Price on the Firm Occupancy Date, in accordance with this Section 3.4(6), but does not in fact make such payment, the Purchaser's failure to make such payment shall constitute a default under this Purchase Agreement and the provisions of Section 8.1(3) of this Schedule shall apply without further notice to the Purchaser. Without limiting the Vendor's rights and remedies when such a default occurs, the Vendor may, in its sole and unfettered discretion, elect to require the Purchaser to (i) pay the balance of the Purchase Price on the Closing Date, (ii) take possession of the Purchased Home in accordance with the Occupancy Arrangement and pay the interest component of the Occupancy Fee during Occupancy, (iii) pay to the Vendor on the Firm Occupancy Date, an administrative fee in the amount of \$500 and (iv) complete this Purchase Agreement, otherwise in accordance with its terms.

- (7) The Purchaser shall, from and after the Firm Occupancy Date, be responsible for and assume and pay all charges for Utility Services, Smart Metering services, Smart Home Services, telephone, internet and any cable television services as well as for any other service or expense directly attributable to the Purchased Home for maintenance and repair or otherwise that would not be the direct responsibility of the Corporation if the Corporation had already come into existence and the Purchaser shall take all necessary steps from and after the Firm Occupancy Date to assume and pay for such charges.
- (8) The Purchaser shall obtain and maintain property damage, public liability and personal property insurance in respect of the Purchased Home, its contents and the Purchaser's Occupancy thereof from the Firm Occupancy Date up to and including the Closing Date and upon the request of the Vendor shall provide evidence of such insurance in a form satisfactory to the Vendor. The Purchaser shall replace and/or repair any betterments or improvements if they are removed, injured or destroyed. The Vendor is not liable for the Purchaser's loss occasioned by fire, theft or other casualty. The Purchaser acknowledges that the Vendor holds a fire insurance policy on the Condominium including all aspects of a standard unit only and not on any improvements or betterments made by or on behalf of the Purchaser. At the request of the Vendor, the Purchaser shall arrange to have the Vendor and any other party requested by the Vendor added as an additional insured under any of the foregoing insurance policies prior to occupancy.
- (9) Subject to the provisions of this Purchase Agreement, the Occupancy Arrangement shall terminate on the Closing Date with an appropriate allowance being given in the statement of adjustments for any portion of the month for which the Occupancy Fee has been paid.
- (10) An estimate of the amount of the adjustments required to be paid by the Purchaser on the Closing Date shall be paid by the Purchaser to the Vendor on the Firm Occupancy Date and such payment by the Purchaser shall, on the Closing Date, be reflected as a credit to the Purchaser in the statement of adjustments.

- (11) Adjustments shall be as of the Closing Date.
- (12) It is acknowledged that possession of the Purchased Home is based on this Purchase Agreement for the purchase of the Purchased Home and such possession shall not alter the terms, conditions and covenants contained in this Purchase Agreement, which shall continue in full force and effect. The Vendor may enforce the provisions of the licence to occupy the Purchased Home separate and apart from the purchase and sale provisions of this Purchase Agreement.
- (13) From and after the Firm Occupancy Date, the Purchaser shall:
 - (a) maintain the Purchased Home and any exclusive use common elements in a clean and sanitary condition as would a prudent owner;
 - (b) comply with the proposed Condominium Documents; and
 - (c) not make any alterations or additions to the Purchased Home or any exclusive use common elements without the prior written approval of the Vendor, which approval may be unilaterally and arbitrarily withheld or granted on terms and conditions satisfactory to the Vendor in its sole and unfettered discretion.
- No work shall be done by, or for the Purchaser in or to the Purchased Home prior to the Closing Date ("Unauthorized Work"). The Purchaser acknowledges that a breach of this condition constitutes a trespass and entitles the Vendor, at its sole option, to take any of the following actions: i) declare the Purchaser to be in breach of this Purchase Agreement whereupon the Vendor shall be entitled to exercise any rights that it may have pursuant to this Purchase Agreement or at law as a result of the same, ii) finish the Unit to the extent possible, as determined by the Vendor in its sole and unfettered discretion, without regard to possible damage to the Unauthorized Work and without incurring any additional expense as a result of the Unauthorized Work, iii) remove and/or repair the Unauthorized Work, and any other portion of the Purchased Home thereby affected, and to receive compensation therefore as an adjustment on the Closing Date in an amount to be determined by the Vendor in its sole and unfettered discretion, including any compensation for time lost by the delay resulting from the Unauthorized Work and the Purchaser acknowledges that the Unauthorized Work may cause to be void the whole or any part of the warranty provided by Tarion. Further, the Purchaser acknowledges that the Vendor's warranty of workmanship is rendered invalid insofar as it relates to matters affected by the Unauthorized Work.
- (15) On or before the Firm Occupancy Date, the Purchaser shall deliver 12 post-dated cheques (or such greater number as the Vendor may require) to the Vendor in payment of the Occupancy Fee for the first 12 full months (or more) of Occupancy together with a certified cheque for the Occupancy Fee with respect to the period between the Firm Occupancy Date and the last day of the month in which the Firm Occupancy Date occurs. The Purchaser shall pay the monthly Occupancy Fee to the Closing Date and the Vendor shall return all unused post-dated Occupancy Fee cheques or destroy same, as determined by the Vendor in its sole and unfettered discretion.
- (16) The Vendor cannot guarantee (and will not be responsible for) the arrangement of a suitable move-in time for purposes of accommodating the Purchaser's occupancy of the Purchased Home on the Firm Occupancy Date or the Closing Date, as appropriate (or any acceleration or extension thereof as hereinbefore provided), and the Purchaser shall be solely responsible for directly contacting the Vendor's customer service office or property management office in order to make suitable booking arrangements with respect to the Condominium's service elevator, if applicable (with such booking being allotted on a "first come, first served" basis), and under no circumstances shall the Purchaser be entitled to any claim, refund, credit, reduction/abatement or set-off whatsoever against any portion of the Purchase Price of the Purchased Home, or against any portion of the monthly Occupancy Fees so paid or payable, if applicable) as a result of the service elevator not being available to accommodate the Purchaser moving into the Condominium on (or within any period of time after) the Firm Occupancy Date or the Closing Date, as appropriate.
- (17) The Purchaser shall be allowed to remain in Occupancy of the Purchased Home so long as the terms of both the Occupancy Arrangement and this Purchase Agreement have not been breached by the Purchaser. Upon any default by the Purchaser of the provisions of this Section 3.4 or other default under this Purchase Agreement, and failure of the Purchaser to remedy such default within 48 hours of written notice thereof to the Purchaser, in addition to the remedies available pursuant to the provisions in Section 8.1 of this Schedule, the Vendor may by written notice require the Purchaser to vacate the Purchased Home within 7 days from the date of written notice by the Vendor. The Purchaser shall bear all costs and expenses incurred by the Vendor to obtain vacant possession of the Purchased Home and all costs to redecorate and repair the Purchased Home as are required by the Vendor in its sole and unfettered discretion.
- (18) The Purchaser agrees to indemnify the Vendor for all losses, costs and expenses incurred as a result or by reason of:

- (a) the Purchaser's neglect, damage or use of the Purchased Home or the Condominium by either the Purchaser, the Purchaser's invitees or those for whom the Purchaser is in law responsible, or injury to any person or property in or upon the Purchased Home or the Condominium either arising from or out of this Purchase Agreement, or any occurrence in, upon or at the Condominium, or the Purchased Home, or the occupancy or use of the Condominium or any part thereof or the Purchased Home, or occasioned wholly or in part by any act or omission of the Purchaser, the Purchaser's invitees or those for whom the Purchaser is in law responsible; and
- (b) the Vendor's steps taken to obtain vacant possession of the Purchased Home upon a default by the Purchaser hereunder. The Purchaser agrees that should the Vendor elect to repair or redecorate all or any part of the Purchased Home or the Condominium as a result of the Purchaser's neglect, damage or use of the Purchased Home or Condominium, the Purchaser will immediately reimburse the Vendor for the cost of doing same. The determination of need for such repairs or redecoration shall be at the sole and unfettered discretion of the Vendor, and such costs may be added to the Purchase Price.
- (19) Except as set forth in the Addendum, the Vendor shall have the right to claim and be reimbursed by the Purchaser on a demand basis for any monies paid, or for security given, and all costs (including legal fees and disbursements on a full indemnity basis), together with interest thereon at 5% over the Prime Rate (both calculated and payable monthly), for any obligation of the Purchaser under this Purchase Agreement including Extras, damages or Occupancy Fees, notwithstanding that the purchase and sale aspect of this transaction may have been terminated for any reason whatsoever.
- (20)The Vendor shall not be liable for any death or injury arising from or out of any occurrence in, upon, at, or relating to the Condominium, or the Purchased Home or damage to property of the Purchaser or of others located on the Condominium or the Purchased Home, nor shall it be responsible for loss of or damage to any property of the Purchaser or others from any cause whatsoever, even if any such death, injury, loss or damage results from the negligence of the Vendor, its agents, servants or employees, or other persons from whom it may at law be responsible. Without limiting the generality of the foregoing, the Vendor shall not be liable for any injury or damage to persons or property resulting from fire, explosion, falling plaster, steam, gas, electricity, water, rain, flood, snow or leaks from any part of the Condominium or the Purchased Home or from the pipes, appliances, plumbing, works, roof or subsurface of any floor or ceiling or from the street or any other place or by dampness or by any other cause whatsoever. The Vendor shall not be liable for any such damage caused by other owners, occupants, tenants, invitees or any other person on the Condominium or by occupants of adjacent property thereto, or the public, or caused by construction or any private, public or quasi public work. All property of the Purchaser kept or stored on the Condominium or on or in the Purchased Home shall be kept or stored at the risk of the Purchaser only and the Purchaser shall indemnify the Vendor and save it harmless from and against any claim arising out of any damages to the same, including any subrogation claims by the Purchaser's insurers.
- (21) On or before the Firm Occupancy Date and at the option of the Vendor, the Purchaser shall execute and deliver to the Vendor any documents, directions, acknowledgements, assumption agreements or any and all other documents required by the Vendor pursuant to this Purchase Agreement, in the same manner as if the Closing of the transaction were taking place at that time.
- (22) The Purchaser agrees that the relationship of landlord and tenant shall not be created by the Occupancy of the Purchased Home as provided herein.
- (23) If any part of the Condominium is damaged before the Creating Documents are registered, the Vendor may in its sole and unfettered discretion either:
 - (a) make such repairs as are necessary to complete this transaction and, if necessary, delay the Firm Occupancy Date in the manner permitted under the Addendum and the Occupancy Fee shall abate during the period the Residential Unit is uninhabitable;
 - (b) terminate this Purchase Agreement and return to the Purchaser all deposit monies paid by the Purchaser to the Vendor, with interest payable under law if the damage to the Condominium has frustrated this Purchase Agreement at law; or
 - (c) apply to a court of competent jurisdiction for an order terminating the Purchase Agreement in accordance with the provisions of subsection 79(3) of the Condo Act,

it being understood and agreed that all insurance policies and the proceeds thereof are to be for the benefit of the Vendor alone.

- (24) The Purchaser agrees to permit the Vendor, its employees, agents and contractors to have access to the Purchased Home without notice to the Purchaser, as may be required from time to time to effect repairs to or complete the construction of the Purchased Home, other Units or the common elements, or for after-sales service purposes.
- (25) Keys will be released to the Purchaser at the construction site or the customer service office of the Vendor, as the Vendor in its absolute discretion determines, upon completion of this transaction on the Firm Occupancy Date or the Closing Date, as the case may be, unless otherwise specifically agreed in writing between the Vendor and the Purchaser. The Purchaser agrees that the Vendor's advice that keys are available for release to the Purchaser constitutes a valid tender of keys on the Purchaser. Upon completion of this transaction, if the Purchaser fails to attend to pick up the keys by five (5:00) p.m. on that day, the Vendor may retain the keys and release same to the Purchaser on the next Business Day.
- (26)If, after thirty-six (36) months from the Firm Occupancy Date, the Vendor for any reason whatsoever is unable to register the Creating Documents and therefore is unable to deliver a registrable transfer/deed to the Purchaser, the Vendor shall have the right to give sixty (60) days' written notice to the Purchaser of the Vendor's intention to terminate the Occupancy Arrangement and this Purchase Agreement. If the Purchaser consents to such termination in writing, the Purchaser shall give up vacant possession on such date and pay the Occupancy Fee to such date, after which this Purchase Agreement and the Occupancy Arrangement shall be terminated and the Deposit shall be returned to the Purchaser (excluding uncashed cheques) and the Purchaser shall sign a written acknowledgement confirming the termination and amount of monies refunded together with interest required by the Condo Act, subject however, to any repair and redecorating expenses of the Vendor necessary to restore the Purchased Home to its original state prior to Occupancy, reasonable wear and tear excepted. The Purchaser and Vendor each agree to provide a release of this Purchase Agreement in the Vendor's standard form. If the Purchaser does not consent to such termination, the provisions of subsection 79(3) of the Condo Act may be invoked by the Vendor.

PART IV - TITLE AND REQUISITIONS

4.1 <u>TITLE</u>

Title shall be good and free from all encumbrances, except for Permitted Encumbrances, any restrictions, easements, conditions or covenants registered on title, restrictions, conditions and covenants set out in the Condominium Documents and except as otherwise provided in this Purchase Agreement. Title shall be examined by the Purchaser at the Purchaser's own expense and the Purchaser shall not call for the production by the Vendor of any surveys, title deeds or abstracts, proof or evidence of title or to have any copies thereof. The Purchaser shall be allowed until 15 days prior to the Closing Date to examine the title at the Purchaser's own expense, and if within that time the Purchaser shall furnish the Vendor in writing with any valid objection to the title which the Vendor shall be unwilling or unable to remove and which the Purchaser will not waive, this Purchase Agreement, notwithstanding any intermediate acts or negotiations in respect of such objections, shall be at an end, any Occupancy Arrangement shall be terminated effective the last day in the month following the month in which this Purchase Agreement shall be at an end and all monies previously paid by the Purchaser to the Vendor shall be returned (except for Occupancy Fees which the Purchaser acknowledges that the Vendor has no obligation to return), together with Interest and the Vendor shall have no further liability or obligation hereunder and shall not be liable for any costs or damages. Save as to any valid objection so made within such time, the Purchaser shall be conclusively deemed to have accepted the title of the Vendor to the Purchased Home. The Purchaser acknowledges and agrees that the Vendor shall be entitled to respond to some or all of the requisitions submitted by or on behalf of the Purchaser through the use of a standard title memorandum or title advice statement by the Vendor's Solicitors, and that the same shall constitute a satisfactory manner of responding to the Purchaser's requisitions.

4.2 PERMITTED ENCUMBRANCES

- (1) The Purchaser agrees to accept title to the Purchased Home, the Lands and the Condominium, subject to the following ("Permitted Encumbrances") and agrees to comply with and abide by all of the terms and provisions thereof, and, shall not require any releases of same or discharges with respect thereto, or, except as provided in the Addendum, evidence (written or otherwise) that same have been complied with or are in good standing, namely:
 - (a) the provisions, covenants, restrictions, easements and conditions contained in the Condominium Documents, notwithstanding that they may be amended from the proposed Condominium Documents disclosed to the Purchaser along with the Disclosure Statement;
 - (b) any registered or unregistered subdivision, financial, security, housekeeping, development, site plan, Section 37, Section 111, distance limiting or servicing agreements, or any other agreements which may be necessary with Approving

Authorities or other authority having jurisdiction over the Purchased Home, the Lands or the Condominium, or any municipal by-laws or regulations affecting the Purchased Home, the Lands or the Condominium; any restrictions or warnings on title imposed by any Approving Authorities related to the site conditions, notices and warnings as more particularly described in Section 6.3 of this Schedule;

- any transfers, easements, rights, rights-of-way, encroachment agreements, (c) existing encroachments from nearby properties, conditions or covenants that run or may run with the Lands, licenses or agreements whatsoever, including such as are necessary or required for the installation and maintenance of public or any other utilities (including the Utility Services) and including telephone, hydro, gas, sewer, water, cable television, Smart Home Services, Smart Metering, and internet facilities and for such other services as may otherwise be required by the Vendor to serve the Purchased Home or any other part or parts of the Condominium or other components of the Galleria Master Development or the Galleria Master Development Lands or to or for the benefit of the Lands or lands adjoining or in the vicinity of the Lands or which may be required by the Vendor, the owner of the Lands or any owners of adjoining lands or lands in the vicinity of the Lands for servicing and/or access to, or entry from, such properties (including all easements, rights of way and/or agreements for access, service, support, swing crane rights (if any), tie-backs, shoring, amenities, recreational and shared facilities, cost-sharing and the like, for and/or with adjacent or neighbouring property owners);
- (d) any agreements which may be necessary for the maintenance, repair, development, servicing and operation of the Condominium and the administration of the affairs and carrying out of the duties and obligations of the Corporation, including an insurance trust agreement and a management agreement;
- (e) any agreements with any Approving Authority, which may be necessary for the maintenance, repair or servicing of any easements in favour of the Lands;
- (f) any leases, service, or maintenance contracts and license rights to occupy portions of the common elements, if any, which are in accordance with the Condominium Documents or the Condo Act and any easements, leases or licenses of portions of the common elements to the declarant and/or the Vendor or any of its affiliates and/or third parties;
- (g) any rights, licences or easements which now or may hereafter be required for vehicular or pedestrian ingress and egress with respect to the Condominium, the Galleria Master Development or the Galleria Master Development Lands or any adjoining property or neighbouring lands;
- (h) the mortgages not being assumed as provided in Section 4.3 of this Schedule;
- (i) the Official Plan and zoning by-laws of Approving Authorities;
- (j) any encroachment agreements with respect to the Lands;
- (k) any agreements or easements in respect of the supply of television, telephone, internet or other telecommunications services which may be necessary in order to provide access to any telecommunications service provider (including Rogers Cable and Bell Canada), or any other telecommunications services providers or similar entities, with respect to services to the Condominium, the Galleria Master Development or the Galleria Master Development Lands or the Purchased Home;
- (I) any agreements or easements in favour of any Approving Authority which may be necessary for fire and emergency vehicular access to the Lands or any neighbouring lands in the vicinity of the Condominium;
- (m) any easements in favour of a any electricity distribution company or local utility provider (including any Utility Service Provider, Smart Metering Provider or Smart Home Services Provider) or any of their subcontractors, successors and assigns which may be necessary for the purposes of conducting inspections, maintenance, repair or the reading of Smart Meters, Smart Home Services or any other electrical devices related thereto and other components of a Utility Service Provider's, Smart Metering Provider's or Smart Home Services Provider's systems;
- (n) the Reciprocal Agreements or any other reciprocal agreements as may be required by one or more persons having an interest in any property adjacent to, or neighbouring the Lands, as same may be modified, amended, extended or restated from time to time, including all easements and rights of way provided for therein;
- (o) any insurance trust agreement entered into by the Condominium or as required pursuant to the provisions of the Reciprocal Agreements;

- (p) Instrument No. WH69572 registered March 21, 1951, being a Transfer Easement in favour of The Corporation of the City of Toronto;
- (q) Instrument No. AT628502 registered October 15, 2004, being a Notice Under Section 71 of the Land Titles Act by the Applicants, Ontario Potato Distributing Inc. and Harzuz Holdings Limited;
- (r) Instrument No. AT3974021 registered August 11, 2015, being an LR's Order by the Land Registrar, Toronto Land Registry Office;
- (s) Instrument No. AT3978439 registered August 14, 2015, being a Transfer from Harzuz Holdings Limited and Ontario Potato Distributing Inc. to 2470347 Ontario Inc.;
- (t) Instrument No. AT5148836 registered May 31, 2019, being a LRs Order by the Land Registrar, Toronto Land Registry Office;
- (u) Instrument No. AT5185238 registered July, 15, 2019 being the Application to Annex Restrictive Covenants S.118 by the Applicant, 2470347 Ontario Inc.; and
- (v) any other easements, restrictions, covenants or agreements which, in the opinion of the Vendor, do not materially or unreasonably interfere with the use and enjoyment of the Purchased Home as a personal residence.
- (2) The Purchaser undertakes and agrees to: (i) comply with all Permitted Encumbrances, including all regulations, covenants and restrictions contained in any subdivision, site plan or other development agreements pertaining to the Purchased Home, the Galleria Master Development and/or the Galleria Master Development Lands; and (ii) execute any agreement or agreements, prior to or after the Closing Date, upon request, necessary: A. for the operation, development, maintenance, repair or servicing of the Galleria Master Development or the Galleria Master Development Lands; B. to grant a right, licence or easement over any part of the Condominium for any of the aforementioned purposes set out in this Section 4.2; and C. to consent to any of the matters referred to in this Section 4.2, and to execute any documents required in connection therewith, including a covenant or agreement in favour of any third party designated by the Vendor, to comply with or be bound by any of the matters referred to therein.
- (3) Subject to the requirements of the Addendum, the Purchaser shall satisfy itself as to the due compliance with the provisions of any such agreements, instruments, restrictions, covenants or matters listed in Section 4.2(1) of this Schedule.
- (4) Whether before or after the Closing Date, the Purchaser agrees to assume or accept, or permit or grant or consent to whatever rights of easement, licences and/or rights of way as shall be required for hydro, gas, water, sanitary and storm sewers, fuel, telephone, cable television, internet, municipal or other services and utilities for the Galleria Master Development, the Galleria Master Development Lands or other lands in the vicinity of the Lands, provided that such do not materially and adversely affect the Purchaser's use of the Purchased Home and the Purchaser agrees to execute all documents and do all things required for such purposes. If it shall be deemed necessary or expedient in the opinion of the Vendor after any transfer of the Purchased Home by the Vendor to the Purchaser, the Purchaser shall execute all documents (without payment by the Vendor) which may be required to convey or confirm such easements, licences and/or rights and the Purchaser agrees to extract a similar covenant in favour of the Vendor from any subsequent purchaser or transferee of the Purchased Home forthwith upon the written request of the Vendor.
- (5) The Purchaser also acknowledges that the wires, cables and fittings comprising any cable television system, telephone system or Smart Home Services serving the Property are, or shall be, owned by the local cable television supplier, local telephone supplier or Smart Home Services Provider as the case may be, and in the event that it requires any rental contract pertaining to the leasing of its chattels or equipment to be registered on title to the Purchased Home, then the Purchaser agrees to accept title subject to the same.
- (6) The Purchaser acknowledges that the Corporation and the Vendor will be required to enter into agreements with Utility Service Providers, Smart Metering Providers or Smart Home Services Providers regarding the system of Utility Services distribution, Smart Metering and Smart Home Services in the Condominium. The Purchaser also acknowledges and agrees that the Purchaser may be required to enter into any agreements with one or more Utility Service Providers and/or Smart Metering Providers and/or or Smart Home Services Providers in order to acknowledge and accept the system of Utility Service distribution, Smart Metering or Smart Home Services to the Properties.
- (7) In the event that the Vendor is not the registered owner of the Lands or the Purchased Home, the Purchaser agrees to accept a conveyance of title from the registered owner together with the owner's title covenants in lieu of the Vendor.

4.3 MORTGAGES NOT BEING ASSUMED

The Purchaser shall complete this transaction in accordance with this Purchase Agreement notwithstanding that on the Firm Occupancy Date and/or the Closing Date the title of the Condominium and/or the Purchased Home may be encumbered by a mortgage or mortgages not intended to be assumed by the Purchaser. The Vendor shall not be obliged to obtain and register any partial discharge or discharges of such mortgage or mortgages insofar as the Purchased Home or the Condominium is affected until after the Closing Date. The Purchaser agrees to accept the undertaking of the Vendor to obtain and register partial or full discharges of any mortgages affecting the Purchased Home within a reasonable time after the Closing Date so long as the Vendor delivers to the Purchaser on Closing an acknowledgement by the mortgagee either that:

- (a) the mortgage being assumed or arranged on Closing, if any, has replaced the mortgage not intended to be assumed, and that a partial or full discharge thereof will be available within a reasonable time; or
- (b) a partial or a full discharge of the mortgage will be available upon payment of a fixed amount to the mortgagee on Closing and the Vendor undertakes to deliver such amount to the mortgagee and to cause the Vendor's Solicitors to register the partial discharge of the mortgage with respect to the Purchased Home upon receipt thereof and to advise the Purchaser or the Purchaser's Solicitors of the registration particulars; or
- (c) a partial or a full discharge of the mortgage will be available without any action on the part of the Vendor.

4.4 POSTPONEMENT

This Purchase Agreement and the Purchaser's rights hereunder are and shall remain subordinate to, and the Purchaser hereby postpones such rights to, (i) any mortgage or charge of the Lands now in place or which may come into existence at any time in the future and any advances from time to time thereunder, (ii) any agreements entered or to be entered into by the Vendor at any time, with any public utility (which may include a public Utility Service Provider) or any Approving Authority or other authority having jurisdiction relating to the development and/or servicing of the Condominium or the Lands, (iii) the Creating Documents; and (iv) any other title matter listed in Section 4.2(1) of this Schedule.

PART V- ADJUSTMENTS

5.1 CLOSING AND ADJUSTMENTS

It is agreed that on the Closing Date:

- (1) the Vendor shall deliver to the Purchaser a transfer of title to the Purchased Home prepared at the expense of the Vendor. If required to do so by the Vendor, the Purchaser shall execute the transfer which, at the Vendor's option, may incorporate any or all of the provisions of this Purchase Agreement;
- (2) the Purchaser shall deliver to the Vendor: (i) a series of 12 post dated cheques commencing on the 1st day of the month following the Closing Date payable to the Corporation to cover the Purchased Home's contribution to the Common Expenses payable by the Purchaser for the first 12 full months following the Closing Date; (ii) a certified cheque payable to the Corporation for one (1) month's Common Expenses to be credited to the reserve fund or as working capital of the Corporation; (iii) any documents that may be required by the Vendor or the Vendor's Solicitors pursuant to the *Family Law Act* (Ontario), as amended from time to time, or similar legislation; and (iv) any other documents and assurances required by the Vendor or the Vendor's Solicitors, in their sole and unfettered discretion, to give full effect to the provisions of this Purchase Agreement;
- (3) any proper and necessary adjustments shall be made with respect to the Occupancy Fee, including in respect of the month in which Closing occurs;
- (4) insurance premiums, realty taxes (including any supplementary assessment with respect thereto, any local improvement charges, sewer impost charges and any fee charged by any Approving Authority to open a tax account or assessment roll number for the Purchased Home), any new taxes applicable to the Purchased Home which are not yet exigible as of the date of acceptance of this Purchase Agreement, hydro, cable T.V., water, gas rates and fuel (except insofar as the same are included in Common Expenses), Common Expenses and that portion of the reserve fund that is attributable to the Purchased Home shall be adjusted to the Closing Date, with the Closing Date itself being apportioned to the Purchaser. With respect to realty taxes (including local improvement charges and sewer impost charges), the same may, in the Vendor's sole and unfettered discretion, be estimated by the Vendor as if the Purchased Home had been assessed and taxed by the relevant taxing authority as fully completed for the calendar year in which the Firm Occupancy Date and/or the Closing Date falls, and shall be adjusted as if such sum had been paid by the

Vendor notwithstanding that the same may not, by the Firm Occupancy Date or the Closing Date (as the case may be), have been assessed, levied or paid, subject however to readjustment upon the actual amount being ascertained;

- (5) the Purchaser shall reimburse the Vendor for (i) the cost of enrolment of the Purchased Home under The ONHWP Act, including taxes; (ii) an amount equal to the real estate transaction insurance levy surcharge imposed by The Law Society of Ontario on the Vendor's portion of this transaction, plus HST; (iii) legal fees in respect of the holding and administering of the Deposit by the Vendor's Solicitors and in respect of the Vendor's Solicitors' holding and administering of monies paid on account of Extras, which legal fees shall not exceed the sum of \$70.00 per cheque processed, plus HST; (iv) the amount of any special assessments levied by the Condominium against the Purchased Home at any time following the turnover meeting respecting the Condominium held pursuant to Section 43 of the Condo Act; (v) legal fees payable for each discharge from the Purchased Home of any of the Vendor's blanket mortgages, charges or debentures registered on the Lands (whether as part of a complete discharge or a partial discharge) registered at any time after the registration of the Creating Documents, whether prior to or after Closing, such legal fees not to exceed \$275.00, plus HST, per discharge, per Purchased Home; (vi) a portion of the cost incurred by the Vendor in respect of the site review conducted pursuant to Tarion Builder Bulletin 19R and all charges relating to, incidental to or incurred in connection with such review, which portion of cost shall be determined by the Vendor, based upon the proportionate share of Common Expenses for the Purchased Home as set out in the Creating Documents, plus HST; (vii) the cost of the Vendor's Solicitors' increased legal fees to electronically register documents as a result of compliance with Rule 3.4-16.7 of the Law Society of Ontario's Rules of Professional Conduct in connection with the closing of this transaction in the amount of \$275.00, plus HST; (viii) legal fees for any electronic document exchange service utilized by the Vendor's Solicitors in connection with Occupancy or Closing, such fee not to exceed \$75.00, plus HST, for each of Occupancy and Closing; and (ix) a charge of \$90.00, plus HST, for each Purchased Home to cover the Vendor's legal fees in respect of monthly financial reporting to its surety and/or construction lender; (x) the amounts attributed to the Purchased Home by the Vendor, based upon the proportionate share of Common Expenses for the Purchased Home as set out in the Creating Documents, plus HST, in respect of any payment for an obligation imposed by the City of Toronto or any other Approving Authority in connection with or in any way related to the Condominium or the Galleria Master Development pursuant to any agreement (including any "Section 37" agreement, land exchange agreement, contribution agreement or any other agreement) with the City of Toronto or any other Approving Authority, pursuant to Section 37 of the Planning Act, R.S.O. 1990, c. P. 13, pursuant to the Official Plan of the City of Toronto or otherwise, including: A. any parkland contribution (or cash-in-lieu of same); B. any art levy; and C. funds for any community benefit.
- (6) the Purchaser shall pay to the Vendor any and all costs or charges in respect of any Smart Metering, Utility Services and Smart Homes Services including any Smart Metering Charges, Utility Service Charges or Smart Home Services Charges which are charged to the Vendor, the cost of all Smart Meters and Smart Home Services installed in the Property or in respect of the Property including the cost of the installation of all such Smart Meters or Smart Home Services or in the event that the Smart Meters or Smart Home Services are leased, the lease payments and any other associated fees, the payment of any security deposit required by any Smart Metering Provider, Utility Service Provider or Smart Home Services Provider, Utility Service Provider for the Property;
- (7) the Purchaser may be responsible for the payment and costs associated with the use and consumption of Utility Services and Smart Metering directly with the Utility Service Provider as measured by or provided through any Smart Meters installed in or in respect of the Property. Any and all expenses and costs associated with such Utility Services or Smart Metering will be paid for directly by the Purchaser;
- (8) the Vendor and the Purchaser shall pay the costs of registration of their own documents and any tax payable in connection therewith, including any land transfer tax or non-resident speculation tax;
- (9) adjustment in favour of the Vendor shall be made in respect of any legal fees and disbursements which the Vendor or Vendor's Solicitors may charge to the Purchaser's Solicitors for not utilizing TERS pursuant to Section 9.5(3) of this Schedule;
- (10) the Purchaser shall pay or reimburse the Vendor for the amount of any increase in any development charge and/or education development charge or levy assessed against or attributable to the Purchased Home, paid or payable by the Vendor pursuant to the Development Charges Act (Ontario), as amended from time to time, the Education Act (Ontario), as amended from time to time, or pursuant to any other similar legislation, or any other charges or levies payable to any Approving Authority, provided that the imposition of said charges or levies have increased the cost of the development of the Purchased Home or the Condominium on the Lands over and above the costs associated therewith as of September 10, 2019, such costs to be determined as if the Vendor were to have applied for

and obtained a building permit for the Condominium on September 10, 2019 and as if the Vendor were to have paid all applicable development charges, education development charges or other such charges on that date;

- (11) if there are chattels involved in this transaction, the allocation of value of such chattels shall be estimated where necessary by the Vendor and HST thereon shall be paid by the Purchaser and collected and remitted by the Vendor;
- (12) the Purchaser shall accept the Vendor's covenant of indemnity regarding any lien claims which are the responsibility of the Vendor in full satisfaction of the Purchaser's rights under the *Construction Act* (Ontario), as amended from time to time, and shall not claim any lien holdback;
- (13) an administrative fee of \$250.00, plus HST, where applicable, shall be charged to, and be payable by, the Purchaser for any cheque delivered by the Purchaser to the Vendor or the Vendor's Solicitors and not accepted by the Vendor's bank (in the case of the Vendor) or the Vendor's Solicitors' bank (in the case of the Vendor's Solicitors) for any reason;
- (14) an administrative fee shall be charged to, and be payable by, the Purchaser to the Vendor's Solicitors for any modifications requested to the documents for Closing or Occupancy once issued or prepared by the Vendor's Solicitors or for any changes necessitated by information not provided by the Purchaser to the Vendor's Solicitors prior to the issuance of such documents (including information in respect of matters related to the Rebate). Provided, however, that the foregoing administrative fee shall not exceed \$500.00, plus HST, where applicable. For greater certainty, the Purchaser acknowledges that the Vendor and the Vendor's Solicitors shall have no obligation to approve any such request;
- (15) the amount of \$100.00, inclusive of HST, for legal fees incurred in connection with the preparation and delivery of a Status Certificate on or before the Closing Date, shall be paid to the Vendor;
- (16) an amount representing interest on the balance of the Purchase Price calculated at the Vendor's borrowing rate from its principal banker from the Closing Date to and including the next following banking date shall be paid to the Vendor;
- (17) Vendor's Solicitors' reasonable fees in respect of the preparation of any security for the Purchaser's financing from the Vendor's Lender if Section 2.2(3) of this Schedule applies, shall be paid to the Vendor; and
- (18) all other adjustments provided for in this Purchase Agreement or any written amendment hereto, shall be adjusted to the Closing Date. For certainty, for any adjustment to the Closing Date referred to in this Agreement, the Closing Date itself shall be apportioned to the Purchaser.

5.2 HARMONIZED SALES TAX

The Purchaser and Vendor acknowledge and agree that the Purchase Price includes a component equivalent to the harmonized sales tax exigible on the purchase of the Purchased Home less the Rebate as defined below ("HST") payable pursuant to the HST Act. The Purchaser shall, at the Purchaser's own cost and expense, be responsible for payment of all taxes including the tax exigible pursuant to the HST Act and any transaction tax, value added tax, sales, use, transfer tax or non-resident speculation tax and any increase in the rate of such taxes imposed by any of the Government of Canada, Government of Ontario or City of Toronto, on all amounts payable under this Purchase Agreement, including without limitation for Extras. The actual consideration for the Purchased Home, exclusive of Extras, changes, upgrades or adjustments as herein provided, is the amount derived by subtracting the HST payable herein (less all refunds, credits and rebates, if any, including the Rebate, available to the Purchaser pursuant to the HST Act, if any) from the Purchase Price. The Purchaser covenants and agrees to pay the HST exigible on the purchase of the Purchased Home as required by the HST Act and the Vendor shall remit the HST to the Canada Revenue Agency on behalf of the Purchaser following the Closing of the transaction.

5.3 HST REBATE

In consideration of the Purchase Price being inclusive of HST, the Purchaser hereby irrevocably assigns to and in favour of the Vendor any and all rights the Purchaser may have on Closing or thereafter to any rebates, refunds or credits available with respect to this transaction pursuant to the HST Act, including if the Purchaser qualifies for it, the HST new housing rebate or any other rebates, refunds or credits of or pertaining to HST that are applicable pursuant to the HST Act (the "Rebate"). If the Purchaser qualifies for the Rebate, and if the Rebate is available for the transaction contemplated herein, the Purchaser acknowledges having received credit for the amount of the Rebate in the calculation of the Purchase Price, and having assigned the Rebate to and in favour of the Vendor, and directs the Vendor to so indicate in any documentation pertaining to the Rebate. The Purchaser represents and warrants to the Vendor that the Purchaser has not claimed and shall not at any time claim for the Purchaser's own account any part of the Rebate in connection with the

Purchaser's acquisition of the Purchased Home, save and except as expressly provided in this Purchase Agreement. To the extent that the Rebate is credited to the Purchaser, the Purchaser hereby releases all of the Purchaser's claims or interests in and to the Rebate, to and in favour of the Vendor and hereby irrevocably authorizes and directs the Canada Revenue Agency to pay or credit the Rebate directly to the Vendor.

- (2) Subject to Section 5.3(4) below, the Purchaser covenants and agrees that the Purchaser shall forthwith at and during Occupancy and following the completion of this transaction, personally occupy the Purchased Home, or cause one or more of the Purchaser's relations (as that term is defined in the HST Act) to occupy the Purchased Home, as their primary place of residence for such period of time as shall be required by the HST Act in order to entitle the Purchaser to the Rebate.
- (3) Subject to Section 5.3(4) below, the Purchaser covenants and agrees to execute and deliver to the Vendor on or before the Closing Date all applications, assignments, authorizations, directions, statutory declarations, forms and other such documents as the Vendor or the Vendor's Solicitors shall request, from time to time, in order to verify entitlement to the Rebate and to effect the proper assignment thereof to the Vendor, including the New Housing Application for Rebate of the Harmonized Services Tax Form as prescribed from time to time, and including compliance with the Purchaser's covenants contained in Sections 5.3(2) and 5.3(4) of this Schedule.
- In the event that the Purchaser shall for any reason fail to qualify for the Rebate as assigned (4) to the Vendor, the Purchaser shall indemnify and hold the Vendor harmless in the amount that the Purchaser would have been entitled to had the Purchaser so qualified for the Rebate and shall pay such amount to the Vendor forthwith upon written demand for same and the Vendor shall be entitled to a vendor's lien on the Purchased Home in an amount equal to the Rebate and the Purchaser does hereby charge and pledge the Purchaser's interest in the Purchased Home in the amount of the Rebate with the intention of creating a lien or charge as security for this indemnity. In addition, in the event that the failure of the Purchaser to qualify for the Rebate is disclosed by the Purchaser to the Vendor on or before the Closing Date, or in the event that the Purchaser fails to provide evidence satisfactory to the Vendor in its sole and unfettered discretion, confirming the Purchaser's entitlement to the Rebate, the Vendor shall be credited on the statement of adjustments, either on the Firm Occupancy Date or on Closing, as determined by the Vendor, with an amount equal to the Rebate, had the Purchaser qualified for same and the Purchaser shall be relieved of the Purchaser's covenant under Section 5.3(2) of this Schedule.
- (5) The Purchaser further acknowledges and agrees and notwithstanding anything to the contrary as may be contained in this Purchase Agreement or otherwise, in the event that any applicable rates of HST increase, the Purchaser shall be responsible for such increase and shall pay same as an adjustment on the Closing Date, less any Rebate, if applicable, of such increased amounts as assigned to the Vendor on similar terms as set out above. If any applicable rates of HST decrease, the Purchaser shall not be entitled to any abatement or reduction of the Purchase Price.
- (6)If the Vendor believes for whatever reason, in its sole and unfettered discretion based on any source of information, including any information provided by the Purchaser to any governmental authority in respect of any land transfer tax or non-resident speculation tax, that the Purchaser does not qualify for the Rebate, regardless of any documentation provided by or on behalf of the Purchaser (including any statutory declaration sworn by the Purchaser) to the contrary, and the Vendor's belief or position on this matter is communicated to the Purchaser or the Purchaser's Solicitors on or before the Closing Date, then notwithstanding anything hereinbefore or hereinafter provided to the contrary, the Purchaser shall be obliged to pay to the Vendor (or to whomsoever the Vendor may in writing direct), by certified cheque delivered on Closing, an amount equivalent to the Rebate, in addition to the outstanding balance of the Purchase Price, and in those circumstances where the Purchaser maintains that the Purchaser is eligible for the Rebate despite the Vendor's belief to the contrary, the Purchaser shall (after payment of the amount equivalent to the Rebate as aforesaid) be fully entitled to file the necessary rebate forms directly with, and pursue the Rebate directly from, Canada Revenue Agency. It is further understood and agreed that in the event that the Purchaser intends to rent out the Purchased Home after Closing only, then the Purchaser shall not be entitled to the Rebate, but may nevertheless be entitled to pursue, on his or her own after the Closing Date, the federal and provincial new residential rental property rebate directly with Canada Revenue Agency.
- (7) Notwithstanding any other provision herein contained, the Purchaser acknowledges and agrees that the Purchase Price does not include HST exigible with respect to any of the adjustments payable by the Purchaser pursuant to this Purchase Agreement, or any Extras, changes or upgrades to the Purchased Home agreed upon by the Vendor and Purchaser following the acceptance by the Vendor of this Purchase Agreement, and the Purchaser covenants and agrees that it shall be responsible for payment of HST to the Vendor for such adjustments, Extras, changes or upgrades in accordance with the HST Act. The Vendor may adjust the Purchase Price to account for any of the adjustments payable by the Purchaser pursuant to this Purchase Agreement and such recalculated Purchase Price will appear on

the statement of adjustments. If the Vendor so elects, the Purchaser will at the Vendor's request acknowledge in writing the recalculated Purchase Price. In addition, and without limiting the generality of the foregoing, in the event that the Purchase Price is increased by the addition of Extras, changes, upgrades or adjustments and as a result of such increase, the quantum of the Rebate that would otherwise be available is reduced or extinguished (the quantum of such reduction being hereinafter referred to as the "Reduction"), then the Purchaser shall pay to the Vendor on the Firm Occupancy Date or the Closing Date (as determined by the Vendor in its sole and unfettered discretion) the Reduction.

5.4 OCCUPANCY ADJUSTMENTS

Notwithstanding any other provision of the Purchase Agreement, the Vendor may, in its sole and unfettered discretion, but shall not be obligated to, allow for payment of amounts for certain adjustments that might otherwise have been payable on the Firm Occupancy Date, to be paid on the Closing Date. Any such amounts will remain payable by the Purchaser and shall be paid by the Purchaser to the Vendor on Closing, in accordance with the provisions of Part V of the Purchase Agreement.

PART VI - CONSTRUCTION MATTERS

6.1 COMPLETION OF CONSTRUCTION

- The Purchaser agrees that the Purchaser or the Purchaser's Designate will meet with the (1) Vendor's representative prior to the Firm Occupancy Date for a pre-delivery inspection of the Purchased Home (the "PDI"), at a time designated by the Vendor on at least 7 days' written notice delivered to the Purchaser, the Purchaser's Designate or the Purchaser's Solicitors. Only the Purchaser or the Purchaser's Designate together with one other person may attend the PDI. At the time of the PDI, the Purchaser, the Purchaser's Designate or the Purchaser's Solicitors and the Vendor shall complete a single Certificate of Completion and Possession (the "CCP"), a Pre-Delivery Inspection Form (the "PDI Form") and a Confirmation of Receipt in the forms prescribed from time to time by, and required to be completed pursuant to The ONHWP Act, listing all outstanding, incomplete work or defective items in the Purchased Home. THE CCP AND THE PDI FORM SHALL CONSTITUTE THE VENDOR'S ONLY UNDERTAKING TO COMPLETE SUCH ITEMS. The Purchaser agrees that: (i) the incomplete and/or deficient items included in the CCP and the PDI Form shall represent the entire balance of work to be completed by the Vendor with respect to the Purchased Home and the Purchaser agrees that no further request for completion of items may be maintained by the Purchaser and this shall serve as good and sufficient release of the Vendor in that regard; (ii) the Vendor or its servants or agents shall have the right to enter upon the Purchased Home, the Condominium and the Lands after the Firm Occupancy Date and after the Closing Date in order to complete the said items; and (iii) the Vendor and any Approving Authority, or their servants or agents, shall have the right to enter upon the Purchased Home, the Condominium and the Lands after the Firm Occupancy Date and after the Closing Date to inspect, repair or complete any exterior work, to complete grading operations, to undertake the modification of the surface drainage, including the installation of catch basins or to effect any other corrective measures as may be required by any Approving Authority or the Vendor. The Vendor shall complete the Purchased Home and the common elements in accordance with the provisions of this Purchase Agreement, notwithstanding that the Purchaser shall have taken possession of the Purchased Home, within a reasonable time thereafter, subject to Unavoidable Delay. Delivery of the PDI Form and the CCP by the Purchaser as aforesaid to the Vendor shall not be regarded as acceptance by the Vendor that any items listed thereon are in fact incomplete, deficient or missing. In the event that the Vendor does any additional work in its discretion, other than as contemplated in this Section 6.1(1), it shall not be deemed to have waived the provisions of this Section 6.1(1) or otherwise enlarged its obligations hereunder. Failure of the Purchaser to complete the CCP, the PDI Form and the Confirmation of Receipt shall entitle the Vendor to complete same in accordance with Section 6.1(3) of this Schedule or in accordance with such Section to declare the Purchaser in default hereunder. In the event that the Purchaser does not appoint a Designate at the time of executing this Purchase Agreement, the Purchaser may appoint a Designate on or before the 15th day prior to the PDI, by contacting the Vendor and completing the Vendor's form for appointing a Designate.
- (2) The Purchaser acknowledges that:
 - (a) the "Homeowner Information Package" referred to in Tarion Builder Bulletin 42R (the "HIP") is available from Tarion. The Vendor agrees to deliver the HIP to the Purchaser or the Purchaser's Designate, at or before the PDI. The Purchaser, or the Purchaser's Designate agrees to execute and provide to the Vendor, the Confirmation of Receipt forthwith upon receipt of the HIP;
 - (b) Tarion requires that the Purchaser complete and submit to Tarion:
 - (i) a "30-Day Form" notifying Tarion of any outstanding warranty items within thirty (30) days of the Purchaser's possession of the Purchased Home;

- (ii) a "Year-End Form" notifying Tarion of any outstanding warranty items during the final thirty (30) days of the Purchaser's first year of possession of the Purchased Home; and
- (iii) a "Second-Year Form" notifying Tarion of any outstanding warranty items during the Purchaser's second year of possession of the Purchased Home,

and that failure to timely submit such forms may affect the Purchaser's ability to make claims under Tarion's new home warranty program.

- Subject to the Addendum, except as to items specifically listed in the CCP and the PDI (3) Form, the Purchaser shall be deemed, upon taking possession of the Purchased Home, to have acknowledged that the Purchased Home has been completed in accordance with this Purchase Agreement and the Purchaser shall be conclusively deemed to have accepted the Purchased Home. If the Purchaser or the Purchaser's Designate does not attend and execute a CCP, PDI Form and Confirmation of Receipt prior to the original Firm Occupancy Date or any extended Firm Occupancy Date as aforesaid, then the Vendor may execute such CCP, PDI Form and a Confirmation of Receipt on behalf of the Purchaser and the Purchaser hereby irrevocably appoints the Vendor as the Purchaser's lawful attorney, in accordance with the provisions of Section 11.1(1) of this Schedule, to complete the CCP, PDI Form and Confirmation of Receipt on the Purchaser's behalf, without liability and the Purchaser shall be bound as if the Purchaser had executed same, or the Vendor may, at its option, declare the Purchaser to be in default under this Purchase Agreement and may terminate this Purchase Agreement by written notice to the Purchaser or the Purchaser's Solicitors and the provisions of Section 8.1 of this Schedule shall apply.
- (4) The Purchaser acknowledges and agrees that after registration of the Condominium and prior to the turnover meeting (while the Vendor controls the Condominium board of directors), the Vendor shall cause the Condominium to enter into an agreement (the "Construction Warranties Agreement") with the Vendor which shall provide as follows: (i) the Condominium shall have no rights against the Vendor beyond those that are specifically granted to the Condominium under the Condo Act, The ONHWP Act and by Tarion; (ii) the Condominium's only recourse against the Vendor for a final and binding resolution of any outstanding, incomplete or deficient construction items and any other construction warranty matters which fall within the jurisdiction of Tarion shall be through the process established for and administered by Tarion; (iii) the Condominium and the Vendor shall appoint and constitute Tarion as the sole and final arbiter of all such matters; (iv) the Purchaser and the Condominium shall indemnify and save the Vendor harmless from all actions, causes of action, claims and demands for damages or loss which are brought by the Condominium in contravention of the Construction Warranties Agreement; and (v) the Construction Warranties Agreement shall not be terminated or terminable by the Condominium following the turnover meeting.
- (5) The Purchaser hereby releases the Vendor from any liability whatsoever in respect of improvements or betterments to, or chattels placed within the Purchased Home. The Purchaser further acknowledges that the Vendor is not responsible for the repair or rectification of any work resulting from ordinary settlement including walkways, patio stones, interlocking bricks or sodded areas or for any damage to improvements or decor caused by material shrinkage, twisting or warpage. Furthermore, the Vendor shall not be liable for any secondary or consequential damages whatsoever which may result from any defect in materials, design or workmanship related to the Condominium and/or the Purchased Home except as required under The ONHWP Act. The Vendor shall have no liability whatsoever for work done by a third party trade at the behest of the Purchaser, either before or after the Closing Date, whether or not such third party trade was referred to the Purchaser by the Vendor. The Purchaser agrees that the foregoing may be pleaded by the Vendor as an estoppel in any action brought by the Purchaser or the Purchaser's successors in title against the Vendor arising from construction deficiencies.
- (6) The Vendor shall complete the common elements as soon as reasonably practicable, however the failure of the Vendor to complete the common elements or to complete the Purchased Home beyond the minimum standards required by the Approving Authority in order to permit occupancy thereof, on or before the Firm Occupancy Date, or the Closing Date, (as the case may be) shall in no event entitle the Purchaser to refuse to take possession of the Purchased Home on the Firm Occupancy Date and/or to close this transaction on the Closing Date, or to fail to remit to the Vendor the purchase monies and other monies required to be paid by the Purchaser hereunder or to set off the Occupancy Fee
- (7) The Purchaser further acknowledges and agrees that any duties of care, warranties of workmanship or materials or liabilities of any sort in respect of any aspect of the construction of the Purchased Home or the common elements of the Condominium, which the Vendor may warrant or owe to the Purchaser or to the Condominium, whether implied by this Purchase Agreement or at law or in equity, or by any statute or otherwise, including breach of contract, breach of warranty, negligence or breach of duty, shall be restricted to only those warranties, duties and liabilities deemed to be given or owed by the Vendor under The

ONHWP Act and shall extend only for the time period and in respect of those items covered or provided by The ONHWP Act. The Vendor and the Purchaser agree that all disputes, if any, respecting any aspect of construction of the Purchased Home or the common elements of the Condominium, including disputes alleging negligence, breach of contract, breach of duty or breach of warranty, shall be limited solely to the dispute resolution mechanisms prescribed by The ONHWP Act, which resolution thereunder shall be binding and conclusive on all parties.

- (8) The Purchaser covenants and agrees that on the Firm Occupancy Date either the Purchaser or a Purchaser's relations (as that term is defined in the HST Act) shall forthwith occupy the Property so as to comply with the provisions of The ONHWP Act regarding occupancy by a new home buyer.
- (9)The Purchaser shall not enter on to the Lands or the other Galleria Master Development Lands or into the Condominium, prior to the Firm Occupancy Date or the Closing Date (as the case may be) without obtaining the prior written consent of the Vendor, which consent may be unreasonably or arbitrarily withheld. In the event that the Purchaser is permitted to enter on to the Lands or the other Galleria Master Development Lands or into the Condominium, the Purchaser shall at all times be accompanied by a representative of the Vendor and shall wear a hard hat and other protective equipment as may be required by the Vendor, if any. In the event that the Purchaser does enter on to the Lands or the other Galleria Master Development Lands or into the Condominium prior to the Firm Occupancy Date or the Closing Date (as the case may be), whether with or without the prior written consent of the Vendor, the Purchaser and any person who enters on to the Lands or the other Galleria Master Development Lands or into the Condominium with the Purchaser, assumes all risk in respect of the Purchaser's or their entrance on to the Lands or the other Galleria Master Development Lands or into the Condominium and the Purchaser hereby irrevocably waives and releases any claims that the Purchaser or any such entrant may have against the Vendor or any Alternate Entity, in respect of any claim that the Purchaser or such entrant might otherwise have against the Vendor, whether arising as a result of the negligence of the Vendor or those for whom the Vendor is at law responsible, or otherwise. The Purchaser will indemnify and hold the Vendor harmless for all losses, costs, causes of action, actions, damages and expenses, including legal fees and disbursements on a full indemnity basis incurred or suffered by the Vendor, occasioned in whole or in part as a result of the Purchaser's entrance on to the Lands or the other Galleria Master Development Lands or the Condominium. Any unauthorized entrance on to the Lands or the other Galleria Master Development Lands or into the Condominium shall constitute a trespass and a default under this Purchase Agreement and in such case, the Vendor may exercise such rights and remedies available to it hereunder and at law.

6.2 CONSTRUCTION AND CHANGES

- (1) The Purchaser acknowledges that the Condominium of which the Purchased Home forms a part will be constructed to at least the minimum Ontario Building Code requirements in existence at the time that the application for the building permit for the Condominium is submitted to the applicable Approving Authorities. The Purchaser covenants and agrees that the Purchaser shall have no claims against the Vendor for any equal, higher or better standards of workmanship or materials. The Purchaser agrees that the foregoing may be pleaded by the Vendor as an estoppel in any action brought by the Purchaser or the Purchaser's successors in title against the Vendor.
- (2) The Vendor may, from time to time, change, vary or modify at the request or requirement of any Approving Authorities or any mortgagee of the Lands, any elevations, building specifications or site plans or any part of the Condominium or the Property (including without limitation reversing the layout of the Property), or to conform with any Approving Authorities or architectural requirements related to building codes, Official Plan or Official Plan Amendments, the zoning by-law respecting the Lands, municipal site plan approval or architectural control, or for any other reasons as the Vendor in its sole and unfettered discretion determines.
- (3) The Purchaser acknowledges and agrees with the Vendor that:
 - (a) any and all stated or represented area measurements made by or on behalf of the Vendor or by any sales agent may vary from actual or usable floor space or living space and that the suite area of the Property, as may be represented or referred to by the Vendor or any sales agent, or which appear in any sales material is approximate only, and is generally measured to the outside of all exterior, corridor and stairwell walls, and to the centre line of all party walls separating one Unit from another:
 - (b) further changes may be made to the plans and specifications existing at inception of the Condominium, or as they existed at the time the Purchaser entered into this Purchase Agreement;

- (c) the plans and specifications illustrated on any marketing materials, sales brochures, models, renderings, floor plans, site drawings, rendering or otherwise that the Purchaser may have reviewed prior to entering into this Purchase Agreement remain conceptual and are subject to the final review and approval of any applicable Approving Authority and the Vendor's design consultants and engineers;
- (d) the floor area measurements are generally calculated based on the middle floor of the Condominium for each unit type, such that Units on lower floors may have less floor space due to thicker structural members, mechanical rooms, etc., while Units on higher floors may have more floor space;
- (e) any changes to the size of the Purchased Home which are not in excess of the variance permitted under Tarion Builder Bulletin 22, shall not be considered material;
- (f) the ceiling height of the Purchased Home is measured from the upper surface of the concrete floor slab (or subfloor) to the underside surface of the concrete ceiling slab (or joists) and that ceiling heights may vary due to the existence of bulkheads, dropped ceilings, other structural requirements or due to the requirements of any Approving Authority, none of which shall constitute a material change to the Purchased Home nor shall such variations entitle the Purchaser to an abatement of the Purchase Price or any claim for compensation whatsoever;
- (g) with respect to any aspect of construction, finishing or equipment, the Vendor shall have the right, without the Purchaser's consent, to substitute materials for those described in this Purchase Agreement or in the plans or specifications, provided the substituted materials are in the judgement of the Vendor's architect, whose determination shall be final and binding, of equal quality or better;
- (h) the colour, texture and/or shading of any wood finishes, carpet, tiles, kitchen and bathroom cabinetry or other manufactured finishing materials are subject to minor variations or shading in dye-lots produced or manufactured by the suppliers and that such variations are acceptable to the Purchaser;
- (i) marble, wood and other natural materials are subject to inherent variations in colour and grain, that ceramic tile and broadloom are subject to pattern, shade and colour variations and that such variations are acceptable to the Purchaser;
- (j) the variations and contrast between the different colours, textures, grains and/or shading of any wood finishes, carpet, tiles, kitchen and bathroom cabinetry or other manufactured finishing materials selected may, when constructed, be much greater than those variations and contrasts may have appeared to be from the sample finishes, colours and materials from which the Purchaser's selections were made; and
- (k) transition strips shall be used to bridge flooring height differences within the Purchased Home due to variations in flooring and their thicknesses.

The Purchaser shall have no claim against the Vendor for any of the foregoing changes, variances or modifications nor shall the Vendor be required to give notice thereof to the Purchaser, and the Purchaser hereby consents to any such changes, variances or modifications and agrees to complete the purchase notwithstanding same.

- (4) The Purchaser acknowledges that any illustration in any sales brochure, any models in the sales office and any model suites may contain features, decorations and chattels that are not included in the Purchased Home or under this Purchase Agreement and that only those items set out in Schedule "A" to this Purchase Agreement are included in the Purchase Price.
- (5) The Purchaser agrees that the Purchase Price is based solely on the Purchased Home and is not dependant upon any common elements, whether or not for exclusive use, appurtenant to the respective Purchased Home being of a certain size, layout or configuration.
- (6) The Purchaser agrees that the Purchaser or the Purchaser's Designate shall meet with the Vendor or its representative at a time or times designated by written notice from the Vendor to the Purchaser or the Purchaser's Designate on no more than two occasions, in order to (i) select (or reselect as a result of any discontinued items) finishes, colours and materials for those items in Schedule "A", where the Purchaser has a choice as indicated in Schedule "A", from a selection offered by the Vendor and at the Vendor's design studio or such other location as determined by the Vendor in its sole and unfettered discretion, and (ii) execute all documentation respecting such selections as required by the Vendor. If the Purchaser or the Purchaser's Designate fails to attend at the designated time to make such selections or attends but refuses or otherwise fails to execute the documentation respecting such selections as required by the Vendor, the Purchaser hereby irrevocably appoints the Vendor

as the Purchaser's lawful attorney in accordance with the provisions of Section 11.1(1) of this Schedule to complete the selection of finishes, colours and materials for such items and to execute all documentation respecting such selections as required by the Vendor, and the Vendor may, but shall not be obliged to, complete same. In the event that the Vendor makes such selections, the Purchaser shall still be responsible to sign and deliver to the Vendor on the Firm Occupancy Date all documents and instruments required to be given to the Vendor, including a written confirmation of any selections made by or documents executed by the Vendor pursuant to this Section 6.2(6), and shall be required to pay all monies as required by this Purchase Agreement. The Purchaser acknowledges that there shall be no reduction in the Purchase Price or credit for any standard feature listed in Schedule "A" which is omitted at the Purchaser's request.

- (7) In the event that the Purchaser requests any alterations to its choice for a finishing item or colour selection after the date on which the original selection is finalized, the Purchaser agrees to pay all costs associated with this amendment plus an administration fee of \$800 plus HST each time an amendment is requested. For greater certainty, the Purchaser acknowledges and agrees that the Vendor shall have no obligation to accommodate any requested amendment. The Vendor's approval shall depend on its construction schedule and the availability of labour, materials and supplies, in the sole and absolute discretion of the Vendor. In the event that the Vendor approves any requested amendment (in whole or in part), all costs associated with the amendment plus the administration fee shall be paid by the Purchaser at the time of such amendment or credited to the Vendor as an adjustment to the balance due on either the Firm Occupancy Date or the Closing Date (payment arrangements to be determined by the Vendor, in its sole and unfettered discretion). The Vendor may charge an administration fee of \$300 plus HST for each appointment scheduled for the purpose of selecting finishing items that is confirmed by the Purchaser with the Vendor and then missed, cancelled and/or re-scheduled by the Purchaser.
- (8) The Vendor may refuse to provide a requested Extra in its sole and unfettered discretion. Without limiting the generality of the foregoing, the Vendor may require as a condition of agreeing to provide a requested Extra that the Purchaser and the Vendor enter into an amending agreement in accordance with the provisions of Section 4 of the Tarion Addendum amending the Critical Dates to provide for additional time that the Vendor deems necessary or desirable in order to accommodate the requested Extra.
- (9)If any of the Extras ordered by the Purchaser, through the Vendor, are not supplied or cannot be completed, for whatever reason, by the Closing Date (the "Uncompleted Extras"), then at the option of the Vendor, the Vendor shall refund to the Purchaser upon the Closing Date all amounts paid by the Purchaser to the Vendor in connection with the Uncompleted Extras, and the amount so refunded by the Vendor to the Purchaser for the Uncompleted Extras (or for which, at the Vendor's option, the Purchaser shall receive a credit on the statement of adjustments on Closing) shall be accepted by the Purchaser as full and final settlement of any claim by the Purchaser with respect to such Uncompleted Extras, and the Purchaser acknowledges that the Vendor's liability with respect to the Uncompleted Extras shall be limited solely to the return or credit of the amounts referred to as aforesaid, and upon such payment being made or credit being given, the Vendor will be released from any and all obligations, claims or demands whatsoever with respect thereto, including the obligation of the Vendor to supply or complete such Uncompleted Extras. If, as a result of the installation of any Extras or any condition of the installers of such Extras, the Purchased Home is not substantially completed sufficient to permit Occupancy thereof by the Firm Occupancy Date, the Vendor shall not be held liable for the installation of any Extras or any condition of the installers of such Extras and the Purchaser shall, on the Firm Occupancy Date: (i) pay to the Vendor those sums required pursuant to this Purchase Agreement, and (ii) execute and deliver to the Vendor all documents and instruments required on the Firm Occupancy Date. In the event that the Vendor does not opt to refund or credit the amounts paid by the Purchaser in connection with the Uncompleted Extras, the Vendor shall supply or complete such Uncompleted Extras within a reasonable time after the Closing Date and the Purchaser shall give the Vendor access to the Purchased Home to permit the Vendor to supply or complete such Uncompleted Extras. In such circumstances, the Vendor shall not be required to refund or credit on the Closing Date any amount paid by the Purchaser in respect of such Uncompleted Extras nor shall the Purchaser be entitled to holdback any part of the Purchase Price to be paid on the Closing Date due to failure of the Vendor to supply or complete the Uncompleted Extras prior to the Closing Date. The Purchaser agrees that the Vendor shall not be obligated to provide Extras.
- (10) It is understood and agreed that the Vendor shall not be responsible or liable in any way to the Purchaser for the quality of (and/or the workmanship with respect to) any Extras unless same are supplied and/or constructed directly by the Vendor and then only if the Vendor specifically agrees in writing to be liable for same.
- (11) The Purchaser agrees that the Purchaser shall satisfy himself or herself as to whether or not any automobile which the Purchaser may wish to park in a Parking Unit will fit in the parking garage forming part of the Building, including the entrance, and in the Parking Unit and the Vendor makes no representation or warranty in respect of any such automobile fitting in the parking garage or in any Parking Unit.

- (12) The Purchaser covenants that it will not interfere with the construction and completion of any portion of the Condominium, other components of the Galleria Master Development by the Developers or their respective trades as they carry out their Construction Activities and agrees to take possession of and occupy the Purchased Home subject to such conditions, provided the Purchased Home is reasonably fit for occupancy.
- (13) The Purchaser hereby irrevocably waives and releases any claims that the Purchaser may have against the Developers or any Alternate Entity in respect of the Construction Activities and the foregoing matters. The Purchaser shall be required to obtain a similar covenant (enforceable by and in favour of the Vendor), from any subsequent transferee of the Purchased Home and/or in any agreement entered into between the Purchaser and any subsequent transferee of the Purchased Home. The Purchaser and the Corporation shall not be entitled to take any action that may interfere with or unduly delay the rights or ability of the Developers to perform and complete the Construction Activities and the foregoing matters.

6.3 <u>SITE CONDITIONS/NOTICES/WARNINGS</u>

The Purchaser acknowledges and agrees that:

- The Purchaser has received and reviewed a copy of the draft condominium plan and a copy (1) of the Disclosure Statement. The Purchaser has read, understands, acknowledges and accepts the site conditions, notices and warnings detailed in the Disclosure Statement, including those listed under the heading "Miscellaneous". Further, the Purchaser expressly acknowledges and agrees that (i) such site conditions, notices and warnings may impact an individual Purchaser's intended use and enjoyment of their Purchased Home; (ii) the Vendor has included such site conditions, notices and warnings in the Disclosure Statement to provide the Purchaser with relevant information to the Purchaser's decision to purchase the Purchased Home; (iii) the Purchaser is satisfied with the location of the Purchased Home in proximity to all proposed activities, structures, amenities and facilities; and (iv) the Vendor has provided no guarantee against the possibility of impacts on the Purchaser's use and enjoyment of their Purchased Home as a result of such site conditions, notices and warnings. The Purchaser hereby agrees to execute any and all acknowledgements of the foregoing matters in the form provided by the Vendor's Solicitors and, further, if the Vendor is required to incorporate any such matters into the final Condominium Documents, the Purchaser hereby agrees to accept same without in any way affecting this transaction. In any event, the Purchaser shall be deemed to accept any such acknowledgements once such acknowledgements have been provided to the Purchaser. The Purchaser hereby waives and releases any claims, including for refund, credit, reduction/abatement or set-off whatsoever against any portion of the Purchase Price or the Occupancy Fee, that the Purchaser may have against the Vendor and any Alternate Entity in respect of, or arising out of or as a result of, the Requirements (as defined in the Disclosure Statement) or as a result of any of the Interferences (as defined in the Disclosure Statement) or the Situations (as defined in the Disclosure Statement). The Purchaser hereby agrees to complete this transaction notwithstanding the existence of the Requirements (as defined in the Disclosure Statement) or as a result of any of the Interferences (as defined in the Disclosure Statement) or the Situations (as defined in the Disclosure Statement). This Section may be pleaded as a complete defence to any claim against the Vendor arising from the foregoing matters in this Section.
- (2) The Toronto Transit Commission may require that a warning clause be inserted into all agreements of purchase and sale and/or lease agreements for the Property in respect of noise and vibration resulting from current or future transit operations.
- (3) Despite the best efforts of the Toronto District School Board, sufficient accommodation may not be available in the neighborhood schools for all students anticipated from the development area. Students may be accommodated in school facilities outside the neighborhood or students may later be transferred to other school facilities. For information regarding designated school(s), please call (416) 394-7526. Purchasers agree that for the purpose of transportation to school, if bussing is provided by the Toronto District School Board in accordance with the TDSB's bussing policy, students will not be bussed from their home to the school facility, but will meet the bus at designated locations in or outside of the area.
- (4) Whereas, despite the best efforts of the Toronto Catholic District School Board, sufficient accommodation may not be available for all anticipated students at local schools, you are hereby notified that it may be necessary for Elementary and/or Secondary students from this development to be accommodated in facilities outside of the community depending on availability of space.
- (5) Canadian Pacific Railway Company or its assigns of successors in interest has or have rights-of-way within 300 metres from the Lands. There may be alterations to or expansions of the railway facilities on such rights-of-way in the future including the possibility that the railway or its assigns or successors as aforesaid may expand its operations, which expansion may affect the living environment of the residents in the vicinity, notwithstanding

the inclusion of any noise and vibration attenuating measures in the design of the development and individual dwelling(s). CPR will not be responsible for any complaints or claims arising from use of such facilities and/or operations on, over or under the aforesaid rights-of-way.

6.4 RIGHT OF ENTRY

Notwithstanding the Purchaser occupying the Purchased Home on the Firm Occupancy Date or the Closing Date, the Vendor or any person authorized by it (including Approving Authorities or any other authority having jurisdiction) shall have the right, without notice, to enter the Purchased Home at all reasonable hours and times and shall have access to the Lands, the Condominium and the Purchased Home in order to make inspections or to do any work or repairs therein which may be deemed necessary by the Vendor or Approving Authorities in connection with the completion, rectification, maintenance or servicing of any installation in the Condominium or the Purchased Home or other Units, or in order to make inspections to permit any mortgagee to make necessary advances due to the Vendor under any mortgage, and such rights shall be in addition to any rights and easements created hereunder or under the Condo Act. At the Vendor's sole and unfettered discretion, a right of entry in favour of the Vendor for a period not exceeding five (5) years similar to the foregoing may be included in the Transfer/Deed provided on the Closing Date and acknowledged by the Purchaser.

6.5 REZONING OF LANDS AND OTHER LANDS

- The Purchaser acknowledges that the Developers and/or their respective successors and assigns and/or any Alternate Entity or a third party purchaser of the adjoining lands, in its or their sole and unfettered discretion may in the future construct another building or buildings or structures on lands adjoining, neighbouring and in the same vicinity as the Lands. The Purchaser agrees not to object to such construction nor deem such construction as an inconvenience or nuisance, nor make a claim for damages or injuries or otherwise. The Purchaser hereby irrevocably consents to any Official Plan amendment, re-zoning, minor variance and/or Plans of Condominium or Subdivision required for the Lands and the lands adjoining, neighbouring and in the same vicinity as the Lands, in order that the Developers and/or any Alternate Entity and/or any affiliate or related company or a third party purchaser of the adjoining lands may proceed with its development of any such lands and hereby agrees not to object to any Official Plan amendment, re-zoning, minor variance and/or Plans of Condominium or Subdivision and other Committee of Adjustment applications brought by the Developers and/or any Alternate Entity and/or any affiliate or related company or a third party purchaser of the adjoining lands. The Purchaser further covenants and agrees to accept title to the Purchased Home subject to this covenant and restriction in favour of the Developers and others and to accept the transfer containing this covenant and restriction or one similar thereto and hereby undertakes and agrees to abide by such covenant and restriction after Closing and to extract a similar covenant and restriction from the Purchaser's immediate successors in title to the Purchased Home, all of which shall be assigned for the benefit of the Vendor and others.
- (2) The Purchaser acknowledges that any such Official Plan amendment, Plans of Condominium or Subdivision, rezoning, minor variance and/or other Committee of Adjustment applications may include applications to: (i) increase or decrease the number of residential units in the building(s) to be constructed on such land adjoining, neighbouring or in the same vicinity as the Lands; (ii) increase or decrease the height of such building(s); (iii) reduce the amenities or recreational facilities required to be included in such building(s) under the existing zoning by-law (if any); (iv) modify or reduce the requirements for parking in such buildings or on the Lands as provided for under the existing zoning by-laws (if any); (v) increase or decrease the gross floor area of the building(s) built or to be built on the Lands; (vi) change the location of the level and floor of the Properties to a different level and floor; and (vii) deal with such other matters as the Vendor and/or any Alternate Entity or a third party purchaser of the adjoining lands may, in its sole and unfettered discretion, determine.

6.6 MATERIAL CHANGE

In the event that there is any change in the interior layout of the Property or change in the size of the Purchased Home or other change, any of which is a material change as contemplated by Section 74(1) of the Condo Act, as determined in the sole and unfettered discretion of the Vendor, the Purchaser covenants and agrees that the Purchaser's sole remedy shall be the rescission rights afforded pursuant to Section 74(6) of the Condo Act, notwithstanding any rule of law or equity to the contrary. The Purchaser is advised that the Vendor shall have the right to increase or decrease the number of Units or floors in the Condominium (including increasing the number of floors by changing the ceiling height of some or all floors in the Condominium), to reduce the number of parking spaces and/or Parking Units or Storage Locker Units in the Condominium, to renumber Units and suites with numbers different from those provided herein and in the Condominium Documents, including any common elements for which the Purchaser has exclusive use such as balconies or terraces to change the municipal address of the Condominium, to change the proposed height, size, density, location, style, profile, cladding massing or design of the proposed Condominium, to alter the facilities comprising a part of the common elements and to

change the style or configuration of the Purchased Home (including changing varying or modifying the number, size and location of any windows in the Property), other Units, floors and common elements and to change the unit mix as set out in the Disclosure Statement, all at its sole and unfettered discretion and for any reason whatsoever.

PART VII - PLANNING ACT APPROVAL

7.1 PLANNING ACT APPROVAL

This Purchase Agreement is conditional until Closing upon compliance with Section 50 of the Planning Act (Ontario), as amended from time to time, by virtue of registration of the Creating Documents. In the event that the Vendor cannot obtain registration of the Creating Documents on or before Closing, the Vendor may, in its sole and unfettered discretion, terminate this Purchase Agreement, by written notice given to the Purchaser or the Purchaser's Solicitors, and which written notice shall terminate the Occupancy Arrangement effective the last day in the month following the month in which said written notice is given. The Purchaser hereby consents to such termination by the Vendor, when and if same occurs, and the Purchaser acknowledges and agrees that this consent is the consent contemplated by Section 79(2) of the Condo Act. The provisions of Section 3.4 of this Schedule (except Sections 3.4(5) and 3.4(9) of this Schedule) and this Section 7.1 shall survive such termination. If this Purchase Agreement is so terminated, then neither party shall be liable to the other for any costs or damages and the Deposit shall be returned to the Purchaser (excluding uncashed cheques). Provided that if the Purchaser is then in possession of the Purchased Home under Section 3.4 of this Schedule, the Deposit shall be returned to the Purchaser only upon the Purchaser giving up possession of the Purchased Home, and the Purchaser shall give up possession on 30 days' prior written notice from the Vendor requiring the Purchaser so to do, and the Purchaser shall pay the Occupancy Fee to the date of giving up possession. Monies paid for Occupancy Fees shall not be returned.

PART VIII - PENDING CLOSING

8.1 PURCHASER DEFAULT

- (1) The Purchaser shall be deemed to be in default under this Purchase Agreement, if (i) the Purchaser fails to observe or fulfil any of the provisions of this Purchase Agreement, or (ii) the Purchaser becomes bankrupt or insolvent or takes the benefit of any statute for bankrupt or insolvent debtors or makes any proposal, an assignment or arrangement with the Purchaser's creditors, or (iii) any lien, execution, certificate of pending litigation, notice or other encumbrance arising from any action or default whatsoever of the Purchaser is registered against the title to any of the Lands or the Purchased Home, or affects the Lands, the Condominium or the Purchased Home. The Vendor shall then be entitled to accelerate the payment of all Deposits by delivering a written notice upon the Purchaser to pay the balance of the Deposits, which shall be paid within two (2) Business Days of the Vendor's demand thereof, failing which the Vendor reserves all remedies set forth in Section 8.1(3) of this Schedule without further notice.
- (2) The Vendor shall be entitled (but not required) to satisfy any such lien, execution or encumbrance and the Vendor is hereby irrevocably appointed lawful attorney for the Purchaser with the power to execute any document or consent that may be necessary to remove any title encumbrance or registration as provided in 8.1(1)(iii). If the Vendor is required to pay any monies to remove any such title encumbrance or registration from the title register, the Purchaser on a demand basis shall reimburse the Vendor for all monies paid or for security given and costs (including legal fees and disbursements on a full indemnity basis) with interest payable thereon from and after the date of payment by the Vendor. Except as otherwise expressed in this Purchase Agreement to the contrary, all amounts which shall be owing pursuant to this Purchase Agreement and not paid when due, shall bear interest, both before and after demand, judgment and default, at the Prime Rate plus 5% both calculated, compounded and payable monthly.
- If the Purchaser is in default under this Purchase Agreement either before or after the Firm (3)Occupancy Date and fails to remedy such default within two (2) Business Days of written notice thereof to the Purchaser, (other than the Purchaser's failure to complete the transactions contemplated by this Purchase Agreement on the Firm Occupancy Date and the Closing Date or other circumstances expressly set out herein, in which event no written advice shall be required to be delivered to the Purchaser and no remedy period shall be available) notwithstanding any act of the Vendor, the Vendor at its sole option, may declare this Purchase Agreement at an end and (i) the Purchaser shall vacate the Purchased Home if the Purchaser has taken possession thereof and (ii) the Deposit together with all Interest thereon shall be forfeited to the Vendor without prejudice to any other rights that the Vendor would have against the Purchaser at law or in equity. The Vendor shall not be obliged to return any monies paid by the Purchaser as an Occupancy Fee or an administrative fee and any monies paid by the Purchaser for Extras, whether or not installed in the Purchased Home or ordered by the Vendor, and same shall be deemed to form part of the Deposit and also be retained by the Vendor. The Vendor reserves all its rights and remedies under this Purchase Agreement, at law and in equity, against the Purchaser and shall be at liberty to re-sell the Purchased Home with or without re-entry and without prejudice to the Vendor's

right to damages at law or in equity and the Vendor shall be released from all of its obligations to the Purchaser under this Purchase Agreement.

- (4) The Vendor may in its sole and unfettered discretion grant a written waiver of any breach of any provision of this Purchase Agreement. No written waiver or extension by the Vendor shall be deemed to condone any future breach and any such waiver shall be without prejudice to the Vendor's rights. The Vendor's failure to notify the Purchaser of a default shall not constitute a waiver of such default.
- (5) Notwithstanding any other provision contained in this Purchase Agreement and except as set forth in the Addendum, if this Purchase Agreement is terminated through no fault of the Purchaser, the Deposit shall be returned to the Purchaser (excluding uncashed cheques) and the Purchaser shall sign a written acknowledgement confirming the termination and amount of monies refunded. The Vendor, however, shall not be obliged to return any monies paid by the Purchaser as an Occupancy Fee or administrative fee. In no event shall the Vendor or its agents be liable for any damages or costs whatsoever whether arising as a result of the negligence of the Vendor or those for whom the Vendor is at law responsible or otherwise, and, without limiting the generality of the foregoing, for any loss of bargain, for any relocating costs, for loss of use of Deposits or for any professional or other fees paid in relation to this transaction. This Section 8.1(5) may be pleaded by the Vendor as a complete defence to any such claim.
- (6) Except as set forth in the Addendum, in the event that this Purchase Agreement is terminated, the Purchaser shall execute such releases or any other assurances or documents as the Vendor may require to release the Purchaser's interest in the Purchased Home, the Condominium and the Lands (including releases of any surety) and in the event the Purchaser refuses or fails to execute such releases, assurances or documents, the Purchaser hereby constitutes and irrevocably appoints the Vendor as the Purchaser's lawful attorney to execute such releases, assurances and documents, pursuant to Section 11.1(1) of this Schedule.

8.2 NON-REGISTRATION

The Purchaser acknowledges that this Purchase Agreement does not create an interest in the Lands, the Condominium or the Purchased Home and that until a Transfer/Deed of Land is registered in favour of the Purchaser, the Purchaser shall have no interest in the Lands, the Condominium or the Purchased Home. The Purchaser further covenants and agrees that the Purchaser will not register or cause or permit this Purchase Agreement to be registered on title to the Lands, the Condominium or the Purchased Home and that no reference to it, or notice of it or any caution or any certificate of pending litigation, purchaser's lien or any other notice or document of any type shall be registered on title whether or not the Vendor is in default hereunder. The Purchaser shall be deemed to be in default under this Purchase Agreement if the Purchaser creates any encumbrance or makes any registration or causes or permits any encumbrance or registration to be made on title to the Lands and/or the Purchased Home on or before the Closing Date.

Should the Purchaser be in default of the Purchaser's obligations under this Section 8.2, the Vendor may, in accordance with the provisions of Section 11.1(1) of this Schedule, as agent and attorney of the Purchaser, cause removal of any such registration from the title to the Lands, the Purchased Home or the Condominium. In addition, should the Purchaser be in default of the Purchaser's obligations under this Section 8.2, the Vendor, at its sole option, shall have the right to declare this Purchase Agreement null and void in accordance with the provisions of Section 8.1 of this Schedule and the provisions of Section 8.1 of this Schedule shall apply. The Purchaser hereby irrevocably consents to a court order removing any notice of this Purchase Agreement, any caution, any certificate of pending litigation, any purchaser's lien or any other notice or document of any sort whatsoever from title to the Lands, the Purchased Home or the Condominium and the Purchaser agrees to pay all of the Vendor's costs and expenses in obtaining such order (including the Vendor's Solicitors' fees and disbursements on a full indemnity basis).

8.3 NO SALE OR LEASE

The Purchaser represents to the Vendor, upon which representation the Vendor has relied in accepting this Purchase Agreement, that the Purchaser is purchasing the Purchased Home for the Purchaser's own personal use or for the use of a family member or relative and not for short term speculative purposes.

The Purchaser further covenants and agrees not to either directly or indirectly list for sale, offer to sell, advertise to sell, sell, transfer, assign, offer for lease, lease, license the use of, direct title on Closing to any third party (other than the Purchaser's spouse), mortgage, encumber or otherwise part with the Purchaser's interest in this Purchase Agreement, the Purchased Home, the Condominium or the Lands (collectively and individually hereinafter referred to as a "**Transfer**"), nor permit any third party to list or advertise the Purchased Home for sale at any time until after the Closing Date, subject to this Section 8.3, without the prior written consent of the Vendor, which consent may be withheld in its sole and unfettered discretion. In the event the Vendor provides its consent as aforesaid, such Purchased Home may only be marketed by an agent designated by the

Vendor, with all applicable commissions payable by the Purchaser in the event of a resale. Any Transfer in contravention of this provision shall be null and void. The Purchaser acknowledges that once a breach of this covenant occurs, it is incapable of rectification and, accordingly, the Purchaser acknowledges, understands and agrees that in the event of such breach, the Vendor shall have the unilateral right and option of terminating this Purchase Agreement and the Occupancy Arrangement effective upon delivery of written notice of termination to the Purchaser or the Purchaser's Solicitors whereupon the provisions of Section 8.1 of this Schedule shall apply.

PART IX - CLOSING

9.1 CLOSING DATE

- (1) Closing shall occur on the Closing Date. On a date prior to or following registration of the Creating Documents, such date to be determined by the Vendor in its sole and unfettered discretion, the Vendor shall give written notice to the Purchaser of the Closing Date (a "Closing Date Notice"). In any event, the Closing Date shall be no earlier than the 10th day following the date on which the Closing Date Notice is given and no later than 36 months from the Firm Occupancy Date.
- (2) The Purchaser covenants and agrees that the Purchaser will at the Purchaser's own expense complete the transaction in accordance with the Closing Date Notice and register the transfer of title, forthwith after delivery of the transfer to the Purchaser.
- (3) The Purchaser agrees that the Vendor's execution of any closing documents required or desired in connection with the Occupancy Arrangement and/or Closing of the Purchased Home may, at the Vendor's sole option, be made by way of an electronic signature, as such term is defined in the *Electronic Commerce Act 2000* (Ontario), as amended from time to time ("ECA"), undertaken by or through a computer program or any other electronic means, as expressly provided for or contemplated by, and in accordance with, the provisions of the ECA. The Purchaser acknowledges and agrees to accept any information, documentation or other communication provided by the Vendor and/or the Vendor's Solicitors pursuant to this Purchase Agreement in an electronic form and, if requested by the Vendor and/or the Vendor's Solicitors in the Vendor's sole and unfettered discretion, to provide any information, documentation or other communication required under this Purchase Agreement to the Vendor and/or the Vendor's Solicitors in an electronic form when and in the form requested by the Vendor and/or the Vendor's Solicitors.
- (4) The Vendor, at its sole option, may direct the Vendor's Solicitors to deliver any or all documents related to this transaction to the Purchaser or the Purchaser's Solicitors, electronically, by way of internet access to an established website. Any documents containing personal information will be delivered via a secure website, accessed through a personal password to be provided to the Purchaser or the Purchaser's Solicitors by the Vendor's Solicitors. The posting of documents on such established website and secure website shall constitute a valid delivery and tender of such documents to and on the Purchaser and the Purchaser's Solicitors immediately upon provision of the personal password by the Vendor's Solicitors to the Purchaser's Solicitors.

9.2 TENDER

The Purchaser hereby irrevocably waives personal tender and agrees that failing other mutually acceptable arrangements, and subject to the provisions of Section 9.5(6) of this Schedule, tender may be validly and effectively made at the Vendor's sole option:

- (1) upon the Purchaser or the Purchaser's Solicitors, if the documents required to be delivered by the Vendor to the Purchaser, including any statutory declarations or affidavits, are delivered by facsimile on or before the Closing Date with originals of same to follow in due course;
- (2) upon the Purchaser or the Purchaser's Solicitors, if the documents required to be delivered by the Vendor to the Purchaser are executed in accordance with the ECA and Section 9.1(3) of this Schedule and posted to one or more internet websites in accordance with Section 9.1(4) of this Schedule; or
- (3) upon the Purchaser or the Vendor, by tender being made upon the Purchaser's Solicitors or the Vendor's Solicitors (as the case may be), or upon the Vendor or the Purchaser.

Notwithstanding the foregoing, if the Purchaser or the Purchaser's Solicitors, including any representative or employee of the Purchaser or the Purchaser's Solicitors, indicates or expresses to the Vendor or the Vendor's Solicitors, on or before the Closing Date, that the Purchaser is unable or unwilling to complete the purchase or take Occupancy, the Vendor shall be relieved of any obligation to make formal tender upon the Purchaser or the Purchaser's Solicitors and the Vendor may exercise forthwith any and all of its rights and remedies provided for in this Purchase Agreement and at law. All payments under this Purchase Agreement must be tendered by certified cheque or bank draft drawn on any of one of Bank of Montreal, Bank of Nova Scotia, Canadian Imperial Bank of Commerce, Royal Bank of Canada, Toronto-Dominion Bank or HSBC Bank

Canada. The Purchaser acknowledges that the Vendor and the Vendor's Solicitors will not, on Closing, be required to accept certified cheques or bank drafts that have been drawn from a bank account that is not the Purchaser's Solicitor's trust account and the Purchaser and the Purchaser's Solicitors shall provide to the Vendor and the Vendor's Solicitors, evidence satisfactory to the Vendor and the Vendor's Solicitors in their sole and unfettered discretion, that any certified cheques or bank drafts delivered on Closing were in fact drawn from the Purchaser's Solicitor's trust account.

9.3 RESIDENCY

The Vendor and the Purchaser both covenant to and with each other that they are not, and will not on the Firm Occupancy Date or the Closing Date be, a non-resident within the meaning of Section 116 of the *Income Tax Act* (Canada), as amended from time to time. Not later than the Closing Date, the Vendor and the Purchaser will furnish each other with and they both agree to accept as the only evidence thereof, a personal statutory declaration, or a declaration of one of the Vendor's or Purchaser's officers if the Vendor or Purchaser is a corporation, confirming the foregoing. If the Vendor believes for whatever reason that the Purchaser is a non-resident of Canada, regardless of any documentation provided by or on behalf of the Purchaser (including any statutory declaration sworn by the Purchaser) to the contrary, and the Vendor's belief or position on this matter is communicated to the Purchaser or the Purchaser's Solicitors on or before the Closing Date, then notwithstanding anything hereinbefore or hereinafter provided to the contrary, the Vendor shall be entitled to withhold the requisite tax owing in respect of any interest that may be payable to the Purchaser as provided in this Purchase Agreement and shall remit same to the Canada Revenue Agency after Closing and the statement of the adjustments shall reflect such withholding.

9.4 HOLDBACKS

The Purchaser agrees that on taking possession of the Purchased Home and/or on the Closing Date the Purchaser will not for any reason whatsoever request, nor will the Vendor be obligated to permit, any holdback of any part of the Purchase Price or any other monies payable by the Purchaser under this Purchase Agreement for any reason whatsoever.

9.5 ELECTRONIC REGISTRATION

In the event that the Teraview Electronic Registration System ("**TERS**") is operative in the Land Registry Office, then at the option of the Vendor's Solicitors, in their sole and unfettered discretion, the following provisions in regard to Closing shall prevail, namely:

- (1) the Purchaser shall retain and authorize the Purchaser's Solicitors to enter into an escrow closing agreement (the "Escrow Agreement") with the Vendor's Solicitors on the latter's standard form which may in the Vendor's Solicitor's sole and unfettered discretion contain some or all of the provisions of the Document Registration Agreement prepared by the Law Society of Ontario and adopted by the Joint LSO-CBAO Committee on Electronic Registration of Title Documents, as may be amended from time to time, establishing the procedures and timing for completing this transaction, such Escrow Agreement to be returned to the Vendor's Solicitors, as executed by the Purchaser's Solicitors, at least 7 days prior to the Closing Date;
- (2) the delivery and exchange of documents and monies and the release thereof to the Vendor and the Purchaser, as the case may be:
 - (a) shall not occur contemporaneously with the registration of the transfer/deed (and other registerable documentation);
 - (b) shall be governed by the Escrow Agreement, pursuant to which the solicitors receiving the documents and certified funds will be required to hold same in escrow, and will not be entitled to release same except in strict accordance with the provisions of the Escrow Agreement;
 - (c) may, in the case of documents to be delivered by the Vendor, occur electronically, through the use of one or more Internet websites, pursuant to Section 9.1 of this Schedule; and
 - (d) may at the option of the Vendor, in the case of funds to be delivered by the Purchaser, occur electronically, through any electronic funds transfer system designated by the Vendor or the Vendor's Solicitors, and in such case:
 - (i) the Purchaser and or the Purchaser's Solicitors shall execute such documents as the Vendor or the Vendor's Solicitors may require in connection therewith; and
 - (ii) the Purchaser's Solicitors shall be registered with such provider and at the request of the Vendor's Solicitors shall provide evidence of such registration to the Vendor's Solicitors at least ten (10) days prior to Closing;

- (3) if the Purchaser's Solicitors are unable to complete this transaction via TERS, in accordance with the provisions of the Escrow Agreement, then the Purchaser's Solicitors (or the authorized agent thereof) shall be obliged to personally attend at the office of the Vendor's Solicitors, at such time on the scheduled closing date as may be directed by the Vendor's Solicitors or as mutually agreed upon, in order to complete this transaction via TERS utilizing the computer facilities in the Vendor's Solicitors' office, at the cost of the Purchaser's Solicitors;
- (4) the Purchaser expressly acknowledges and agrees that the Purchaser will not be entitled to receive the transfer/deed to the Purchased Home for registration or the keys, until the balance of funds due on Closing, in accordance with the statement of adjustments, and all other amounts required to be paid by the Purchaser hereunder are remitted by certified cheque or bank draft via personal delivery to the Vendor's Solicitors (or in such other manner as the Vendor's Solicitors may direct) prior to the release of the transfer/deed for registration;
- (5) the Purchaser agrees that the delivery of any documents not intended for registration on title to the Purchased Home may be delivered to the Purchaser or the Purchaser's Solicitors by facsimile transmission (or by a similar system reproducing the original or by electronic transmission of electronically signed documents through the Internet as described in this Purchase Agreement), provided that all documents so transmitted have been duly and properly executed by the appropriate parties/signatories thereto which may be by electronic signature; and
- (6) notwithstanding anything contained in this Purchase Agreement to the contrary, in the event that Closing is to take place in accordance with the provisions of this Section 9.5, it is expressly understood and agreed by the parties hereto that an effective tender shall be deemed to have been validly made by the Vendor upon the Purchaser when the Vendor's Solicitors have:
 - (a) delivered all closing documents and/or funds to the Purchaser's Solicitors in accordance with the provisions of the Escrow Agreement including, if elected by the Vendor, by the method described in Section 9.1(4) of this Schedule;
 - (b) advised the Purchaser's Solicitors, in writing, that the Vendor is ready, willing and able to complete the transaction in accordance with the terms and provisions of this Purchase Agreement; and
 - (c) completed all steps required by TERS in order to complete this transaction that can be performed or undertaken by the Vendor's Solicitors without the cooperation or participation of the Purchaser's Solicitors, and which do not first require the Purchaser's Solicitors to complete information in the transfer/deed;

without the necessity of personally attending upon the Purchaser or the Purchaser's Solicitors with documents, keys and/or funds, and without any requirement to have an independent witness evidencing the foregoing.

PART X - CONDOMINIUM MATTERS

10.1 ACKNOWLEDGEMENT

The Purchaser hereby acknowledges and accepts that:

- during and until the date that is 20 years following the completion of all construction, marketing and selling of condominium units, and all leasing activities, within the Galleria Master Development, the Developers and any Alternate Entity, and any of their respective sales staff, agents, employees, contractors, invitees and customer service staff shall, without cost, have free, uninterrupted access to the common elements of the Condominium (including the Shared Facilities thereon), the Shared Facilities, any unsold Units and any Units owned by the Corporation for: (i) any construction or maintenance purposes; (ii) use of such areas for administrative and/or construction offices; (iii) for the purpose of any sales promotion and leasing; and (iv) use of such areas for a sales, rental, administrative, customer services office, advertising signage and displays and model suites for display purposes all as determined from time to time by the Developers in their collective and unfettered discretion;
- (2) the Vendor, or an affiliated company, may enter into a management agreement with the Corporation;
- (3) the size of the Condominium and/or the Purchased Home as represented by the sales personnel is measured in accordance with the requirements of Tarion Builder Bulletin 22 and, accordingly, may differ from measurements made using the Purchased Home boundaries set out in the Declaration and in addition that the dimensions, floor area, or square footage of the Purchased Home as set out in this Purchase Agreement, or in any addendum or schedule annexed hereto or in any material provided to the Purchaser,

including any sales literature of the Vendor, are approximate only and that in the event such dimensions, floor area or square footage are determined to be less or more than as set out in this Purchase Agreement, or in any addendum or schedule appended hereto or in any material provided to the Purchaser, the Purchaser agrees to accept the Condominium and/or the Purchased Home with such lesser or greater dimensions, floor area or square footage without any abatement of the Purchase Price;

- (4) any right to be exercised by or benefit granted to the Vendor in respect of the development of the Condominium or in any other manner related to the Building (including, without limitation, all easements, rights of way or rights to use any portion or portions of the Shared Facilities or common elements of the Condominium or the other components of the Galleria Master Development and all determinations to be made in respect of the Building) may be exercised or enjoyed by the Developers or any successor, assign, affiliate, related or associated entity, holding body corporate, subsidiary body corporate, agent or representative of a Developer (each, an "Alternate Entity" and collectively, the "Alternate Entities"), irrespective of whether or not any such Alternate Entity exists as at the date hereof and whether such rights are expressed to be in favour of any or all of, the Developers, or any Alternate Entity; and
- (5) the Vendor shall have the right, in its sole and unfettered discretion and without the consent of the Purchaser, at any time hereafter, to assign the Vendor's interest under this Purchase Agreement to any third party that the Vendor may select in its sole and unfettered discretion, and that upon written notice of such assignment being provided to the Purchaser, the Vendor shall automatically and without further formality, be released from any and all obligations or liabilities arising under this Purchase Agreement and the Purchaser agrees to complete the transaction with the assignee.

10.2 MODIFICATIONS TO CONDOMINIUM DOCUMENTS

- (1) The Purchaser agrees that the Purchaser's rights, obligations and ownership of the Purchased Home shall be governed by the terms, conditions, provisos, rights and responsibilities contemplated by, and contained in, the Condominium Documents. The Purchaser acknowledges that the Condominium Documents required by the Condo Act have not been registered by the Vendor and agrees that the Vendor may, in its sole and unfettered discretion, from time to time, make any modifications to the Condominium Documents in accordance with its own requirements, and the requirements of any Approving Authorities, any mortgagee, the Regional Surveyor, the Land Registry Office, or by virtue of the Condo Act to permit registration thereof, and the Purchaser hereby irrevocably consents to any such modifications and agrees to be bound by same notwithstanding that such modifications occur or are made after the entering into of this Purchase Agreement by the Purchaser.
- (2) The Vendor agrees to proceed with diligence to have the Condominium Documents registered and to fulfil all other requirements in order to bring the Corporation under the Condo Act as soon as reasonably possible.
- (3) The Purchaser acknowledges and agrees that notwithstanding any other term or provision contained in this Purchase Agreement to the contrary, in the event that one or more material changes are made to the Condominium Documents, the Purchased Home or to the Condominium to be developed on the Lands, the Purchaser's only remedies shall be those provided by Section 74 of the Condo Act, notwithstanding any rule of law or equity to the contrary, and the Purchaser shall not have, initiate or maintain any claim or cause of action for damages or specific performance of this Purchase Agreement, in connection with such material change or changes. If the Purchaser does not exercise any such remedies, the Purchaser agrees to accept title to the Purchased Home subject to the aforesaid changes to the Condominium Documents, the Purchased Home or the Condominium to be developed on the Lands. The Purchaser hereby irrevocably consents to any amendment of or change to the Condominium Documents to reflect either structural changes or boundary amendments or any variations in the number of Units or floors, the numbering of the Purchased Home (including its Suite number) and the numbering of Levels.

10.3 DECLARANT'S INTENTION TO LEASE UNSOLD UNITS

The Vendor as either proposed or actual declarant hereby reserves the right to lease, from time to time, any unsold proposed Unit or Units in the Condominium and this Section 10.3 shall constitute notice to the Purchaser, as the registered owner of the Purchased Home, of the Vendor's intention.

PART XI - GENERAL

11.1 POWER OF ATTORNEY

(1) The Purchaser hereby irrevocably constitutes and appoints the Vendor to be and act as the Purchaser's lawful attorney, in the Purchaser's name, place and stead in order to execute the deposit receipt issued pursuant to The ONHWP Act, and the regulations thereunder, as well as the application form for the HST new housing rebate, any excess condominium

deposit insurance (and related documents) issued by any insurer or entity providing prescribed security for the Purchaser's deposit monies pursuant to the Condo Act or as otherwise provided in this Purchase Agreement and all documents necessary to fully release all interest of the Purchaser in the Purchased Home, the Condominium and the Lands, pursuant to Section 8.1(6) of this Schedule and to do such other things as are provided for in this Purchase Agreement, all in accordance with the provisions of the *Powers of Attorney Act* (Ontario), as amended from time to time, and any Regulations made thereunder. The Purchaser hereby confirms and agrees that this power of attorney may be exercised by the Vendor during any subsequent legal incapacity of the Purchaser and that such appointment and power of attorney, being coupled with an interest, shall be irrevocable and shall not be revoked by any action of the Purchaser.

- (2) If any documents required to be executed and delivered by the Purchaser to the Vendor are, in fact, executed by a third party other than the Vendor appointed as the attorney for the Purchaser, then the power of attorney appointing such person must be registered in the Land Registry Office, and a duplicate registered copy thereof (together with a statutory declaration sworn by the attorney for the Purchaser confirming that such power of attorney has not been revoked) shall be delivered to the Vendor along with such documents. The Purchaser and/or the Purchaser's attorney shall execute such other documents and cause the Purchaser's and/or the Purchaser's attorney's solicitor to execute such documents as the Vendor or the Vendor's Solicitors may in their sole and unfettered discretion require and the Vendor shall be entitled to refuse to deal with any such attorney in the event that the Purchaser, the Purchaser's attorney or their respective solicitors do not provide such documents. In addition, any additional requirements of TERS, the Law Society of Ontario or of any Approving Authority, in respect of powers of attorney, shall be complied with by the Purchaser, the Purchaser's Solicitors, the attorney and the attorney's solicitors.
- (3) Where a third party has been appointed as the attorney for the Purchaser for the purposes of executing any documents contemplated by this Purchase Agreement, the Purchaser must provide an address or contact number of such attorney to the Vendor in accordance with Section 14 of the Addendum. Thereafter, any written notices required or desired to be delivered to the Purchaser in accordance with Section 11.2 of this Schedule may be given to such attorney, in lieu of the Purchaser and shall be deemed to have been received by the Purchaser when so delivered to the Purchaser's attorney.
- (4) Each of the individuals comprising the Purchaser, if more than one (hereinafter referred to as the "Donor") hereby constitutes and appoints the other (hereinafter referred to as the "Donee") to be and act as the Donor's lawful agent and attorney, in order to execute the Purchaser's acknowledgment of receipt of the Disclosure Statement (or amended Disclosure Statement) and the fully executed copy of this Purchase Agreement, and/or for the purposes of receiving written notices required or desired to be delivered by the Vendor in accordance with Section 11.2 of this Schedule, amendments to this Purchase Agreement and any other documents relating to Extras, colour and material selections or changes. In accordance with the provisions of the Powers of Attorney Act (Ontario), as amended from time to time, the Donor hereby confirms and agrees that the power of attorney may be exercised by the Donee during any subsequent legal incapacity of the Donor, and shall only be revoked upon the death of the Donor or upon the Donor delivering written notice of such revocation to the Vendor. The Donor hereby confirms that the Donor has or may have multiple powers of attorney and that this power of attorney does not revoke any other power of attorney granted by the Donor in existence as of the date hereof and that the Donor may give additional powers of attorney in the future.

11.2 **NOTICE**

- (1) Any notice required to be given pursuant to the Addendum shall be delivered in the manner required by the Addendum. The Purchaser is hereby advised that the Vendor shall be entitled to send notices or communications to the Purchaser at the address, fax number or email address set out on page 2 of the Addendum, unless the Purchaser provides valid notice of a change in their contact information in the manner contemplated in the Addendum.
- (2) Any other notice given pursuant to the terms of this Purchase Agreement shall be deemed to have been validly given if it is in writing and is delivered personally, electronically pursuant to Section 9.1(4) of this Schedule, by prepaid mail, fax, or electronic mail to the attention of the Purchaser or the Purchaser's Solicitors to their respective addresses in this Purchase Agreement or to the Property after the Firm Occupancy Date or to the Vendor or the Vendor's Solicitors at their respective addresses in this Purchase Agreement, or such other address as may from time to time be given by notice in accordance with this Section 11.2. Such notice shall be deemed received the same day if delivered personally, electronically pursuant to Section 9.1(4) of this Schedule, by electronic mail or by fax and upon the third day following sending same by prepaid mail, excluding weekends and statutory holidays.

11.3 SERVICE AGREEMENTS

The purchaser agrees to enter into any agreements required by any Smart Metering Provider, Smart Home Provider or Utility Service Provider in order to acknowledge and accept the terms of such parties in respect of their delivery of any Smart Metering, Utility Services or Smart Home Services provided in respect of the Purchased Home or any devices or equipment leased or delivered therewith.

11.4 MISCELLANEOUS

- (1) **Non-Merger**. The provisions and covenants of this Purchase Agreement which require fulfilment by the Purchaser or the Vendor after the Closing Date shall remain in full force and effect until such matters are completed, notwithstanding the delivery and registration of the transfer of title and all representations and acknowledgements of the Purchaser shall remain in effect and shall not merge on the Closing Date.
- (2) Entire Agreement. This offer, when accepted, shall constitute a binding agreement of purchase and sale and constitutes the entire agreement and understanding between the Vendor and the Purchaser with respect to the subject matter of this Purchase Agreement and supersedes all prior negotiations between the Vendor and the Purchaser, whether written or verbal, with respect to the subject matter of this Purchase Agreement. There is no representation, warranty, collateral agreement or condition affecting this Purchase Agreement, the Condominium or the Purchased Home, for which the Vendor can be held responsible or liable in any way, whether contained, portrayed, illustrated or represented by, or in, any plan, drawing, brochure, display, model or any other sales/marketing material, or made by a sales agent other than as expressed herein in writing and this Purchase Agreement contains all of the representations, warranties, undertakings and agreements of the parties. This is notwithstanding that the Vendor's sales agents and representatives and all persons working in the Vendor's sales office are acting on behalf of the Vendor and are representing the interests of the Vendor and are compensated through and/or by the Vendor. This Purchase Agreement may not be amended other than in writing and executed by the Purchaser and the Vendor or by their respective solicitors who are hereby expressly authorized in that regard. The Purchaser is encouraged to have this Purchase Agreement reviewed by the Purchaser's Solicitors prior to signing same or prior to the end of the rescission period provided by the Condo Act. In the event there is a conflict between any two or more provisions in this Purchase Agreement, the Vendor shall determine which conflicting provision(s) prevail(s).
- (3) **HST on Fees**. Any payment, charge or fee referred to in this Purchase Agreement for which HST is exigible but not referred to shall be deemed not to include HST, and any such exigible HST shall be payable to the party legally obliged to collect such HST on such payment, charge or fee.
- (4) Binding on Heirs and Death of Purchaser. This Purchase Agreement shall inure to the benefit of and be binding upon the parties hereto, the successors and assigns of the Vendor and the heirs, executors, administrators, estate trustees and permitted assigns of the Purchaser. Notwithstanding the generality of the foregoing, should the Purchaser die after the time period prescribed by Section 73(2) of the Condo Act and before the Firm Occupancy Date or the Closing Date (as the case may be), the Purchaser's estate shall furnish the Vendor with a notarial copy of a Certificate of Appointment of Estate Trustee prior to the Firm Occupancy Date or the Closing Date (as the case may be) and the estate of the deceased Purchaser shall continue to be bound by and perform all obligations of the Purchaser under the Purchase Agreement. Any failure to provide such Certificate of Appointment of Estate Trustee prior to the Firm Occupancy Date or the Closing Date (as the case may be) shall be a default under this Purchase Agreement, entitling the Vendor to terminate the transaction in accordance with the provisions of Section 8.1 of this Schedule and the provisions of Section 8.1 of this Schedule shall otherwise apply to such default. If the Purchaser is more than one person, the Vendor shall have the option of treating the surviving purchaser as the Purchaser under this Purchase Agreement in all respects and in such event the estate of the deceased Purchaser, the Purchaser's estate and estate trustee hereby releases the Vendor from any claim whatsoever under this Purchase Agreement.
- (5) **Third Parties**. The Purchaser acknowledges and agrees that notwithstanding any rights which the Purchaser might have at law or in equity arising of this Purchase Agreement, the Purchaser shall not assert any of such rights, nor have any claim or cause of action as a result of any matter or thing arising under or in connection with this Purchase Agreement against any person, firm, corporation or other legal entity, other than the person, firm, corporation or legal entity specifically named or defined as the Vendor herein, even though the Vendor may be found to be a nominee, agent or representative of another person, firm, corporation or other legal entity, and this acknowledgment and agreement may be pleaded as an estoppel and bar against the Purchaser in any action or proceeding brought by the Purchaser to assert any of such rights, claims or causes of action. Furthermore, the Purchaser and the Vendor acknowledge and agree that this Purchase Agreement shall be deemed to be a contract under seal.
- (6) **Assignment by Vendor**. At any time prior to Closing, the Vendor shall be permitted to assign this Purchase Agreement (and its rights, benefits and interests hereunder) to any person, firm, partnership or corporation, or combination thereof, and upon any such assignee assuming all obligations under this Purchase Agreement and notifying the

Purchaser or the Purchaser's Solicitors of such assignment, the Vendor named herein shall thereupon be automatically released from, and relieved of, all obligations and liabilities to the Purchaser arising from this Purchase Agreement, and such assignee shall be deemed for all purposes to be the vendor herein as if it had been an original party to this Purchase Agreement, in the place and stead of the Vendor.

- (7) **Time of Essence**. Time shall be of the essence of this Purchase Agreement and any waiver, extension, abridgement or other modification of any time provisions shall not be effective unless it is made in accordance with the Addendum, is in writing and is signed by the parties hereto or by their respective solicitors who are hereby expressly authorized in that regard.
- (8) **Number and Gender**. This Purchase Agreement shall be read with all changes of gender and number required by the context.
- (9) **Headings**. The headings of this Purchase Agreement form no part hereof and are inserted for convenience of reference only.
- (10) **Governing Law**. This Purchase Agreement is made pursuant to and shall be governed by and construed in accordance with the laws in effect in the Province of Ontario.
- (11) Severability. If any covenant, obligation or provision contained in this Purchase Agreement, or the application thereof to any person or circumstance, shall, to any extent, be illegal, invalid or unenforceable, the remainder of this Purchase Agreement or the application of such covenant, obligation or provision to person or circumstances other than those as to which it is held invalid or unenforceable shall not be affected thereby and each covenant, obligation or provision of this Purchase Agreement shall be separately valid and enforceable to the fullest extent permitted by law.
- (12) **Joint and Several Liability**. If two or more individuals, corporations, partnerships or other business associations (or any combination of two or more thereof) execute this Purchase Agreement as Purchaser, the liability and obligations of each such individual, corporation, partnership or other business association hereunder shall be joint and several. Furthermore, if the Purchaser named herein is a partnership or other business association, the members of which are by virtue of statutory or general law subject to personal liability, the liability of each such member shall be joint and several.
- (13) Title. The Purchaser hereby agrees to notify the Vendor's Solicitors as to the manner in which title to the Purchased Home is to be taken and the date(s) of birth of all persons taking title to the Purchased Home, and in the event that the Purchaser fails to so notify the Vendor's Solicitors at least thirty (30) days prior to the Firm Occupancy Date, then the Vendor or the Vendor's Solicitors shall be entitled and are hereby directed to prepare the Transfer/Deed of title to the Purchased Home in the name of the Purchaser noted on the Agreement Cover and the Purchaser agrees to accept the conveyance in such manner, and will be bound thereby and shall be estopped from requiring any changes to the Transfer/Deed so prepared.
- (14) **T-5 Return**. The Purchaser acknowledges that the Vendor is required to issue to the Purchaser a T-5 interest income return in accordance with the provisions of the *Income Tax Act* (Canada), as amended from time to time, in respect of any interest accrued to, or earned by, the Purchaser under the terms and provisions of this Purchase Agreement and/or the Condo Act.
- (15) **Delayed Occupancy Date**. In the event the Vendor cannot provide Occupancy on the Firm Occupancy Date, then references in this Purchase Agreement to the Firm Occupancy Date shall be deemed to mean the Delayed Occupancy Date and any action required by the Vendor or the Purchaser on the Firm Occupancy Date shall, in such case, occur on the Delayed Occupancy Date.
- (16) **Unavoidable Delay**. In the event that the Vendor is unable to fulfil any obligation hereunder by reason of any Unavoidable Delay, the time period for fulfilment of such obligation by the Vendor shall be in accordance with Section 5 of the Addendum.
- (17) **Further Assurances**. The Purchaser agrees to execute and deliver from time to time and at the request of the Vendor or the Vendor's Solicitors such further assurances (including further closing documents) as the Vendor or the Vendor's Solicitors shall reasonably require in order to more effectually carry out the intent of this Purchase Agreement.
- (18) **Electronic Execution**. This Purchase Agreement and any amendment or supplement to this Purchase Agreement may be signed by any party hereto electronically by the application of an electronic signature; and delivery of same may be completed by electronic mail or any other electronic transmission.
- (19) **Counterparts**. This Purchase Agreement, and any amendment or supplement to this Purchase Agreement may be executed and delivered (in accordance with Section 11.4(18)

or otherwise) in counterparts and all such counterparts taken together shall constitute one and the same Purchase Agreement.

- (20) **Execution not in English**. In the event that any of the documents delivered by the Vendor's Solicitor to the Purchaser or Purchaser's Solicitors for execution by the Purchaser are signed in foreign characters or lettering which bear no relation to the Purchaser's name in English, as same appears in the document(s) being executed then the Purchaser agrees to ensure that the Purchaser's signature is duly witnessed and that a statement is added in English by such witness confirming that the witness saw the Purchaser sign the document after same had been read to the Purchaser and the Purchaser appeared to fully understand same.
- (21) **Not Purchased "in trust"**. The Purchaser acknowledges and agrees that the purchase of Units within the Condominium "in trust" for another party is prohibited.
- (22)Utility Costs. The Purchaser acknowledges that, as a result of uncertainty in the natural gas, hydro and water distribution markets, the Vendor's reasonable assumptions regarding such utility costs for utility services supplied to common elements may be incorrect as a result of circumstances which are not capable of being accurately predicted as of the date of this Purchase Agreement and which are beyond the Vendor's control. Consequently, prior to registration of the Condominium, the projected costs for such utilities for the first year shall be updated to reflect market conditions as of the Closing Date as an alternative (in the Vendor's sole and unfettered discretion) to applying the inflation factor referenced in the Disclosure Statement and Budget Statement. The Budget Statement and Common Expenses applicable to each Unit shall be revised accordingly. The Purchaser specifically acknowledges and agrees that any increase in any such utility costs from that which was originally represented in the Budget Statement shall not be the responsibility of the Vendor, despite section 75 of the Condo Act. The Purchaser acknowledges that the possibility of an increase in such utility costs has been properly disclosed and, consequently, any increase shall not constitute a material change to the Disclosure Statement and Budget Statement. In addition, the Purchaser agrees that this acknowledgement may be pleaded by the Vendor as complete defence to any application or objection raised by the Purchaser in this regard.
- (23) **Execution by Corporation**. Where the Purchaser is a corporation, the execution of this Purchase Agreement by the principal or principals of such corporation shall be deemed and construed to constitute the personal guarantee of such person or person so signing with respect to the obligations of the Purchaser herein, and such person or persons shall also be correspondingly obliged to unconditionally guarantee any mortgage(s) required to be given by the Purchaser on Closing Date, in accordance with the provisions hereof.
- (24) **Execution of Closing Documents**. The Purchaser agrees to execute all of the Vendor's standard closing documents which are presented to the Purchaser by the Vendor for Closing or for Occupancy whether or not they are specifically referred to in this Purchase Agreement, and any other documents which are requested by the Vendor, acting reasonably.

PART XII – PURCHASER'S CONSENT TO THE COLLECTION AND LIMITED USE OF PERSONAL INFORMATION

12.1 PURCHASER'S CONSENT

- (1) For the purposes of facilitating compliance with the provisions of any applicable Federal or Provincial privacy legislation (including the *Personal Information Protection and Electronic Documents Act* (Ontario), as amended from time to time), the Purchaser hereby consents to the Vendor's collection and use of the Purchaser's personal information as necessary and sufficient to enable the Vendor to proceed with the Purchaser's purchase of the Purchased Home. The Vendor shall not sell or otherwise provide or distribute such personal information to anyone other than the following parties and entities listed in this Section 12.1.
- (2) Name, Address, Date of Birth and Marital Status

The Vendor will collect the Purchaser's name, home address, e-mail address, facsimile and telephone numbers, date of birth and marital status, as required to complete the transaction. The Vendor may disclose this information (as well as the municipal address/Purchased Home number and sale price) to:

- (a) any companies or legal entities that are associated with, related to or affiliated with the Vendor, other future condominium declarants that are likewise associated with, related to or affiliated with the Vendor (or with the Vendor's parent/holding company) and are developing one or more other condominium projects or communities that may be of interest to the Purchaser or members of the Purchaser's family, for the limited purposes of marketing, advertising and/or selling various products and/or services to the Purchaser and/or members of the Purchaser's family;
- (b) one or more third party data processing companies which handle or process marketing campaigns on behalf of the Vendor or other companies that are associated with, related to or affiliated with the Vendor, and who may send (by e-mail or other

means) promotional literature and brochures about new condominiums and/or related services to the Purchaser or members of the Purchaser's family;

- (c) the municipality and if applicable, regional municipality in which the Condominium will be located;
- (d) any relevant Municipal or Provincial government authorities or agencies, including:
 - (i) Tarion;
 - (ii) the Land Registry Office;
 - (iii) Municipal Property Assessment Corporation; and
 - (iv) the Ministry of Finance for the Province of Ontario, including with respect to Land Transfer Tax:
- (e) any relevant Federal or government authorities or agencies, including Canada Revenue Agency, including with respect to HST matters and other tax reporting purposes and any matters concerning income tax information returns or non-resident withholding tax returns;
- (f) the Vendor's Solicitors, to facilitate Occupancy and/or the Closing, who may disclose such information to internet service providers in connection with the distribution and execution of closing documents, as well as to Teranet Inc., where the transfer of title with respect to the Purchaser's Purchased Home occurs through TERS;
- (g) any warranty bond provider or excess condominium deposit insurer, where applicable;
- (h) any financial institutions providing or wishing to provide mortgage financing, banking or other financial or related services to the Purchaser or members of the Purchaser's family with respect to the Purchased Home, including the Vendor's construction lender, Tarion, or any warranty bond provider or excess condominium insurer required in connection with the construction, development or financing of the Condominium, or the financing of the Purchaser's acquisition of the Purchased Home from the Vendor;
- (i) any insurance companies, including title insurance companies, insurance trustee, reserve fund consultant and performance audit consultant performing insurance coverage, insurance coverage advice or financial management advice with respect to any part of the development or the Condominium;
- (j) any tradespeople, suppliers, subtrades or surveyors working on the development in order to complete the Purchased Home in accordance with the Purchaser's décor specifications, to contact the Purchaser to offer extended warranties and other services, to install any Extras as requested by the Purchaser and for after sales service purposes;
- (k) any Utility Service Providers, Smart Metering Providers, Smart Home Services Providers and the providers of cable television, telephone, internet, telecommunication, security alarm systems, hydro, water or any other related services to the Property;
- (I) the Corporation, for the purposes of maintaining the Corporation's records of owners of Units in the Condominium and to the Corporation's property manager, for general Condominium management and administration purposes; and
- (m) any person, where the Purchaser further consents to such disclosure or such other disclosures as required by law.
- (3) Banking and other Financial Information

In addition to the information contained in Section 12.1 of this Schedule, the Vendor may collect banking and other financial information with respect to the Purchaser, including net worth and annual income statements, and may disclose this information:

- (a) to credit bureaus for the purpose of evaluating the Purchaser's credit worthiness; and
- (b) to any financial institution which is providing, or wishing to provide, financial services such as mortgage financing, banking or any other financial or related services to the Purchaser or members of the Purchaser's family including the Vendor's construction lender(s), the project monitor, the Vendor's designated construction lender(s), Tarion and/or any warranty bond provider and/or excess condominium deposit insurer, required in connection with the development and or construction financing of the

Condominium or the financing of the Purchaser's acquisition of the Purchased Home from the Vendor.

(4) Residency Status, Family Relationship and SIN

In addition to marital status information collected as referred to above, the Vendor may also collect information relating to the Purchaser's residency status and Social Insurance Number for the purposes of submitting to Canada Revenue Agency, a T-5 interest income tax information return and/or the NR4 non-resident withholding tax information return and for reporting in regard to HST.

(5) Smart Devices

- (a) The Purchaser acknowledges that the Smart Home Devices and Smart Tech Devices referred to in the Disclosure Statement (referred to in this Section as the "Smart Devices") are anticipated to be wirelessly connected to the internet and may continuously collect information, including information about when a Residential Unit is locked and unlocked, when packages are delivered to Residents and information regarding the vehicles of visitors to the Condominium. The Purchaser hereby consents to all such information collection.
- (b) The Purchaser acknowledges that the Smart Devices are anticipated to be administered by third party equipment and service providers, which will have access to the data collected by the Smart Devices and may use such data for their own purposes, including evaluation of usage of the Smart Devices, potentially offering additional products or services as well as selling aggregated information to other organizations. The Purchaser hereby consents to all such administration and use of such information. Such information may also be shared with the Declarant and the other Developers and the Purchaser hereby consents to all such information sharing.
- (c) The Purchaser acknowledges that it is currently anticipated that the Condominium will enter into agreements with the providers for the Smart Devices as set out in the Disclosure Statement but that the list of providers may change from time-to-time, and that an updated list can be obtained at any time as described in the Disclosure Statement.
- (d) The Purchaser acknowledges that some of the providers may be located outside of Canada, and therefore, information about Smart Device usage may be transmitted to and stored in other countries, and may be accessible to courts, law enforcement, and national authorities in such other jurisdictions. The Purchaser hereby consents to such transmission of all such information.

The Purchaser acknowledges that information collected by the Smart Devices could be used to infer the times when Residents are, and are not, in their Residential Units. Also, the information is transmitted online, and although precautions are taken to protect the information (e.g., encryption in transit), it is possible that it could be intercepted by unknown third parties. In addition, if the credentials for the Smart Devices are compromised, a third party could take control of them (e.g., have the ability to open locks). Residents will be able to change the credentials for each of the Smart Devices by going online, which may reduce the risk of compromise if Residents choose a secure password and ensure that the password is protected from unauthorized access.

(e) The Purchaser acknowledges that that Residents may not have the option to disable any or all of the Smart Devices, but with respect to the Smart Devices which control door locks, Residents will have the option of using a manual key for entry instead of using the Smart Device application.

(6) Consent to Disclosure

Unless the Purchaser indicates otherwise by contacting the Privacy Officer of the Vendor c/o the Vendor's Legal Department at (416) 223-4403, the Purchaser agrees to:

- (a) the disclosure by the Vendor of the Purchaser's name, home address, e-mail address, facsimile and telephone numbers to financial institutions and insurance companies which wish to provide financing and insurance services to the Purchaser and/or members of the Purchaser's family; and
- (b) the disclosure by the Vendor of the Purchaser's name, home address, e-mail address, facsimile and telephone numbers to companies related to or affiliated with, the Vendor and the Vendor's business partners and agents, in order to bring to the Purchaser's attention other residential communities that may be of interest.

(7	Consent to receiving	commercia	l electronic mes	ssades (Anti-Sr	oam)	١

The Vendor may from time to time wish to send the Purchaser electronic messages including but not limited to emails and telephone calls, regarding products and services being offered by the Vendor, or the Vendor's business partners and agents, that might be of interest to the Purchaser. By initialling below this paragraph, the Purchaser expressly consents to receive these electronic messages. This consent may be withdrawn at any time by contacting the Vendor's Anti-Spam and privacy officer at (416) 223-4403, or by following the unsubscribe mechanism set out in the electronic message.

I consent to receiving	commercial elec	tronic mes	sages fror		he Vendor. Purchaser's Initials		
By executing this Purchase disclosure of the information n		Purchaser	consents	to the	collection,	use	and
Purchaser's Initials	Purchaser's Ir	nitials	Ver	ndor's In	itials		



Property: GALLERIA 01

UNIT NO.: LEVEL NO.:

SUITE NO.:

Statement of Critical Dates

Delayed Occupancy Warranty

This Statement of Critical Dates forms part of the Addendum to which it is attached, which in turn forms part of the agreement of purchase and sale between the Vendor and the Purchaser relating to the Property. **The Vendor must complete all blanks set out below. Both the Vendor and Purchaser must sign this page.**

NOTE TO HOME BUYERS: Please visit Tarion's website: www.tarion.com for important information about all of Tarion's warranties including the Delayed Occupancy Warranty, the Pre-Delivery Inspection and other matters of interest to new home buyers. You can also obtain a copy of the Homeowner Information Package which is strongly recommended as essential reading for all home buyers. The website features a calculator which will assist you in confirming the various Critical Dates related to the occupancy of your home.

VENDOR	GALLERIA DEVELOPMENTS (BLOCK 5 A) INC.	
PURCHASER	Full Name(s)	
	Full Name(s)	
	Occupancy Date, which is the date that the Vendor anticipates appleted and ready to move in, is:	November 24, 2023
subsequent Tentat	delay Occupancy on one or more occasions by setting a ive Occupancy Date , in accordance with section 1 of the proper written notice as set out in section 1.	
with at least 90 da	days after the Roof Assembly Date (as defined in section 12), ys prior written notice, the Vendor shall set either (i) a Final icy Date ; or (ii) a Firm Occupancy Date .	
Occupancy Date is	ments signed after the Roof Assembly Date, the First Tentative inapplicable and the Vendor shall instead elect and set either a upancy Date or Firm Occupancy Date. or	theday of, 20 Final Tentative Occupancy Date
		theday of, 20 Firm Occupancy Date
Occupancy by the Firm Occupancy I	s a Final Tentative Occupancy Date but cannot provide Final Tentative Occupancy Date, then the Vendor shall set a Date that is no later than 120 days after the Final Tentative ith proper written notice as set out in section 1 below.	Time Codepancy Bate
Purchaser is entitle	ot provide Occupancy by the Firm Occupancy Date, then the od to delayed occupancy compensation (see section 7 of the Vendor must set a Delayed Occupancy Date which cannot be de Occupancy Date.	
The Outside Occup to provide Occupand	Dancy Date , which is the latest date by which the Vendor agrees cy, is:	November 30, 2027
Changing an Occup the Purchaser's con with section 1 of the Notice of a delay be later than: (i.e., at least 90 days	for an Occupancy Delay ancy date requires proper written notice. The Vendor, without sent, may delay Occupancy one or more times in accordance Addendum and no later than the Outside Occupancy Date. yond the First Tentative Occupancy Date must be given no before the First Tentative Occupancy Date), or else the First Date automatically becomes the Firm Occupancy Date.	August 25, 2023
can terminate the tra "Purchaser's Term agreement, will end If the Purchaser ter Period, then the Pu to a full refund of all Addendum). Note: Any time a Critica the parties must refer to calculate revised Critica	omplete by the Outside Occupancy Date, then the Purchaser ansaction during a period of 30 days thereafter (the ination Period "), which period, unless extended by mutual on: minates the transaction during the Purchaser's Termination rchaser is entitled to delayed occupancy compensation and monies paid plus interest (see sections 7, 10 and 11 of the all Date is set or changed as permitted in the Addendum, other Critical Dates of the most recent revised Statement of Critical Dates; or agreement or writted Dates using the formulas contained in the Addendum. Critical Dates can a	en notice that sets a Critical Date, and
delays (see section 5 o		•
• -	_day of, 20 PURCHASER:	
		



Addendum to Agreement of Purchase and Sale

Delayed Occupancy Warranty

This addendum, including the accompanying Statement of Critical Dates (the "Addendum"), forms part of the agreement of purchase and sale (the "Purchase Agreement") between the Vendor and the Purchaser relating to the Property. This Addendum is to be used for a transaction where the home is a condominium unit (that is not a vacant land condominium unit). This Addendum contains important provisions that are part of the delayed occupancy warranty provided by the Vendor in accordance with the *Ontario New Home Warranties Plan Act* (the "ONHWP Act"). If there are any differences between the provisions in the Addendum and the Purchase Agreement, then the Addendum provisions shall prevail. PRIOR TO SIGNING THE PURCHASE AGREEMENT OR ANY AMENDMENT TO IT, THE PURCHASER SHOULD SEEK ADVICE FROM A LAWYER WITH RESPECT TO THE PURCHASE AGREEMENT OR AMENDING AGREEMENT, THE ADDENDUM AND THE DELAYED OCCUPANCY WARRANTY.

Tarion recommends that Purchasers register on Tarion's **MyHome** on-line portal and visit Tarion's website – **tarion.com**, to better understand their rights and obligations under the statutory warranties.

The Vendor shall complete all blanks set out below.

The vendor	Shan compi	ete ali biariks set out below.				
VENDOR	GALLERIA	DEVELOPMENTS (BLOCK 5 A) INC				
	Full Name(s) B47753		2235 Sheppard Avenue East			
		istration Number 4403	Address Toronto	Ontario	rio M2J 5B5 nce Postal Code nce Postal Code	
	Phone		City	Province	Postal Code	
	(416) 223-	8246				
	Fax		Email*			
PURCHASER						
	Purchaser Purchaser					
	Address	Purchaser 1 Home:	City Purchaser 1	Province	Postal Code	
	Phone	Purchaser 2 Home:	Purchaser 2			
		Purchaser 1 Fax:	Purchaser 1	Email:		
		Purchaser 2 Fax:	Purchaser 2	Email:		
PROPERTY	DESCRIPTION	ON				
	1245 Dupo	nt Street				
	Municipal A					
	Toronto			Ontario		
	City			Province	Postal Code	
		Description Registered Plan 61-York, desi	ignated as Part 1	1 45 and 54 on Referen	ce Plan 66R30758	
	being part of	of PIN 21313-0003 (LT)	griatou do r dit	1, 10, 414 01 011 140101011	55 1411 551 551 5 5,	
INFORMATI	ON REGARI	DING THE PROPERTY				
The Vendor	confirms that	:				
If no, the	Vendor shal	ned Formal Zoning Approval for t I give written notice to the Purcha val for the Building is obtained.	•	ys after the date that	∕ Yes ○ No	
` '	cement of Copril, 2020.	onstruction: □has occurred; or □	is expected to	occur by the 30th		
The Vendor s Construction	•	tten notice to the Purchaser within	n 10 days after th	he actual date of Comme	encement of	

*Note: Since important notices will be sent to this address, it is essential that you ensure that a reliable email address is provided and that your computer settings permit receipt of notices from the other party.

CONDO TENTATIVE - 2012 Page 2 of 12





SETTING AND CHANGING CRITICAL DATES

1. Setting Tentative Occupancy Dates and the Firm Occupancy Date

- (a) **Completing Construction Without Delay**: The Vendor shall take all reasonable steps to complete construction of the Building subject to all prescribed requirements, to provide Occupancy of the home without delay, and, to register without delay the declaration and description in respect of the Building.
- (b) **First Tentative Occupancy Date**: The Vendor shall identify the First Tentative Occupancy Date in the Statement of Critical Dates attached to this Addendum at the time the Purchase Agreement is signed.
- (c) Subsequent Tentative Occupancy Dates: The Vendor may, in accordance with this section, extend the First Tentative Occupancy Date on one or more occasions, by setting a subsequent Tentative Occupancy Date. The Vendor shall give written notice of any subsequent Tentative Occupancy Date to the Purchaser at least 90 days before the existing Tentative Occupancy Date (which in this Addendum may include the First Tentative Occupancy Date), or else the existing Tentative Occupancy Date shall for all purposes be the Firm Occupancy Date. A subsequent Tentative Occupancy Date can be any Business Day on or before the Outside Occupancy Date.
- (d) Final Tentative Occupancy Date: By no later than 30 days after the Roof Assembly Date, the Vendor shall by written notice to the Purchaser set either (i) a Final Tentative Occupancy Date; or (ii) a Firm Occupancy Date. If the Vendor does not do so, the existing Tentative Occupancy Date shall for all purposes be the Firm Occupancy Date. The Vendor shall give written notice of the Final Tentative Occupancy Date or Firm Occupancy Date, as the case may be, to the Purchaser at least 90 days before the existing Tentative Occupancy Date shall for all purposes be the Firm Occupancy Date. The Final Tentative Occupancy Date or Firm Occupancy Date, as the case may be, can be any Business Day on or before the Outside Occupancy Date. For new Purchase Agreements signed after the Roof Assembly Date, the Vendor shall insert in the Statement of Critical Dates of the Purchase Agreement either: a Final Tentative Occupancy Date; or a Firm Occupancy Date
 (e) Firm Occupancy Date: If the Vendor has set a Final Tentative Occupancy Date but cannot provide
- (e) Firm Occupancy Date: If the Vendor has set a Final Tentative Occupancy Date but cannot provide Occupancy by the Final Tentative Occupancy Date then the Vendor shall set a Firm Occupancy Date that is no later than 120 days after the Final Tentative Occupancy Date. The Vendor shall give written notice of the Firm Occupancy Date to the Purchaser at least 90 days before the Final Tentative Occupancy Date, or else the Final Tentative Occupancy Date shall for all purposes be the Firm Occupancy Date. The Firm Occupancy Date can be any Business Day on or before the Outside Occupancy Date.
- (f) **Notice**: Any notice given by the Vendor under paragraph (c), (d) or (e) must set out the stipulated Critical Date, as applicable.

2. Changing the Firm Occupancy Date - Three Ways

- (a) The Firm Occupancy Date, once set or deemed to be set in accordance with section 1, can be changed only:
 - (i) by the Vendor setting a Delayed Occupancy Date in accordance with section 3;
 - (ii) by the mutual written agreement of the Vendor and Purchaser in accordance with section 4; or
 - (iii) as the result of an Unavoidable Delay of which proper written notice is given in accordance with section 5.
- (b) If a new Firm Occupancy Date is set in accordance with section 4 or 5, then the new date is the "Firm Occupancy Date" for all purposes in this Addendum.

3. Changing the Firm Occupancy Date - By Setting a Delayed Occupancy Date

- (a) If the Vendor cannot provide Occupancy on the Firm Occupancy Date and sections 4 and 5 do not apply, the Vendor shall select and give written notice to the Purchaser of a Delayed Occupancy Date in accordance with this section, and delayed occupancy compensation is payable in accordance with section 7.
- (b) The Delayed Occupancy Date may be any Business Day after the date the Purchaser receives written notice of the Delayed Occupancy Date but not later than the Outside Occupancy Date.
- (c) The Vendor shall give written notice to the Purchaser of the Delayed Occupancy Date as soon as the Vendor knows that it will be unable to provide Occupancy on the Firm Occupancy Date, and in any event at least 10 days before the Firm Occupancy Date, failing which delayed occupancy compensation is payable from the date that is 10 days before the Firm Occupancy Date, in accordance with paragraph 7(c). If notice of a new Delayed Occupancy Date is not given by the Vendor before the Firm Occupancy Date, then the new Delayed Occupancy Date shall be deemed to be the date which is 90 days after the Firm Occupancy Date.
- (d) After the Delayed Occupancy Date is set, if the Vendor cannot provide Occupancy on the Delayed Occupancy Date, the Vendor shall select and give written notice to the Purchaser of a new Delayed Occupancy Date, unless the delay arises due to Unavoidable Delay under section 5 or is mutually agreed upon under section 4, in which case the requirements of those sections must be met. Paragraphs (b) and (c) above apply with respect to the setting of the new Delayed Occupancy Date.
- (e) Nothing in this section affects the right of the Purchaser or Vendor to terminate the Purchase Agreement on the bases set out in section 10.

4. Changing Critical Dates - By Mutual Agreement

(a) This Addendum sets out a framework for setting, extending and/or accelerating Critical Dates, which cannot be altered contractually except as set out in this section 4. Any amendment not in accordance with this section is voidable at the option of the Purchaser. For greater certainty, this Addendum does not restrict any extensions of the Closing date (i.e., title transfer date) where Occupancy of the home has already been given to the Purchaser.



- (b) The Vendor and Purchaser may at any time, after signing the Purchase Agreement, mutually agree in writing to accelerate or extend any of the Critical Dates. Any amendment which accelerates or extends any of the Critical Dates must include the following provisions:
 - (i) the Purchaser and Vendor agree that the amendment is entirely voluntary the Purchaser has no obligation to sign the amendment and each understands that this purchase transaction will still be valid if the Purchaser does not sign this amendment;
 - (ii) the amendment includes a revised Statement of Critical Dates which replaces the previous Statement of Critical Dates;
 - (iii) the Purchaser acknowledges that the amendment may affect delayed occupancy compensation payable; and
 - (iv) if the change involves extending either the Firm Occupancy Date or the Delayed Occupancy Date, then the amending agreement shall:
 - i. disclose to the Purchaser that the signing of the amendment may result in the loss of delayed occupancy compensation as described in section 7;
 - ii. unless there is an express waiver of compensation, describe in reasonable detail the cash amount, goods, services, or other consideration which the Purchaser accepts as compensation; and
 - iii. contain a statement by the Purchaser that the Purchaser waives compensation or accepts the compensation referred to in clause ii above, in either case, in full satisfaction of any delayed occupancy compensation payable by the Vendor for the period up to the new Firm Occupancy Date or Delayed Occupancy Date.

If the Purchaser for his or her own purposes requests a change of the Firm Occupancy Date or the Delayed Occupancy Date, then subparagraphs (b)(i), (iii) and (iv) above shall not apply.

- (c) A Vendor is permitted to include a provision in the Purchase Agreement allowing the Vendor a one-time unilateral right to extend a Firm Occupancy Date or Delayed Occupancy Date, as the case may be, for one (1) Business Day to avoid the necessity of tender where a Purchaser is not ready to complete the transaction on the Firm Occupancy Date or Delayed Occupancy Date, as the case may be. Delayed occupancy compensation will not be payable for such period and the Vendor may not impose any penalty or interest charge upon the Purchaser with respect to such extension.
- (d) The Vendor and Purchaser may agree in the Purchase Agreement to any unilateral extension or acceleration rights that are for the benefit of the Purchaser.

5. Extending Dates - Due to Unavoidable Delay

- (a) If Unavoidable Delay occurs, the Vendor may extend Critical Dates by no more than the length of the Unavoidable Delay Period, without the approval of the Purchaser and without the requirement to pay delayed occupancy compensation in connection with the Unavoidable Delay, provided the requirements of this section are met.
- (b) If the Vendor wishes to extend Critical Dates on account of Unavoidable Delay, the Vendor shall provide written notice to the Purchaser setting out a brief description of the Unavoidable Delay, and an estimate of the duration of the delay. Once the Vendor knows or ought reasonably to know that an Unavoidable Delay has commenced, the Vendor shall provide written notice to the Purchaser by the earlier of: 20 days thereafter; and the next Critical Date.
- (c) As soon as reasonably possible, and no later than 20 days after the Vendor knows or ought reasonably to know that an Unavoidable Delay has concluded, the Vendor shall provide written notice to the Purchaser setting out a brief description of the Unavoidable Delay, identifying the date of its conclusion, and setting new Critical Dates. The new Critical Dates are calculated by adding to the then next Critical Date the number of days of the Unavoidable Delay Period (the other Critical Dates changing accordingly), provided that the Firm Occupancy Date or Delayed Occupancy Date, as the case may be, must be at least 10 days after the day of giving notice unless the parties agree otherwise. Either the Vendor or the Purchaser may request in writing an earlier Firm Occupancy Date or Delayed Occupancy Date, and the other party's consent to the earlier date shall not be unreasonably withheld.
- (d) If the Vendor fails to give written notice of the conclusion of the Unavoidable Delay in the manner required by paragraph (c) above, then the notice is ineffective, the existing Critical Dates are unchanged, and any delayed occupancy compensation payable under section 7 is payable from the existing Firm Occupancy Date.
- (e) Any notice setting new Critical Dates given by the Vendor under this section shall include an updated revised Statement of Critical Dates.

EARLY TERMINATION CONDITIONS

6. Early Termination Conditions

- (a) The Vendor and Purchaser may include conditions in the Purchase Agreement that, if not satisfied, give rise to early termination of the Purchase Agreement, but only in the limited way described in this section.
- (b) The Vendor is not permitted to include any conditions in the Purchase Agreement other than: the types of Early Termination Conditions listed in Schedule A; and/or the conditions referred to in paragraphs (i), (j) and (k) below. Any other condition included in a Purchase Agreement for the benefit of the Vendor that is not expressly permitted under Schedule A or paragraphs (i), (j) and (k) below is deemed null and void and is not enforceable by the Vendor, but does not affect the validity of the balance of the Purchase Agreement.



- (c) The Vendor confirms that this Purchase Agreement is subject to Early Termination Conditions that, if not satisfied (or waived, if applicable), may result in the termination of the Purchase Agreement. ☑ Yes ☐ No
- (d) If the answer in (c) above is "Yes", then the Early Termination Conditions are as follows. The obligation of each of the Purchaser and Vendor to complete this purchase and sale transaction is subject to satisfaction (or waiver, if applicable) of the following conditions and any such conditions set out in an appendix headed "Early Termination Conditions":

Condition #1 (if applicable)

Description of the Early Termination Condition:

The Approving Authority (as that term is defined in Sched	ule A) is:		
The date by which Condition #1 is to be satisfied is the	day of	, 20	
Condition #2 (if applicable) Description of the Early Termination Condition:			
The Approving Authority (as that term is defined in Sched	ule A) is:		
The date by which Condition #2 is to be satisfied is the	day of	, 20	

The date for satisfaction of any Early Termination Condition may be changed by mutual agreement provided in all cases it is set at least 90 days before the First Tentative Occupancy Date, and will be deemed to be 90 days before the First Tentative Occupancy Date if no date is specified or if the date specified is later than 90 days before the First Tentative Occupancy Date. This time limitation does not apply to the condition in subparagraph 1(b)(iv) of Schedule A which must be satisfied or waived by the Vendor within 60 days following the later of: (A) the signing of the Purchase Agreement; and (B) the satisfaction or waiver by the Purchaser of a Purchaser financing condition permitted under paragraph (k) below.

Note: The parties must add additional pages as an appendix to this Addendum if there are additional Early Termination Conditions.

- (e) There are no Early Termination Conditions applicable to this Purchase Agreement other than those identified in subparagraph (d) above and any appendix listing additional Early Termination Conditions.
- (f) The Vendor agrees to take all commercially reasonable steps within its power to satisfy the Early Termination Conditions identified in subparagraph (d) above.
- (g) For conditions under paragraph 1(a) of Schedule A the following applies:
 - (i) conditions in paragraph 1(a) of Schedule A may not be waived by either party;
 - (ii) the Vendor shall provide written notice not later than five (5) Business Days after the date specified for satisfaction of a condition that: (A) the condition has been satisfied; or (B) the condition has not been satisfied (together with reasonable details and backup materials) and that as a result the Purchase Agreement is terminated; and
 - (iii) if notice is not provided as required by subparagraph (ii) above then the condition is deemed not satisfied and the Purchase Agreement is terminated.
- (h) For conditions under paragraph 1(b) of Schedule A the following applies:
 - (i) conditions in paragraph 1(b) of Schedule A may be waived by the Vendor;
 - (ii) the Vendor shall provide written notice on or before the date specified for satisfaction of the condition that:
 (A) the condition has been satisfied or waived; or (B) the condition has not been satisfied nor waived, and that as a result the Purchase Agreement is terminated; and
 - (iii) if notice is not provided as required by subparagraph (ii) above then the condition is deemed satisfied or waived and the Purchase Agreement will continue to be binding on both parties.
- (i) The Purchase Agreement may be conditional until Closing (transfer to the Purchaser of title to the home), upon compliance with the subdivision control provisions (section 50) of the *Planning Act* and, if applicable, registration of the declaration and description for the Building under the *Condominium Act*, 1998, which compliance shall be obtained by the Vendor at its sole expense, on or before Closing.
- (j) The Purchaser is cautioned that there may be other conditions in the Purchase Agreement that allow the Vendor to terminate the Purchase Agreement due to the fault of the Purchaser.
- (k) The Purchase Agreement may include any condition that is for the sole benefit of the Purchaser and that is agreed to by the Vendor (e.g., the sale of an existing dwelling, Purchaser financing or a basement walkout). The Purchase Agreement may specify that the Purchaser has a right to terminate the Purchase Agreement if any such condition is not met, and may set out the terms on which termination by the Purchaser may be effected.



MAKING A COMPENSATION CLAIM

7. Delayed Occupancy Compensation

- (a) The Vendor warrants to the Purchaser that, if Occupancy is delayed beyond the Firm Occupancy Date (other than by mutual agreement or as a result of Unavoidable Delay as permitted under sections 4 and 5), then the Vendor shall compensate the Purchaser up to a total amount of \$7,500, which amount includes: (i) payment to the Purchaser of a set amount of \$150 a day for living expenses for each day of delay until the Occupancy Date or the date of termination of the Purchase Agreement, as applicable under paragraph (b) below; and (ii) any other expenses (supported by receipts) incurred by the Purchaser due to the delay.
- (b) Delayed occupancy compensation is payable only if: (i) Occupancy and Closing occurs; or (ii) the Purchase Agreement is terminated or deemed to have been terminated under paragraph 10(b) of this Addendum. Delayed occupancy compensation is payable only if the Purchaser's claim is made to Tarion in writing within one (1) year after Occupancy, or after termination of the Purchase Agreement, as the case may be, and otherwise in accordance with this Addendum. Compensation claims are subject to any further conditions set out in the ONHWP Act.
- (c) If the Vendor gives written notice of a Delayed Occupancy Date to the Purchaser less than 10 days before the Firm Occupancy Date, contrary to the requirements of paragraph 3(c), then delayed occupancy compensation is payable from the date that is 10 days before the Firm Occupancy Date.
- (d) Living expenses are direct living costs such as for accommodation and meals. Receipts are not required in support of a claim for living expenses, as a set daily amount of \$150 per day is payable. The Purchaser must provide receipts in support of any claim for other delayed occupancy compensation, such as for moving and storage costs. Submission of false receipts disentitles the Purchaser to any delayed occupancy compensation in connection with a claim.
- (e) If delayed occupancy compensation is payable, the Purchaser may make a claim to the Vendor for that compensation after Occupancy or after termination of the Purchase Agreement, as the case may be, and shall include all receipts (apart from living expenses) which evidence any part of the Purchaser's claim. The Vendor shall assess the Purchaser's claim by determining the amount of delayed occupancy compensation payable based on the rules set out in section 7 and the receipts provided by the Purchaser, and the Vendor shall promptly provide that assessment information to the Purchaser. The Purchaser and the Vendor shall use reasonable efforts to settle the claim and when the claim is settled, the Vendor shall prepare an acknowledgement signed by both parties which:
 - (i) includes the Vendor's assessment of the delayed occupancy compensation payable;
 - (ii) describes in reasonable detail the cash amount, goods, services, or other consideration which the Purchaser accepts as compensation (the "Compensation"), if any; and
 - (iii) contains a statement by the Purchaser that the Purchaser accepts the Compensation in full satisfaction of any delayed occupancy compensation payable by the Vendor.
- (f) If the Vendor and Purchaser cannot agree as contemplated in paragraph 7(e), then to make a claim to Tarion the Purchaser must file a claim with Tarion in writing within one (1) year after Occupancy. A claim may also be made and the same rules apply if the sale transaction is terminated under paragraph 10(b), in which case, the deadline for a claim is one (1) year after termination.
- (g) If delayed occupancy compensation is payable, the Vendor shall either pay the compensation as soon as the proper amount is determined; or pay such amount with interest (at the prescribed rate as specified in subsection 19(1) of O.Reg. 48/01 of the *Condominium Act, 1998*), from the Occupancy Date to the date of Closing, such amount to be an adjustment to the balance due on the day of Closing.

8. Adjustments to Purchase Price

Only the items set out in Schedule B (or an amendment to Schedule B), shall be the subject of adjustment or change to the purchase price or the balance due on Closing. The Vendor agrees that it shall not charge as an adjustment or readjustment to the purchase price of the home, any reimbursement for a sum paid or payable by the Vendor to a third party unless the sum is ultimately paid to the third party either before or after Closing. If the Vendor charges an amount in contravention of the preceding sentence, the Vendor shall forthwith readjust with the Purchaser. This section shall not: restrict or prohibit payments for items disclosed in Part I of Schedule B which have a fixed fee; nor shall it restrict or prohibit the parties from agreeing on how to allocate as between them, any rebates, refunds or incentives provided by the federal government, a provincial or municipal government or an agency of any such government, before or after Closing.

MISCELLANEOUS

9. Ontario Building Code - Conditions of Occupancy

- (a) On or before the Occupancy Date, the Vendor shall deliver to the Purchaser:
 - (i) an Occupancy Permit (as defined in paragraph (d)) for the home; or
 - (ii) if an Occupancy Permit is not required under the Building Code, a signed written confirmation by the Vendor that all conditions of occupancy under the Building Code have been fulfilled and Occupancy is permitted under the Building Code.



- (b) Notwithstanding the requirements of paragraph (a), to the extent that the Purchaser and the Vendor agree that the Purchaser shall be responsible for one or more prerequisites to obtaining permission for Occupancy under the Building Code, (the "Purchaser Occupancy Obligations"):
 - (i) the Purchaser shall not be entitled to delayed occupancy compensation if the reason for the delay is that the Purchaser Occupancy Obligations have not been completed;
 - (ii) the Vendor shall deliver to the Purchaser, upon fulfilling all prerequisites to obtaining permission for Occupancy under the Building Code (other than the Purchaser Occupancy Obligations), a signed written confirmation that the Vendor has fulfilled such prerequisites; and
 - (iii) if the Purchaser and Vendor have agreed that such prerequisites (other than the Purchaser Occupancy Obligations) are to be fulfilled prior to Occupancy, then the Vendor shall provide the signed written confirmation required by subparagraph (ii) on or before the Occupancy Date.
- (c) If the Vendor cannot satisfy the requirements of paragraph (a) or subparagraph (b)(ii), the Vendor shall set a Delayed Occupancy Date (or new Delayed Occupancy Date) on a date that the Vendor reasonably expects to have satisfied the requirements of paragraph (a) or subparagraph (b)(ii), as the case may be. In setting the Delayed Occupancy Date (or new Delayed Occupancy Date), the Vendor shall comply with the requirements of section 3, and delayed occupancy compensation shall be payable in accordance with section 7. Despite the foregoing, delayed occupancy compensation shall not be payable for a delay under this paragraph (c) if the inability to satisfy the requirements of subparagraph (b)(ii) is because the Purchaser has failed to satisfy the Purchaser Occupancy Obligations.
- (d) For the purposes of this section, an "Occupancy Permit" means any written or electronic document, however styled, whether final, provisional or temporary, provided by the chief building official (as defined in the *Building Code Act*) or a person designated by the chief building official, that evidences that permission to occupy the home under the Building Code has been granted.

10. Termination of the Purchase Agreement

- (a) The Vendor and the Purchaser may terminate the Purchase Agreement by mutual written agreement. Such written mutual agreement may specify how monies paid by the Purchaser, including deposit(s) and monies for upgrades and extras are to be allocated if not repaid in full.
- (b) If for any reason (other than breach of contract by the Purchaser) Occupancy has not been given to the Purchaser by the Outside Occupancy Date, then the Purchaser has 30 days to terminate the Purchase Agreement by written notice to the Vendor. If the Purchaser does not provide written notice of termination within such 30-day period, then the Purchase Agreement shall continue to be binding on both parties and the Delayed Occupancy Date shall be the date set under paragraph 3(c), regardless of whether such date is beyond the Outside Occupancy Date.
- (c) If: calendar dates for the applicable Critical Dates are not inserted in the Statement of Critical Dates; or if any date for Occupancy is expressed in the Purchase Agreement or in any other document to be subject to change depending upon the happening of an event (other than as permitted in this Addendum), then the Purchaser may terminate the Purchase Agreement by written notice to the Vendor.
- (d) The Purchase Agreement may be terminated in accordance with the provisions of section 6.
- (e) Nothing in this Addendum derogates from any right of termination that either the Purchaser or the Vendor may have at law or in equity on the basis of, for example, frustration of contract or fundamental breach of contract.
- (f) Except as permitted in this section, the Purchase Agreement may not be terminated by reason of the Vendor's delay in providing Occupancy alone.

11. Refund of Monies Paid on Termination

- (a) If the Purchase Agreement is terminated (other than as a result of breach of contract by the Purchaser), then unless there is agreement to the contrary under paragraph 10(a), the Vendor shall refund all monies paid by the Purchaser including deposit(s) and monies for upgrades and extras, within 10 days of such termination, with interest from the date each amount was paid to the Vendor to the date of refund to the Purchaser. The Purchaser cannot be compelled by the Vendor to execute a release of the Vendor as a prerequisite to obtaining the refund of monies payable as a result of termination of the Purchase Agreement under this paragraph, although the Purchaser may be required to sign a written acknowledgement confirming the amount of monies refunded and termination of the purchase transaction. Nothing in this Addendum prevents the Vendor and Purchaser from entering into such other termination agreement and/or release as may be agreed to by the parties.
- (b)The rate of interest payable on the Purchaser's monies shall be calculated in accordance with the *Condominium Act, 1998*.
- (c)Notwithstanding paragraphs(a) and (b) above, if either party initiates legal proceedings to contest termination of the Purchase Agreement or the refund of monies paid by the Purchaser, and obtains a legal determination, such amounts and interest shall be payable as determined in those proceedings.

12. Definitions

"Building" means the condominium building or buildings contemplated by the Purchase Agreement, in which the Property is located or is proposed to be located.

"Business Day" means any day other than: Saturday; Sunday; New Year's Day; Family Day; Good Friday; Easter Monday; Victoria Day; Canada Day; Civic Holiday; Labour Day; Thanksgiving Day; Remembrance Day; Christmas Day; Boxing Day; and any special holiday proclaimed by the Governor General or the Lieutenant Governor; and where New Year's Day, Canada Day or Remembrance Day falls on a Saturday or Sunday, the following Monday is



not a Business Day, and where Christmas Day falls on a Saturday or Sunday, the following Monday and Tuesday are not Business Days; and where Christmas Day falls on a Friday, the following Monday is not a Business Day.

"Closing" means completion of the sale of the home, including transfer of title to the home to the Purchaser.

"Commencement of Construction" means the commencement of construction of foundation components or elements (such as footings, rafts or piles) for the Building.

"Critical Dates" means the First Tentative Occupancy Date, any subsequent Tentative Occupancy Date, the Final Tentative Occupancy Date, the Firm Occupancy Date, the Delayed Occupancy Date, the Outside Occupancy Date and the last day of the Purchaser's Termination Period.

"Delayed Occupancy Date" means the date, set in accordance with section 3, on which the Vendor agrees to provide Occupancy, in the event the Vendor cannot provide Occupancy on the Firm Occupancy Date.

"Early Termination Conditions" means the types of conditions listed in Schedule A.

"Final Tentative Occupancy Date" means the last Tentative Occupancy Date that may be set in accordance with paragraph 1(d).

"Firm Occupancy Date" means the firm date on which the Vendor agrees to provide Occupancy as set in accordance with this Addendum.

"First Tentative Occupancy Date" means the date on which the Vendor, at the time of signing the Purchase Agreement, anticipates that the home will be complete and ready for Occupancy, as set out in the Statement of Critical Dates

"Formal Zoning Approval" occurs when the zoning by-law required for the Building has been approved by all relevant governmental authorities having jurisdiction, and the period for appealing the approvals has elapsed and/or any appeals have been dismissed or the approval affirmed.

"Occupancy" means the right to use or occupy the home in accordance with the Purchase Agreement.

"Occupancy Date" means the date the Purchaser is given Occupancy.

"Outside Occupancy Date" means the latest date that the Vendor agrees to provide Occupancy to the Purchaser, as confirmed in the Statement of Critical Dates.

"Property" or "home" means the home being acquired by the Purchaser from the Vendor, and its interest in the related common elements.

"Purchaser's Termination Period" means the 30-day period during which the Purchaser may terminate the Purchase Agreement for delay, in accordance with paragraph 10(b).

"Roof Assembly Date" means the date upon which the roof slab, or roof trusses and sheathing, as the case may be, are completed. For single units in a multi-unit block, whether or not vertically stacked, (e.g., townhouses or row houses), the roof refers to the roof of the block of homes unless the unit in question has a roof which is in all respects functionally independent from and not physically connected to any portion of the roof of any other unit(s), in which case the roof refers to the roof of the applicable unit. For multi-story, vertically stacked units, (e.g. typical high rise) roof refers to the roof of the Building.

"Statement of Critical Dates" means the Statement of Critical Dates attached to and forming part of this Addendum (in form to be determined by Tarion from time to time), and, if applicable, as amended in accordance with this Addendum

"The ONHWP Act" means the Ontario New Home Warranties Plan Act including regulations, as amended from time to time.

"Unavoidable Delay" means an event which delays Occupancy which is a strike, fire, explosion, flood, act of God, civil insurrection, act of war, act of terrorism or pandemic, plus any period of delay directly caused by the event, which are beyond the reasonable control of the Vendor and are not caused or contributed to by the fault of the Vendor

"Unavoidable Delay Period" means the number of days between the Purchaser's receipt of written notice of the commencement of the Unavoidable Delay, as required by paragraph 5(b), and the date on which the Unavoidable Delay concludes.

13. Addendum Prevails

The Addendum forms part of the Purchase Agreement. The Vendor and Purchaser agree that they shall not include any provision in the Purchase Agreement or any amendment to the Purchase Agreement or any other document (or indirectly do so through replacement of the Purchase Agreement) that derogates from, conflicts with or is inconsistent with the provisions of this Addendum, except where this Addendum expressly permits the parties to agree or consent to an alternative arrangement. The provisions of this Addendum prevail over any such provision.

14. Time Periods, and How Notice Must Be Sent

- (a) Any written notice required under this Addendum may be given personally or sent by email, fax, courier or registered mail to the Purchaser or the Vendor at the address/contact numbers identified on page 2 or replacement address/contact numbers as provided in paragraph (c) below. Notices may also be sent to the solicitor for each party if necessary contact information is provided, but notices in all events must be sent to the Purchaser and Vendor, as applicable. If email addresses are set out on page 2 of this Addendum, then the parties agree that notices may be sent by email to such addresses, subject to paragraph (c) below.
- (b) Written notice given by one of the means identified in paragraph (a) is deemed to be given and received: on the date of delivery or transmission, if given personally or sent by email or fax (or the next Business Day if the date of delivery or transmission is not a Business Day); on the second Business Day following the date of sending by courier; or on the fifth Business Day following the date of sending, if sent by registered mail. If a postal stoppage or interruption occurs, notices shall not be sent by registered mail, and any notice sent by registered mail within 5



Business Days prior to the commencement of the postal stoppage or interruption must be re-sent by another means in order to be effective. For purposes of this section 14, Business Day includes Remembrance Day, if it falls on a day other than Saturday or Sunday, and Easter Monday.

- (c) If either party wishes to receive written notice under this Addendum at an address/contact number other than those identified on page 2 of this Addendum, then the party shall send written notice of the change of address, fax number, or email address to the other party in accordance with paragraph (b) above.
- (d) Time periods within which or following which any act is to be done shall be calculated by excluding the day of delivery or transmission and including the day on which the period ends.
- (e) Time periods shall be calculated using calendar days including Business Days but subject to paragraphs (f), (g) and (h) below.
- (f) Where the time for making a claim under this Addendum expires on a day that is not a Business Day, the claim may be made on the next Business Day.
- (g) Prior notice periods that begin on a day that is not a Business Day shall begin on the next earlier Business Day, except that notices may be sent and/or received on Remembrance Day, if it falls on a day other than Saturday or Sunday, or Easter Monday.
- (h) Every Critical Date must occur on a Business Day. If the Vendor sets a Critical Date that occurs on a date other than a Business Day, the Critical Date is deemed to be the next Business Day.
- (i) Words in the singular include the plural and words in the plural include the singular.
- (j) Gender-specific terms include both sexes and include corporations.

15. Disputes Regarding Termination

- (a) The Vendor and Purchaser agree that disputes arising between them relating to termination of the Purchase Agreement under section 11 shall be submitted to arbitration in accordance with the *Arbitration Act*, 1991 (Ontario) and subsection 17(4) of the ONHWP Act.
- (b) The parties agree that the arbitrator shall have the power and discretion on motion by the Vendor or Purchaser or any other interested party, or of the arbitrator's own motion, to consolidate multiple arbitration proceedings on the basis that they raise one or more common issues of fact or law that can more efficiently be addressed in a single proceeding. The arbitrator has the power and discretion to prescribe whatever procedures are useful or necessary to adjudicate the common issues in the consolidated proceedings in the most just and expeditious manner possible. The *Arbitration Act*, 1991 (Ontario) applies to any consolidation of multiple arbitration proceedings.
- (c) The Vendor shall pay the costs of the arbitration proceedings and the Purchaser's reasonable legal expenses in connection with the proceedings unless the arbitrator for just cause orders otherwise.
- (d) The parties agree to cooperate so that the arbitration proceedings are conducted as expeditiously as possible, and agree that the arbitrator may impose such time limits or other procedural requirements, consistent with the requirements of the *Arbitration Act, 1991* (Ontario), as may be required to complete the proceedings as quickly as reasonably possible.
- (e) The arbitrator may grant any form of relief permitted by the *Arbitration Act*, 1991 (Ontario), whether or not the arbitrator concludes that the Purchase Agreement may properly be terminated.

For more information please visit www.tarion.com



Property: GALLERIA 01

SCHEDULE A

Types of Permitted Early Termination Conditions

The Vendor of a condominium home is permitted to make the Purchase Agreement conditional as follows:

- (a) upon receipt of Approval from an Approving Authority for:
 - (i) a change to the official plan, other governmental development plan or zoning by-law (including a minor variance);
 - (ii) a consent to creation of a lot(s) or part-lot(s);
 - (iii) a certificate of water potability or other measure relating to domestic water supply to the home;
 - (iv) a certificate of approval of septic system or other measure relating to waste disposal from the home;
 - (v) completion of hard services for the property or surrounding area (i.e., roads, rail crossings, water lines, sewage lines, other utilities);
 - (vi) allocation of domestic water or storm or sanitary sewage capacity;
 - (vii) easements or similar rights serving the property or surrounding area;
 - (viii) site plan agreements, density agreements, shared facilities agreements or other development agreements with Approving Authorities or nearby landowners, and/or any development Approvals required from an Approving Authority; and/or
 - (ix) site plans, plans, elevations and/or specifications under architectural controls imposed by an Approving Authority.

The above-noted conditions are for the benefit of both the Vendor and the Purchaser and cannot be waived by either party.

(b) upon:

- (i) receipt by the Vendor of confirmation that sales of condominium dwelling units have exceeded a specified threshold by a specified date;
- (ii) receipt by the Vendor of confirmation that financing for the project on terms satisfactory to the Vendor has been arranged by a specified date;
- (iii) receipt of Approval from an Approving Authority for a basement walkout; and/or
- (iv) confirmation by the Vendor that it is satisfied the Purchaser has the financial resources to complete the transaction.

The above-noted conditions are for the benefit of the Vendor and may be waived by the Vendor in its sole discretion.

2. The following definitions apply in this Schedule:

"Approval" means an approval, consent or permission (in final form not subject to appeal) from an Approving Authority and may include completion of necessary agreements (i.e., site plan agreement) to allow lawful access to and use and occupancy of the property for its intended residential purpose.

"Approving Authority" means a government (federal, provincial or municipal), governmental agency, Crown corporation, or quasi-governmental authority (a privately operated organization exercising authority delegated by legislation or a government).

3. Each condition must:

- (a) be set out separately;
- (b) be reasonably specific as to the type of Approval which is needed for the transaction; and
- (c) identify the Approving Authority by reference to the level of government and/or the identity of the governmental agency, Crown corporation or quasi-governmental authority.

4. For greater certainty, the Vendor is not permitted to make the Purchase Agreement conditional upon:

- (a) receipt of a building permit;
- (b) receipt of an occupancy permit; and/or
- (c) completion of the home.

GALLERIA 01

APPENDIX TO ADDENDUM Additional Early Termination Conditions

CONDITION #1 – PURCHASER'S FINANCIAL RESOURCES

<u>Description of the Early Termination Condition:</u> The Purchase Agreement is conditional upon confirmation by the Vendor that it is satisfied the Purchaser has the financial resources to complete the transaction. This condition is for the benefit of the Vendor and may be waived by the Vendor in its sole discretion. The date by which this condition is to be satisfied or waived by the Vendor is noted below. In the event that no notice is received from the Vendor by the Purchaser on or before the date set forth below that this condition has or has not been satisfied, it shall be deemed to have been satisfied.

The Approving Authority (as that term is defined in Schedu	ıle A) is: N/A	
The date by which Condition #1 is to be satisfied is the	day of	, 20

ADDITIONAL CONDITION #2 - SALES THRESHOLD

Description of the Early Termination Condition: The Purchase Agreement is conditional upon receipt by the Vendor of confirmation that eighty-five (85%) percent of the Residential Units in the Condominium have been sold. This condition is for the benefit of the Vendor and may be waived by the Vendor in its sole discretion. The date by which this condition is to be satisfied or waived by the Vendor is noted below. In the event that no notice is received from the Vendor by the Purchaser on or before the date set forth below that this condition has or has notbeen satisfied, it shall be deemed to have been satisfied.

The Approving Authority (as that term is defined in Schedule A) is: N/A The date by which Condition #2 is to be satisfied is **August 25, 2023.**

ADDITIONAL CONDITION #3 – FINANCING

<u>Description of the Early Termination Condition:</u> The Purchase Agreement is conditional upon receipt by the Vendor of confirmation that financing for the Condominium has been arranged, on terms satisfactory to the Vendor. This condition is for the benefit of the Vendor and may be waived by the Vendor in its sole discretion. The date by which this condition is to be satisfied or waived by the Vendor is noted below. In the event that no notice is received from the Vendor by the Purchaser on or before the date set forth below that this condition has or has not been satisfied, it shall be deemed to have been satisfied.

The Approving Authority (as that term is defined in Schedule A) is: N/A. The date by which Condition #3 is to be satisfied is **August 25, 2023**.

Purchaser's Initials
Purchaser's Initials
Vendor's Initials

Page 10-A of 12



Property: GALLERIA 01

SCHEDULE B

Adjustments to Purchase Price

PART I Stipulated Amounts/Adjustments

These are additional charges, fees or other anticipated adjustments to the final purchase price or balance due on Closing, the dollar value of which is stipulated in the Purchase Agreement and set out below.

- (i) administration fee of \$500 plus HST in the event that the Purchaser has agreed to pay the balance of the Purchase Price on the Firm Occupancy Date, but has failed to do so (s. 3.4(6));
- (ii) legal fees in respect of the holding and administering of the Deposit by the Vendor's Solicitors and in respect of the Vendor's Solicitors' holding and administering of monies paid on account of Extras, which legal fees shall not exceed the sum of \$70.00 per cheque processed, plus HST (s. 5.1(5)(iii));
- (iii) legal fees payable for each discharge of any of the Vendor's blanket mortgages, charges or debentures registered on the Land (in whole or in part) from the Purchased Home whether before or after either the registration of the Creating Documents or Closing, such legal fees not to exceed \$275.00, plus HST, per partial discharge, per Property (s. 5.1(5)(v));
- (iv) the cost of the Vendor's Solicitors' increased legal fees to electronically register documents as a result of compliance with Rule 3.4-16.7 of the Law Society of Ontario's Rules of Professional Conduct in connection with the closing of this transaction in the amount of \$275.00, plus HST(s. 5.1(5)(vii));
- (v) the Vendor's Solicitors' fees for any electronic document exchange service utilized by the Vendor's Solicitors in connection with Occupancy or Closing, such fee not to exceed \$75,00, plus HST, for each of Occupancy and Closing (s. 5.1(5)(Viii));
- (vi) a charge of \$90.00, plus HST, for each Purchased Home to cover the Vendor's legal fees in respect of monthly financial reporting to its surety and/or construction lender (s. 5.1(5)(ix));
- (vii) an administrative fee of \$250.00 plus HST for any cheque delivered by the Purchaser to the Vendor or the Vendor's Solicitors and not accepted by the Vendor's bank (in the case of the Vendor) or the Vendor's Solicitors' bank (in the case of the Vendor's Solicitors) for any reason (s. 5.1(13));
- (viii) an administrative fee not exceed \$500.00 plus HST to the Vendor's Solicitors for any modifications requested to the documents for Closing or Occupancy once issued in final form by the Vendor's Solicitors or for any changes necessitated by information not provided by the Purchaser to the Vendor's Solicitors prior to the issuance of such documents (including information in respect of matters related to the Rebate) (s. 5.1(14));
- (ix) the amount of \$100.00, inclusive of HST, for the preparation and delivery of a Status Certificate on the Closing Date (s. 5.1(15));
- (x) an administration fee of \$800 plus HST for each request by the Purchaser to alter a finishing item or colour selection after the date on which the original selection is finalized (s. 6.2(7)); and
- (xi) an administration fee of \$300 plus HST for each appointment scheduled by the Purchaser for selecting finishing items that is confirmed by the Purchaser and then missed, cancelled or rescheduled by the Purchaser (s. 6.2(7)).

PART II All Other Adjustments – to be determined in accordance with the terms of the Purchase Agreement

These are additional charges, fees or other anticipated adjustments to the final purchase price or balance due on Closing which will be determined after signing the Purchase Agreement, all in accordance with the terms of the Purchase Agreement.

- (xii) interest on the deposit paid by the Vendor to the Purchaser (s. 2.1);
- (xiii) a fee for consenting to an assignment, sublet or sublicense of the Purchased Home (s. 3.4(2));
- (xiv) Occupancy Fee and any adjustments thereto (s. 3.4(3)(a) and (b), s. 3.4(7) and s. 5.1(3));
- (xv) an estimate of adjustments as of the Closing Date shall be paid on the Firm Occupancy Date (s. 3.4(10));
- (xvi) the cost to remove and/or repair any Unauthorized Work, and any other portion of the Purchased Home thereby affected, including without limitation any compensation for time lost by the delay resulting from any Unauthorized Work (s. 3.4(14));
- (xvii) any costs or expenses incurred by the Vendor as a result of damages caused by the Purchaser or default of the Purchaser and interest thereon (s. 3.4(19));
- (xviii) insurance premiums, hydro, cable TV, water, gas rates and fuel (except insofar as the same are included in Common Expenses), Common Expenses and that portion of the reserve fund that is attributable to the Purchased Home (s. 5.1(2) and (4));
- (xix) the adjustment in respect of realty taxes (including any supplementary assessment with respect thereto, any local improvement charges, sewer-impost charges and any fee charged by any Approving Authority to open a tax account or assessment roll number for the Purchased Home) and any new taxes applicable to the Purchased Home which are not yet exigible as of the date of acceptance of this Purchase Agreement (s. 5.1(4));



Property: GALLERIA 01

- (xx) the cost of enrolment of the Purchased Home under ONHWP Act, including taxes (based on the price of the unit) (s. 5.1(5)(i));
- the amounts attributed to the Purchased Home by the Vendor, based upon the proportionate share of Common Expenses for the Purchased Home as set out in the Creating Documents, plus HST, in respect of any payment for an obligation imposed by the City of Toronto or any other Approving Authority in connection with or in any way related to the Condominium or the Galleria Master Development pursuant to any agreement (including any "Section 37" agreement, land exchange agreement, contribution agreement or any other agreement) with the City of Toronto or any other Approving Authority, pursuant to Section 37 of the Planning Act, R.S.O. 1990, c. P. 13, pursuant to the Official Plan of the City of Toronto or otherwise, including: (A) any parkland contribution (or cash-in-lieu of same); (B) any art levy; and (C) funds for any community benefit (s. 5.1(5)(x));
- (xxii) all costs or charges in respect of any Smart Metering, Utility Services and Smart Homes Services including any Smart Metering Charges, Utility Service Charges or Smart Home Services Charges which are charged to the Vendor, the cost of all Smart Meters and Smart Home Services installed in the Property or in respect of the Property including the cost of the installation of all such Smart Meters or Smart Home Services or in the event that the Smart Meters or Smart Home Services are leased, the lease payments and any other associated fees, the payment of any security deposit required by any Smart Metering Provider, Utility Service Provider or Smart Home Services Provider or Smart Home Services Provider for the Property (s. 5.1(6));
- (xxiii) an amount equal to the real estate transaction insurance levy surcharge imposed by The Law Society of Ontario on the Vendor's portion of the transaction, plus HST (s. 5.1(5)(ii));
- (xxiv) the amount of any special assessments levied by the Condominium against the Purchased Home at any time following the turnover meeting in respect of the Condominium held pursuant to Section 43 of the Condo Act (s. 5.1(5)(iv));
- (xxv) a portion of the cost incurred by the Vendor in respect of the site review conducted pursuant to Tarion Builder Bulletin 19R and all charges relating to, incidental to or incurred in connection with such review, which portion of cost shall be determined by the Vendor, based upon the proportionate share of common expenses for the Purchased Home, plus HST (s. 5.1(5)(vi));
- (xxvi) all costs or charges in respect of any Smart Metering, Utility Services and Smart Homes Services which are charged to the Vendor (s. 5.1(6);
- (xxvii) costs of any legal fees and disbursements which the Vendor or Vendor's Solicitors may charge to the Purchaser's Solicitors for not utilizing TERS (s. 5.1(9));
- (xxviii) the amount of any increase in any development charge and/or education development charge or levy assessed, paid or payable by the Vendor pursuant to the *Development Charges Act* (Ontario), the *Education Act* (Ontario), or pursuant to any other similar legislation, or any other charges or levies payable to any Approving Authority since September 10, 2019 (s. 5.1(10));
- (xxix) HST for any chattels involved in the transaction (s. 5.1(11));
- an amount representing interest on the balance of the Purchase Price calculated at the Vendor's borrowing rate from its principal banker from the Closing Date to and including the next following banking date (s. 5.1(16));
- (xxxi) Vendor's Solicitors' reasonable fees in respect of the preparation of any security for the Purchaser's financing from the Vendor's Lender if Section 2.2(3) of the Purchase Agreement applies (s. 5.1(17));
- (xxxii) all other adjustments provided for in the Purchase Agreement (s. 5.1(18));
- (xxxiii) HST exigible on the purchase of the Purchased Home less the Rebate, if the Purchaser qualifies (s. 5.2);
- (xxxiv) any applicable credit to the Vendor on account of HST Rebate if the Purchaser fails to provide satisfactory evidence that it is entitled to the HST Rebate (s. 5.3(4), (5) and (6));
- (xxxv) HST exigible regarding any adjustments payable by the Purchaser pursuant to this Purchase Agreement, or any Extras, changes or upgrades to the Purchased Home agreed upon by the Vendor and Purchaser following the acceptance by the Vendor of this Purchase Agreement (s. 5.3(7));
- (xxxvi) the amount of any reduction in the HST Rebate as a result of any increases in the Purchase Price due to the addition of Extras, changes, upgrades or adjustments (s. 5.3(7));
- (xxxvii) all costs associated with the Purchaser's request to alter a finishing item or colour selection after the date on which the original selection is finalized (s. 6.2(7));
- (xxxviii) any costs of the Vendor to remove any title encumbrance or registration from title, or for any monies paid or for securities given (including any legal fees and disbursements) and interest on any unpaid amounts (s. 8.1(2));
- (xxxix) any costs and expenses of the Vendor in obtaining and registering an order to remove any caution, certificate of pending litigation, Purchaser's lien, or any other notice or document of any type which is registered on title (including any legal fees and disbursements) (s. 8.2); and
- (xl) if applicable, any amount withheld by the Vendor on account of the requisite tax owing to Canada Revenue Agency in respect of interest payable to the Purchaser under the Purchase Agreement (s. 9.3).

DocuSign Envelope ID: 322166C9-12C6-480C-AA03-1169E8ECCA95

DEMONSTRATION DOCUMENT ONLY
PROVIDED BY DOCUSIGN ONLINE SIGNING SERVICE
999 3rd Ave, Suite 1700 • Seattle • Washington 98104 • (206) 219-0200
www.docusign.com

_____ I confirm receipt of the fully executed Agreement of Purchase and Sale.